Understanding Relevant Sustainable Development Goal Targets Related to Labour Migration in the Association of Southeast Asian Nations During the Coronavirus Disease Pandemic

By

Shivakumar Srinivas Satya Sivaraman



Understanding Relevant Sustainable Development Goal Targets Related to Labour Migration in the Association of Southeast Asian Nations During the Coronavirus Disease Pandemic

Economic Research Institute for ASEAN and East Asia (ERIA) Sentral Senayan II 6th Floor Jalan Asia Afrika no.8, Gelora Bung Karno Senayan, Jakarta Pusat 12710 Indonesia

© Economic Research Institute for ASEAN and East Asia, 2021 ERIA Research Project Report FY2021 No. 04 Published in June 2021

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form by any means electronic or mechanical without prior written notice to and permission from ERIA.

The findings, interpretations, conclusions, and views expressed in their respective chapters are entirely those of the author/s and do not reflect the views and policies of the Economic Research Institute for ASEAN and East Asia, its Governing Board, Academic Advisory Council, or the institutions and governments they represent. Any error in content or citation in the respective chapters is the sole responsibility of the author/s.

Material in this publication may be freely quoted or reprinted with proper acknowledgement.

Contents

| List of Figures | iv |
|--|---|
| List of Tables | v |
| List of Boxes | vi |
| Abbreviations and Acronyms | vii |
| Executive Summary | ix |
| | |
| Regional Labour Migration in the Association of Southeast Asian Nations: Background and Context | 1 |
| Impact of the Coronavirus Disease Pandemic | 15 |
| Impact of the Coronavirus Disease Pandemic on the Sustainable Development Goals | 30 |
| Way Forward | 70 |
| References | 86 |
| | List of Tables List of Boxes Abbreviations and Acronyms Executive Summary Regional Labour Migration in the Association of Southeast Asian Nations: Background and Context Impact of the Coronavirus Disease Pandemic Impact of the Coronavirus Disease Pandemic on the Sustainable Development Goals Way Forward |

List of Figures

| Figure 1.1 | An Overview of the Association of Southeast Asian Nations Consensus | 8 |
|------------|--|----|
| Figure 2.1 | Estimates of Gross Domestic Product Loss in the Association of Southeast Asian Nations | 18 |
| Figure 2.2 | Coronavirus Disease Containment and Mitigation Measures in Southeast Asia | 20 |
| Figure 2.3 | Impacts of Job Loss on Migrant Workers from Myanmar | 28 |
| Figure 3.1 | Links between the Sustainable Development Goals and Migration | 31 |
| Figure 3.2 | The Association of Southeast Asian Nations' Performance on the Sustainable Development Goals, 2015 | 32 |
| Figure 3.3 | Post-Coronavirus Disease Pandemic Job Loss amongst Intra- Association of Southeast Asian Nations Migrant Workers | 34 |
| Figure 3.4 | Employment in the Tourism Sector as a Share of Total Employment and Share of Informality in Tourism Employment, Latest Available Year | 35 |
| Figure 3.5 | Employment in the Tourism Sector as a Share of Total Employment by Sex, Latest Available Year | 36 |
| | (%) | |
| Figure 3.6 | Association of Southeast Asian Nations Women Migrants at Mid-Year | 52 |
| Figure 3.7 | Proportion of People Who Experienced Job Losses or Decreased Paid Work Hours since the Spread of Coronavirus Disease, by Sex (January–June 2020) | 54 |
| Figure 3.8 | Top Remittance Recipients in the East Asia and Pacific Region (2019) | 66 |
| Figure 3.9 | Remittance Fees to the Philippines versus East Asia and the Pacific | 69 |

List of Tables

| Table 1.1 | Socio-economic Indicators, by the Association of Southeast Asian Nations Member States | 3 |
|-----------|---|----|
| Table 1.2 | Social Security Coverage of Migrant Workers by Country and Branch, 2017 | 9 |
| Table 1.3 | Ratification of Conventions Related to Migrant Workers by Association of Southeast Asian Nations Member States | 13 |
| Table 3.1 | Minimum Wages in Association of Southeast Asian Nations Member Countries | 64 |
| Table 3.2 | Impact on Global Remittance Inflows (Worst-Case Scenario) | 67 |
| Table 4.1 | Summary of Association of Southeast Asian Nations Member States Stimulus Packages to Counter the Impact of the Pandemic | 75 |
| Table 4.2 | Migration-Related Responses to the Coronavirus Disease within the Association of Southeast Asian Nations | 76 |

List of Boxes

| Box 1 | Social Protection Floors | 11 |
|--------|---|----|
| Box 2 | Association of Southeast Asian Nations Responses and Reactions | 19 |
| Box 3 | Association of Southeast Asian Nations Stimulus Packages | 26 |
| Box 4 | Civil Society Organizations Response to the Coronavirus Disease Pandemic | 29 |
| Box 5 | Collapse of the Travel and Tourism Industry | 35 |
| Box 6 | Importance of Equal Treatment | 40 |
| Box 7 | Noncommunicable Diseases | 46 |
| Box 8 | Risks of Increasing Child Labour | 50 |
| Box 9 | Impact on Trafficking | 54 |
| Box 10 | Irregular Workers | 57 |
| Box 11 | Migration Policies | 65 |

Abbreviations and Acronyms

| ACMW | ASEAN Committee on the Implementation of the Declaration on the Protection and Promotion of the Rights of Migrant Workers |
|----------|---|
| ACRF | ASEAN Comprehensive Recovery Framework |
| ADB | Asian Development Bank |
| AEC | ASEAN Economic Community |
| AMS | ASEAN Member States |
| ASEAN | Association of Southeast Asian Nations |
| ATUC | ASEAN Trade Union Council |
| BHRRC | Business and Human Rights Resource Centre |
| CFR | Council on Foreign Relations |
| CMIM | Chiang Mai Initiative Multilateralisation |
| COVID-19 | Coronavirus Disease |
| CSO | Civil Society Organisations |
| CSIS | Center for Strategic and International Studies |
| FAO | UN Food and Agriculture Organization |
| GDP | gross domestic product |
| GFMD | Global Forum on Migration and Development |
| HRW | Human Rights Watch |
| IFPRI | International Food Policy Research Institute |
| ILO | International Labour Organisation |
| IMF | International Monetary Fund |
| IOM | International Organization for Migration |
| IRGSC | Institute of Resource Governance and Social Change |
| Lao PDR | Lao People's Democratic Republic |
| MFA | Migrant Forum in Asia |
| MGI | Migration Governance Indicators |
| Migof | Migration Governance Framework |
| MOU | Memorandum of Understanding |
| MRA | mutual recognition arrangements |
| MSME | micro, small and medium-sized enterprises |
| NCD | non-communicable diseases |
| NGO | non-government Agency |
| ODI | Overseas Development Institute |
| OECD | Organisation for Economic Co-operation and Development |
| OFW | overseas Filipino workers |
| | |

| RCEP | Regional Comprehensive Economic Partnership |
|---------|--|
| SAARC | South Asian Association for Regional Cooperation |
| SDG | Sustainable Development Goals |
| SPF | social protection floors |
| TF-AMW | Task Force on ASEAN Migrant Workers |
| UHC | Universal Health Coverage |
| UN | United Nations |
| UN DESA | United Nations Department of Economic and Social Affairs |
| UNESCAP | United Nations Economic and Social Commission for Asia and the Pacific |
| UNICEF | United Nations International Children's Emergency Fund |
| UNODC | United Nations Office on Drugs and Crime |
| WHO | World Health Organization |

Executive Summary*

Southeast Asia has been one of the fastest growing regions in the world for the last several decades. Since 1980, the 10 member countries of the Association of Southeast Asian Nations (ASEAN) have experienced growth at an average rate of 5.4%, well above the global average of 3.4%. Although the 1997 Asian Economic Crisis dealt a severe blow to ASEAN economies, they emerged even stronger, with gross domestic product (GDP) growing at an average rate of 5.1% during 2000–2004, and 4.9% during 2005–2009 (Lee, 2015). ASEAN countries also rebounded from the global economic crisis of 2008, with the region's GDP growing by 4.7% between 2008 and 2010. With economic recovery underway across the region, the ASEAN+6 countries (including Australia, China, India, Japan, the Republic of Korea [henceforth Korea], and New Zealand) saw average annual growth of 6% from 2011 to 2015, and maintained this growth momentum until the outbreak of the coronavirus disease (COVID-19) in early 2020 (Vu, 2020). In addition to the flow of capital, labour migration has emerged as a significant driver of economic growth and development in both origin and destination countries within the ASEAN region.[†]

According to the most recent figures from the United Nations Department of Economic and Social Affairs (UN DESA, 2016), some 20.2 million ASEAN nationals live outside their country of birth for work or as accompanying spouses.[‡] According to the International Labour Organisation (ILO, 2020j; also see United Nations [UN], 2020a) and the United Nations Entity for Gender Equality and the Empowerment of Women (UN Women) (2017), the total population of international migrants in ASEAN is 9.9 million, of whom nearly 6.9 million people have moved between countries within the region.

^{*} The authors would like to acknowledge the support received from Beryl Jane Duterte and Thanigan. Comments and suggestions received from Dr. V. Anbumozhi and the Economic Research Institute for ASEAN and East Asia's specialists are gratefully acknowledged.

[†] Some of the information for this note on migration, migrant workers, and Sustainable Development Goals (SDGs) comes from the publications of (i) the International Labour Organizatio (ILO) Southeast Office in Bangkok, (ii) International Organization for Migration (<u>www.iom.org</u>), (iii) UN DESA (2016) on SDGs, and (iv) the United Nations Economic and Social Commission for Asia and the Pacific (UNESCAP) (2020c): Asia and the Pacific SDG Progress Report 2020.

[‡] In the ILO Article 11(1) of the Convention, the term 'migrant for employment' means a person who migrates from one country to another with a view to being employed otherwise than on his or her own account. https://www.ilo.org/public/english/standards/relm/ilc/ilc87/r3-1b2.htm



Figure 1: Association of Southeast Asian Nations Map

Source: Devonshire-Ellis, C. 2012. 'Why ASEAN Matters for Your China Business', *China Briefing*. 22 May. <u>https://www.china-briefing.com/news/why-asean-matters-for-your-china-business/</u> (accessed 1 September 2020)

The ILO June 2020 report asserts that the actual number is probably significantly higher, as undocumented, irregular, or short-term migration is rampant in the region. In addition to agriculture, construction, and manufacturing, migrant workers in the ASEAN region are employed in a variety of service sectors, such as tourism, retail, and hospitality, which also employ a large number of seasonal workers.[§]

Since the beginning of 2020, the COVID-19 pandemic has significantly threatened people's lives and the region's economic outlook. The International Monetary Fund (IMF) acknowledged this in their October 2020 report, projecting that growth in the ASEAN+5 region (including Indonesia, Malaysia, the Philippines, Thailand, and Viet Nam) would shrink to -3.4% in 2020 – a drastic decline from the 2019 growth rate of 4.7% – because of the impact of the COVID-19 crisis. However, in 2021, growth in the region is projected to reach as high as 6.1% as economies rebound.

The present health and economic crisis has given rise to significant public policy issues in

[§] The ILO Migration for Employment Convention (Revised), 1949 (No. 97) defines a 'migrant worker' as a person who migrates from one country to another with a view to being employed otherwise than on his or her own account. The scope of this convention excludes frontier workers, the short-term entry of members of the liberal professions and artists, and seafarers. Additionally, the ILO Migrant Workers (Supplementary Provisions) Convention, 1975 (No. 143) follows this definition but excludes two additional categories including (i) persons coming specifically for training or education, and (ii) persons admitted temporarily to a country at the request of their employer to undertake specific duties or assignments for a limited and defined period of time, and who are required to leave that country upon the completion of their duties or assignments.

Southeast Asia. Importantly, the economic downturn and severe social disruption caused by the pandemic are expected to impair the region's progress towards meeting the 17 Sustainable Development Goals (SDGs), which encompass a very wide range of development sectors including reducing poverty, improving health and education indicators, facilitating women's empowerment, and tackling climate change. In this regard, the experiences gained in managing COVID-19 offer some insights for ASEAN Member States (AMS), such as enforcement of individual quarantine, early border controls, and extending access to public health facilities. The lessons learned should motivate governments to discuss challenges and policies related to macroeconomics, regional labour migration, strengthening social welfare systems, and reforms, which are critically necessary for the region's recovery and cross-border cooperation.

The impact on livelihoods is expected to be particularly severe. Although exact data for the pandemic's impact on employment are not yet available, this is projected to be substantial, especially for regional labour migrants and informal employment, which comprises over 75% of all jobs in Southeast Asia (ILO News and Letzing, 2020).

Civil society groups, academics, policy makers, and media in the region have rightly underlined the plight of migrant workers as being disproportionately at risk from the negative consequences of the pandemic. Even before the current crisis, many migrant workers were periodically subject to uncertain work and poor work conditions, inadequate and crowded living conditions, lockdowns and harsh containment measures, limited access to health care and basic services, and exploitative labour systems.

ASEAN governments must form a collective policy response to the COVID-19 crisis that is rapid and effective in reducing income insecurity and increasing people's capacity to manage and overcome ongoing economic shocks. As a first step, a collective declaration adopted in April 2020 at the Special ASEAN Summit on Coronavirus Disease 2019 reaffirmed the regional grouping's determination and commitment 'to act jointly and decisively to control the spread of the disease while mitigating its adverse impact on our people's livelihood, our societies and economies' (ASEAN, 2020k).

The purpose of this report is to analyse how the social and economic crisis precipitated by the COVID-19 pandemic will impact the very large number of cross-border migrant workers in the region. The report conducts this analysis by measuring the progress of the SDGs as well as the longer-term prospects of the Regional Cooperation and ASEAN Economic Community (AEC).

Methodology

The study was undertaken as the COVID-19 pandemic was still unfolding in the region and the situation of migrant workers was in flux. Therefore, estimates of the impact of the crisis on migrant workers are based on published reports, media articles, blogs, and interviews with key informants. A thorough review of the available literature and discussions for the report was conducted in May–August 2020. The review included various media and online reports published by different development organisations and civil society groups. Legal, institutional, technical, social, and other aspects of the impact of the COVID-19 crisis on migrant workers as well as lessons learned have been synthesised. These were confirmed through online interviews with key stakeholders in Cambodia, Myanmar, the Philippines, and Viet Nam. The interviews provided anecdotal evidence, validated initial findings, and confirmed conclusions and recommendations. The report presents, summarises, and interprets the evidence that has emerged so far based on contextual factors that have shaped the current situation in the region.

Some specific research questions examined in the report include the following:

- (i) What are the social and financial impacts of COVID-19 on migrant labour within the ASEAN region?
- (ii) How will the COVID-19 pandemic impact the achievement of SDGs in the ASEAN region?
- (iii) How has migrant labour coped with the disaster in the immediate term? Was there a delivery mechanism for providing migrant labourers with a social safety net?
- (iv) What measures have been taken at the regional level to address the problems of migrant labour?

Some assumptions made in the report include the following:

- (i) The COVID-19 pandemic will last until mid-2021, after which economic recovery can begin.
- (ii) The pandemic's economic impact, while severe, is likely reversible.
- (iii) Regional and international treaties will continue to be respected.
- (iv) Despite challenges, regional cooperation at the ASEAN level will remain strong.
- (v) Social and political conflict within and between AMS will be minimal.

Rationale for Choosing to Focus on the Sustainable Development Goals

The 2030 Development Agenda for Sustainable Development adopted on 25 September 2015 by 193 member countries of the UN General Assembly comprises 17 goals and 169 targets encompassing a wide range of development themes. It marks a milestone by mainstreaming migration, migrant workers, and the issues of inequality and decent work as integral components of development policy; and includes a number of targets recognising the economic value of migrants, such as SDGs 4, 5, 8, 10, 16, and 17 (Foresti and Hagen-Zanker, 2018). Target 10.7 in particular calls for the facilitation of 'safe, regular and responsible migration' and the implementation of 'well-managed migration policies'. While the SDGs are all relevant in a cross-cutting way to the situation of migrant workers

(International Organization for Migration [IOM], 2018), this report focuses on SDGs 1, 2, 3, 4, 5, 8, and 10, in which areas the impact of COVID-19 on regional migrant workers is more direct and immediate. It is also currently relatively easier to gather information to advocate policymaking and regional cooperation in these areas. The longer-term impact of the remaining SDGs will need to be studied separately in the future.

Limitations

This review has two principal limitations. First, there is a paucity of primary data on the impact of COVID-19 on different aspects of migrant workers' lives, including food security, loss of household income, the education of children, and health indicators. Second, as the pandemic is still unfolding, there is a widespread lack of clarity about the long-term impact of COVID-19 on ASEAN economies and policies, particularly with respect to country social protection and welfare systems and the type of regulatory reforms needed to address some of the issues confronted by intraregional migrant workers. The conclusions and recommendations of this report are therefore likely tentative and subject to further study.

Structure of the Report

This report is divided into four chapters. Chapter 1 explains the background and context for this study. Chapter 2 provides an overview of impact of the COVID-19 pandemic on the lives of migrant workers, as well as different aspects of labour migration. Chapter 3 presents a step-by-step analysis of how the current crisis will impact different SDGs relevant to migrants, and provides an analytical basis for determining the welfare requirements of migrant workers with a special emphasis on providing them with decent work (SDG 8). Chapter 4 summarises key recommendations made by civil society groups and academics, notes further areas for research, and lists areas for further ASEAN engagement in ensuring the rights of cross-country migrant workers and enhancing regional cooperation amongst the AMS, all of which will help to progress SDG targets in the future.

Chapter 1

Regional Labour Migration in the Association of Southeast Asian Nations: Background and Context

1. The Association of Southeast Asian Nation's Migrant Workers: Political and Economic Context

The ASEAN regional grouping consists of 10 countries comprising a population of more than 650 million.⁵ The regional grouping also encompasses several diverse economic, political, and social systems. In 2019, the Council for Foreign Relations reported that, in terms of economic development, Singapore has the highest GDP per capita amongst the group's members, at nearly \$65,000; while Myanmar's is the lowest at less than \$1,400.⁶ As of 2019, ASEAN had a combined GDP of \$2.8 trillion, and was also the third fastest-growing regional economy from 2010 to 2020, after China and India (United States [US]-ASEAN Business Council, 2019). More than \$3.6 trillion in global trade transits through the region each year.

The ASEAN regional grouping promotes economic, political, and security cooperation amongst its 10 members; and over the years has contributed to regional stability by developing common norms and principles to address shared challenges. In 1992, the ASEAN Free Trade Area was launched with the goal of creating a single market, increasing intra-ASEAN trade and investments, and attracting foreign investment. Intra-ASEAN trade as a share of the bloc's overall trade grew from about 19% in 1993 to 23% in 2017 (ASEAN, 2014; Maizland and Albert, 2020). According to the July 2020 ASEAN Policy Brief (Chandra, Muhajid, and Mahyassari, 2020), more than 90% of goods are currently traded with no tariffs across ASEAN.

The 1997 Asian financial crisis, which started in Thailand and spread across the region, highlighted the lack of social safety nets and welfare systems to protect poorer groups in times of crisis. In response, the AMS decided to push further integration of their economies through a series of agreements both amongst themselves and outside the region to protect their economies from speculative capital flight and other associated socioeconomic risks. In 2007, the AMS adopted the ASEAN Charter, a constitutional document that provided the grouping with legal status and an institutional framework, and that strives to facilitate regional integration while maintaining national autonomy and preserving regional diversity. With respect to labour, a social charter was formulated as part of the broader 2007 ASEAN Charter

⁵ ASEAN was established in August 1967 with five member states: Indonesia, Malaysia, the Philippines, Singapore, and Thailand. Brunei Darussalam joined on 8 January 1984, Viet Nam on 28 July 1995, the Lao People's Democratic Republic (Lao PDR) and Myanmar on 23 July 1997, and Cambodia on 30 April 1999. At present, ASEAN has 10 member states, and counts several other nations such as Australia, China, the European Union, India, Japan, Korea, and the United States (US) as dialogue partners.

⁶ All monetary denominations are in US dollars, unless otherwise stated.

(Economic Research Institute for ASEAN and East Asia, 2015; ASEAN, 2016a).

As per ASEAN (2008), the 2007 charter enshrines core principles, delineates requirements for membership, and lays out a blueprint for a community made up of three branches: the AEC, the ASEAN Political-Security Community, and the ASEAN Socio-Cultural Community. The AEC seeks to reduce or remove trade barriers to facilitate the free movement of goods, services, capital, and skilled labour within the region. By the end of 2019, all ASEAN nations had joined the ASEAN Single Window Live Operations, which facilitates the granting of preferential tariff treatment to goods originating within the regional grouping (ASEAN Single Window, 2020).

ASEAN currently has regional free trade agreements with Australia and New Zealand, China, India, Japan, and Korea; and since 2012 has also been negotiating the Regional Comprehensive Economic Partnership (RCEP), a proposed free trade agreement that would include all AMS, as well as Australia, China, India, Japan, Korea, and New Zealand. The RCEP aims to create an integrated market with 16 countries, making the products and services of each of these countries more accessible across the region (ASEAN, 2016b; US-ASEAN Business Council, 2019). The negotiations focus on trade in goods and services, investment, intellectual property, dispute settlement, e-commerce, small and medium-sized enterprises (MSMEs), and economic cooperation. Final adoption of the RCEP was scheduled for March 2020 but has been postponed due to COVID-19.

| | | | 2 | | g | Healt | th Protect | E. | |
|----------------------|-------------------------------|-------------------------------|------------------------|-------------------------------------|--|---|-------------------------------|---|--------------------------------------|
| | GDP per capita, 2018 (USD) | Population, 2018 (million) | Gini coefficient, 2017 | Unemployment (%) (various years) | Average monthly carnings of employe (USD equivalent, various years) | UHC service coverage index (out of 100) (%) | Insurance coverage (%) (a) | Catastrophic health spending (%).[10% (h) | Social protection for memployment |
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) |
| Brunei Darussalam | 30,645 | 0.4 | - | 8.7 | 1,147.8 | ≥80.0 | 100.0 | NA | None |
| Cambodia | 1,541 | 16.0 | 0.31* | 0.7 | 182.7 | 55.0 | 37.5 | 10.7 | Limited provisions |
| Indonesia | 3,930 | 265.0 | 0.39 | 4.4 | 173.6 | 49.0 | 84.0 | 3.6 | Limited provisions |
| Lao PDR | 2,627 | 6.9 | 0.36 | 9.4 | 285.5 | 94.0 | 94.0 | 3.0 | Yet to be implemented |
| Malaysia | 11,067 | 32.4 | 0.46 | 3.3 | 765.1 | 100.0 | 100.0 | 0.7 | Limited provisions |
| Myanmar | 1,441 | 53.6 | 0.38 | 0.9 | 141.6 | 2.0 | 2.0 | 3.4 | Yet to be implemented |
| Philippines | 3,215 | 106.6 | 0.40 | 2.3 | 256.1 | 78.0 | 78.0 | 6.3 | Limited provisions |
| Singapore | 64,567 | 5.6 | 0.46 | 4.1 | 3,557.8 | 93.0 | 93.0 | 9.0 | None |
| Thailand | 7,446 | 67.8 | 0.38 | 0.7 | 489.5 | 100 | 100 | 3.4 | Yes |
| Viet Nam | 2,546 | 94.7 | 0.35 | 2.0 | 286.8 | 89.9 | 89.9 | 9.7 | Yes |

Table 1.1: Socio-economic Indicators, by the Association of Southeast Asian Nations Member States

GDP = gross domestic product, Lao PDR = Lao People's Democratic Republic, UHC = universal health coverage. Note: Data presented in the table are from the sources identified, and may not be the most updated nor from the same years. For more, detailed information on the indicators, please check the data sources, all of which are available online. (a) Population coverage; (b) Proportion of population spending more than 10% of household consumption or income on out-of-pocket healthcare expenditure. *2016 data.

Source: Zulkhibri, M. and J.B. Sinay (2020), 'Assessing ASEAN Economy Policy Responses in a Pandemic', *ASEAN Policy Brief*, No. 2. 28 May. <u>https://asean.org/storage/2020/06/ASEAN-Policy-Brief-02_31-May-2020.pdf</u> (accessed on 28 October 2020).

2. Economic Integration and Migrant Workers

The main labour destination countries in the ASEAN region (accounting for around 91% of intra-ASEAN migrant workers) are Brunei Darussalam, Malaysia, Singapore, and Thailand (ILO, 2015a and ILO, 2020b). Cambodia, Indonesia, Lao PDR, the Philippines, and Viet Nam are the most prolific origin, or labour-sending, countries; and the largest flows occur amongst countries that share borders such as Cambodia, the Lao PDR, Myanmar, and Thailand; and Indonesia, Malaysia, and the Philippines. A substantial number of Malaysians also work in Singapore as 'guest workers' (Paitoonpong, 2011).⁷

⁷ For example, foreign workers have assumed low-wage, '3D' (dirty, dangerous, and demeaning) jobs in Singapore and Malaysia that the local populations no longer prefers to take. They also provide domestic workers so that middle-class women in Singapore can participate in the labour force. In addition, foreign workers have kept wages down and maintain the receiving countries' competitiveness in labour-intensive industries.

A study by Harkins and Lindgren (2018) indicates that the seven AMS – Cambodia, Indonesia, the Lao PDR, Myanmar, the Philippines, Thailand, and Viet Nam – deployed more than 2.1 million migrant workers within ASEAN. Intraregional labour migration is especially significant for Myanmar and Cambodia, which deploy 93.4% (Myanmar) and 66.8% (Cambodia) of their migrant workers to other AMS.

Many of these migrant workers engage in low- and medium-skilled occupations (such as construction workers, garment workers, fishers, and plantation labourers) under seasonal, temporary, or short-term contracts, with women constituting a disproportionate share (ILO and the Asian Development Bank [ADB], 2014). Six of the seven sending countries listed above accounted for more than 350,000 women migrant workers, representing 52% of deployed workers. In 2015, 88.3% of migrant workers from Indonesia were women, the highest share of all the AMS (ASEAN Trade Union Council [ATUC], 2020).

Regional labour migration across the AMS is triggered by several factors. Harkins and Lindgren (2018) identified the following critical factors: (i) disparities between countries in terms of socioeconomic development, (ii) political and civil strife displacing certain sections of the society (e.g. refugees from Myanmar living in Thailand) seeking local employment for survival, and (iii) demographic differences amongst AMS populations and shortages of cheap and skilled workers in higher income countries (see also Harkins, Lindgren, and Suravoranon, 2017). The political process of regional integration also plays an important role in encouraging migration within the regional grouping. Thus, the potential for intraregional migration to contribute to the growth and development of the bloc has been frequently recognised as a key issue in the grouping's various declarations and policy statements. Facilitating the cross-border movement of skilled labour within ASEAN features prominently in the newly established AEC.

One of the AEC's central purposes is to create a single market and production base that is stable, prosperous, highly competitive, and economically integrated. It also aims to facilitate effective trade and investment involving the free flow of goods, services, and investment; the smooth movement of business persons, professionals, talent, and labour; and freer flow of capital.

The AEC Blueprint (2009–2015) calls for 'managed mobility or facilitated entry for the movement of natural persons engaged in trade in goods, services, and investments, according to the prevailing regulations of the receiving country' (ASEAN Secretariat, 2008). Recognising the challenges of increasing labour mobility, ASEAN leaders mandated the organisation to strengthen national systems of social protection through the 2004 Vientiane Action Programme and to work towards adopting appropriate measures at the regional level to provide minimum uniform social protection coverage for skilled workers in the region (ASEAN, 2004). These efforts yielded the 2007 ASEAN Social Charter and 2016 ASEAN Consensus on Migrant Workers. The ASEAN Vientiane Action Programme 2004–2010 also included measures to protect high-skilled labour following the liberalisation of services within ASEAN and mutual recognition arrangements (MRAs) to facilitate the movement of skilled workers by the end of 2008 (ASEAN Secretariat, 2005).

The ASEAN Consensus on the Protection and Promotion of the Rights of Migrant Workers was launched in 2017 at the 31st ASEAN Summit in Manila, where several social and labour protections measures were proposed (ASEAN, 2018a and ASEAN, 2018b). This was followed by a high-level, multi-stakeholder dialogue in September 2018 amongst ASEAN representatives in Manila for implementing the consensus on responsible business conduct for safe labour migration in ASEAN.

ASEAN has begun expanding labour mobility under the AEC for eight high-skill occupations, including accountancy, architecture, dentistry, engineering, medicine, nursing, and tourism. However, progress in operationalising the MRAs 'remains painfully slow and uneven across countries and for all occupations' (Mendoza and Sugiyarto, 2017). Consequently, greater mobility for highly skilled professionals as envisioned by ASEAN and the AEC remains far from being realised.

According to a Migration Policy Institute and ADB report, for the MRAs to be fully implemented, the laws must be translated into a clear working process of mutual recognition and registration (Mendoza and Sugiyarto, 2017). The report reveals that many governments lack the institutional capacity to implement the MRAs, while wage disparities and poor working conditions in some areas have generally discouraged professional movement.

A 2019 ASEAN Secretariat report (ASEAN, 2019) on regional integration stated that jobs requiring high skills collectively represent only 0.3%–1.4% of total employment in AMS. Most workers involved in intra-ASEAN labour migration are employed in low-skilled, labour-intensive jobs in agriculture, fisheries, domestic work, manufacturing, and construction. Harkins and Lindgren (2018) suggested that, although workers in low-skilled jobs are estimated to constitute as much as 87% of intraregional labour migration flows, ASEAN has yet to establish policies regarding them.

2.1. Intra-Association of Southeast Asian Nations Agreements on Migrant Workers

Following the Vientiane Action Programme of 2005, the 2007 ASEAN Declaration on the Protection and Promotion of the Rights of Migrant Workers led to the establishment of an ASEAN committee on the implementation of the declaration, as well as the creation of the ASEAN Forum on Migrant Labour as a venue for regular consultations. By 2013, six AMS (Indonesia, Malaysia, Myanmar, the Philippines, Thailand, and Singapore) had ratified the Equality of Treatment (Accident Compensation) Convention, 1925 (No. 19) that provides occupational injury protection for non-national workers (Orbeta et al., 2013). However, there has been less progress with respect to other migrant worker-related conventions or bilateral social security agreements between AMS.

Despite several different efforts since 2004, many ASEAN countries still rely on bilateral memoranda of understanding (MOUs) focussing on regulating the flow of migrant labour, while clauses on welfare and social protection are mostly absent or treated as a byproduct. A recent ILO study found that, while MOUs may serve as a starting point, a formal bilateral agreement is necessary to address welfare and social security issues to ensure that provident funds and social insurance schemes are well coordinated and weighted appropriately for different jobs and worker contributions, in both receiving and

sending countries.8

With respect to Thailand's MOUs with Cambodia, the Lao PDR, and Myanmar, the study found widespread differences and determined that the MOUs were outdated (ILO, 2015b). Nevertheless, some regional trade union groupings see these MOUs as prototypes that can potentially evolve into a social protection scheme in the longer term. Under the MOUs, authorised agencies in the countries of employment are required to set up a savings fund to which migrant workers contribute 15% of their monthly wages or salary, and the accumulated contributions are refunded to them upon termination of their employment and their return to their home country (ATUC, 2020).

Pursuing specific bilateral social security agreements is not an easy task, as receiving states typically have more developed social protection and social security legislations than do sending states. Sending states are in a particularly weak position, having no leverage to negotiate unless they first improve social protection and security provisions for their own nationals at home. Since reciprocity is part of any bilateral agreement, receiving and sending states may both face political and practical constraints.

3. Timeline of Association of Southeast Asian Nations Efforts to Protect Migrant Worker Rights

- **2003** Association of Southeast Asian Nations (ASEAN) heads of states and governments adopt the Declaration of ASEAN Concord II (Bali Concord II), which envisages the establishment of an ASEAN Community by 2020. The ASEAN Community comprises three pillars: the ASEAN Political-Security Community, the ASEAN Economic Community, and the ASEAN Socio-Cultural Community.
- **2004** The Vientiane Action Programme, 2004–2010, on protecting migrant labour, is adopted in Vientiane, Lao People's Democratic Republic on 29 November 2004. The plan calls for an 'Elaboration of an ASEAN instrument on the protection and promotion of the rights of migrant workers'.
- **2007** The ASEAN Economic Community is established, envisaging full economic integration of the ASEAN Member States (AMS) and calling for the freer flow of skilled labour.

The ASEAN Declaration on the Protection and Promotion of the Rights of Migrant Workers is adopted at the 12th ASEAN Summit in Cebu, Philippines. This declaration takes important steps to move ASEAN towards compliance with existing United Nations conventions and treaties that many AMS have already ratified. It recognises the need to share responsibilities regionally for the protection of migrant workers while also acknowledging both the differences and similarities of AMS concerns. The International Labor Organization established the Triangle Project (ASEAN Migrant Labour Forum) to support the elaboration and implementation of this declaration.

2008 The ASEAN Committee on the Implementation of the Declaration on the

⁸ Some AMS have experience negotiating formal bilateral agreements with countries outside the region. For example, the Philippines already has nine such agreements in place, and Viet Nam is upgrading its national laws to enable it to commence business service agreement negotiations with Germany and Korea.

Protection and Promotion of the Rights of Migrant Workers (ACMW) is established to support the implementation of the Cebu Declaration.

AMS adopts the ASEAN Charter, setting forth a vision of movement of labour within the region. Amongst other things, the charter sets out 'to create a single market and production base which is stable, prosperous, highly competitive and economically integrated with effective facilitation for trade and investment in which there is free flow of goods, services and investment; facilitated movement of business persons, professionals, talents and labour; and freer flow of capital...'

2009 The ASEAN Framework Instrument on the Protection and Promotion of the Rights of Migrant Workers is formally proposed to the ASEAN Senior Labour Officials by the Task Force on ASEAN Migrant Workers, a civil society body comprising trade unions, human rights and migrant rights nongovernment organisations, and migrant worker associations. The group supports the development of a rights-based framework for the protection and promotion of the rights of migrant workers, in line with ASEAN's Vientiane Action Plan.

2012 The adoption of the ASEAN Political-Security Community Blueprint promotes closer cooperation of various ASEAN sectoral bodies to expedite the ASEAN Committee on the ACMW's work in developing an instrument to ensure that the rights of migrant workers are well protected within the region, in accordance with the laws, regulations, and policies of the AMS.

The ASEAN Socio-Cultural Community Blueprint is adopted, providing for the protection and promotion of the rights of migrant workers and other vulnerable groups.

2013 The ASEAN Declaration on Strengthening Social Protection recognises migrants as a vulnerable category of workers.

2015 AMS endorse the United Nations Sustainable Development Goals.

- **2016** The ACMW Work Plan 2016–2020 is adopted. This covers four areas: (i) mechanisms governing labour mobility, such as recruitment, job placement, and reintegration programmes; (ii) social protections for migrant workers in ASEAN; (iii) the protection and promotion of the rights of migrant workers; and (iv) the labour dimension of human trafficking.
- 2017 AMS take a significant step forward in their efforts to protect and promote the rights of migrant workers by signing the ASEAN Consensus on the Protection and Promotion of the Rights of Migrant Workers, which elaborates the rights of migrant workers and their family members, and expands the obligations of AMS, subject to national laws.
- **2018** The ASEAN Action Plan 2018–2025 is adopted to guide the implementation of the ASEAN consensus.

Figure 1.1: An Overview of the Association of Southeast Asian Nations Consensus



Source: United Nations Entity for Gender Equality and the Empowerment of Women (2017), *Women Migrant Workers in the ASEAN Economic Community*. New York, NY: United Nations Entity for Gender Equality and the Empowerment of Women. https://asiapacific.unwomen.org/-/media/field%20office%20eseasia/docs/publications/2017/06/aec-women-migrationstudy.pdf?la=en&vs=4122 (accessed 11 May 2020).

4. Portability of Rights of Migrant Workers

The portability of migrant workers' social security rights remains a major problem within ASEAN (Ong and Peyron Bista, 2015). Trade union organisations and civil society groups in the region have been calling for migration policies and institutions built on the recognition and portability of social protection (ATUC, 2020), and have argued that migrant workers should be able to preserve, maintain, and transfer benefits from one country's social security programme to another, as well as between localities within a country (spatial portability), between jobs, and between members of a household (social portability). Since the right to social protection and security is attached to the migrant worker, this should include the ability to maintain and access the benefits of the social protection system of the country where he or she currently works.

| Country | | Medical care | | Sickness | - | Unemployment | | Old age | | Work injury | | Family | | Maternity | : | Invalidity | | Survivors | National workers abroad? |
|-----------------------|--------------|-------------------|--------------|-------------------|--------------|-------------------|--------------|-------------------|--------------|-------------------|--------------|-------------------|--------------|-------------------|--------------|-------------------|--------------|-------------------|-----------------------------|
| | Nationals | Non- nationals | |
| Brunei Darussalam | \checkmark | √*a | √** | √** | - | - | \checkmark | √* | √** | √** | - | - | -√ | √* | \checkmark | √* | \checkmark | √* | No |
| Cambodia | \checkmark | \checkmark | √** | √** | - | - | \checkmark | \checkmark | \checkmark | \checkmark | - | - | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | No |
| Indonesia | \checkmark | \checkmark | √** | √** | - | - | \checkmark | \checkmark | \checkmark | \checkmark | - | - | √** | √** | \checkmark | \checkmark | \checkmark | \checkmark | Yes |
| Lao PDR | \checkmark | \checkmark | \checkmark | - | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | Yes ^b |
| Malaysia | \checkmark | √*** | √** | √** | - | - | \checkmark | √*** | \checkmark | √*** | - | - | √** | √** | \checkmark | √*** | \checkmark | √*** | No |
| Myanmar ^c | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | - | \checkmark | \checkmark | Yes ^d |
| Philippines | \checkmark | \checkmark | \checkmark | \checkmark | - | - | \checkmark | \checkmark | \checkmark | \checkmark | - | - | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | Yes |
| Singapore | \checkmark | √*e | √** | √** | - | - | \checkmark | \checkmark^* | √** | √** | - | - | √** | √** | \checkmark | √**6 | \checkmark | √**f | Yes ^g |
| Thailand | \checkmark | √ ^h | \checkmark | \checkmark | Yes |
| Viet Nam ⁱ | \checkmark | \checkmark | \checkmark | \checkmark | \checkmark | - | \checkmark | \checkmark | \checkmark | \checkmark | - | - | \checkmark | \checkmark | - | - | \checkmark | \checkmark | Yes |

Table 1.2: Social Security Coverage of Migrant Workers by Country and Branch, 2017

Lao PDR = Lao People's Democratic Republic.

Note: * permanent residents only, ** employer liability, *** separate scheme.

^a Universal coverage – permanent residents; employer liability – other migrant workers (insurance-based).

^b Via a labour fund.

^c Not applicable to establishments with fewer than five employees. Such employees can register voluntarily.

^d Voluntary contribution possible.

^e For those who are not permanent residents, employer-based or employer-insured provision is available.

^f This includes foreign domestic workers as well as work injury-related cases, through mandatory insurance coverage (feedback from the Senior Labour Officials Meeting Singapore, July 2017).

^g The Government of Singapore provides for continued contributions to Medishield Life, even while overseas.

^h Undocumented non-nationals (except those who have completed the national verification process) are covered under a separate scheme.

¹ 2018 position indicated here: from 1 January 2018, regular migrant workers are covered by compulsory social insurance.

Source: Olivier, M. (2018), Social Protection for Migrant Workers in ASEAN: Developments, Challenges, and Prospects. Geneva: International Labor Organization.

However, in the absence of bilateral agreements between sending and destination countries, these portability rights remain unrecognised (Tamagno, 2008). A large majority of migrant workers are unable to enrol in either their own national social security systems or that of the host country, or to transfer the accrued contributions or entitlements between social security systems. Migrant workers are often doubly disadvantaged because they receive less social protection both at home and in their host country.⁹ They are also often excluded from tax-financed schemes such as social assistance programmes or social pension schemes despite contributing to the host country economy through work, consumption, and taxation. Documented workers are only entitled to a few benefits from their host country, depending upon their immigration status category (Olivier, 2018). For example, in Singapore and Brunei Darussalam, participation in the national provident funds is possible only for workers with permanent residence status (i.e. documented or registered persons) (UN Women, 2019; Pasadilla and Abella, 2012).

To compensate for this lack of protection for its large labour population working abroad, the Philippine Overseas Workers Welfare Administration operates schemes that cover overseas workers for invalidity and death risks, in which they can optionally enrol under the Philippine Social Security System.

Indonesia has one of the world's largest migrant worker communities.¹⁰ In 2016, an estimated 9 million Indonesians were working abroad, the majority of whom were undocumented, and of whom half were women, mostly employed in the informal sector as domestic workers (Arisman and Jaya, 2020).¹¹ Acknowledging the plight of Indonesian migrant workers, in 2017, Indonesia enacted a new law on the protection of Indonesian migrant workers (Law No. 18/2017). In line with the 2019 Recommendations of the High-Level Political Forum, the new law reinforced policies to provide end-to-end protection to overseas workers; however, this new provision has not yet been implemented (Global Forum on Migration and Development, 2019).

With respect to health coverage, health insurance for migrant workers is mandatory in Malaysia, Singapore, and Thailand, the main migrant-recipient AMS. However, Thailand's tax-financed Universal Coverage Scheme excludes migrant workers, ethnic minorities, and displaced or stateless persons who lack a national identity document. Arisman and Jaya (2018) suggest that around half of migrant workers from Cambodia, the Lao PDR, and Myanmar working in Thailand who do not qualify for government welfare systems were obliged to take up the Compulsory Migrant Health Insurance to access public healthcare facilities because of their lack of formal immigration status. Several media sources cited

⁹ See also the ILO Maintenance of Social Security Rights Convention, 1982 (No. 157).

¹⁰ Since 2003, the Indonesian civil society organization (CSO) Migrant Care has been helping villages hold safe migration workshops for locals about to move abroad for work, providing assistance with contracts and information about their legal rights in their destination countries and where to get help if they find themselves in trouble. Migrant Care was one of several CSOs who lobbied the government for the enactment of this new law in 2017. Although not perfect, this law provides an umbrella for further elaboration and enforcement. During the COVID-19 pandemic, this law proved useful for migrant workers returning home to seek relief support from the Government of Indonesia.

¹¹ See the authors' interview with Dr. Elcid Li on 18 August 2020. Discussions with Dr. Dominggus Elcid Li is the Executive Director of IRGSC (Institute of Resource Governance and Social Change) (2013-to current), an Indonesian think tank based in Kupang, East Nusa Tenggara, Indonesia.

in this report report that such workers have faced challenges in quickly accessing welfare services, particularly in the early stages of the COVID-19 crisis.

In Malaysia where public healthcare providers are tax-financed, the Foreign Workers Health Insurance Protection Scheme involving private medical insurers was implemented between 2011 and 2013 to reduce the government's subsidisation of migrant workers' healthcare. Similarly, Singapore obliges employers to purchase private insurance for nonpermanent resident migrant workers and to bear any excess medical expenses. Migrant workers who are permanent residents in Singapore are covered under the compulsory medical savings and opt-out insurance schemes, but receive fewer healthcare subsidies than Singapore citizens. In 2015, Brunei Darussalam introduced mandatory take-up of private insurance for non-permanent resident migrant workers.

Given the imminent AEC integration, decent work conditions for migrant workers and a multilateral social security agreement as proposed in the 2007 ASEAN Declaration on the Protection and Promotion of the Rights of Migrant Workers (or the Cebu Declaration on Migrant Workers) would be indispensable to manage intraregional migration responsibly.

Box 1: Social Protection Floors

At the 23rd ASEAN Summit in Brunei Darussalam in October 2013, the 10 Association of Southeast Asian Nations (ASEAN) leaders adopted a Declaration on Strengthening Social Protection, reaffirming their commitment to build a 'socially responsible and people-oriented' ASEAN community by establishing nationally defined social protection floors (SPFs) for all (Ong and Peyron Bista, 2015). The declaration reflects a growing regional consensus that the establishment of SPFs is fundamental to reduce poverty and inequality and promote inclusive and sustainable growth.

In support of this, a more focused 3-year programmatic project was launched in 2016, with support from the Government of Japan, in Indonesia and Viet Nam. Its objectives were increased social security coverage for workers in ASEAN Member States, and greater coverage of social security schemes, particularly amongst informal workers and migrants, through improved policies, legal frameworks, and enforcement and delivery mechanisms in Indonesia and Viet Nam (International Labor Organization, 2016).

Following the adoption of the 2016 ASEAN Consensus on Migrant Workers, despite differing levels of coverage and speeds of change, most countries are now moving towards establishing sound policy and institutional frameworks to deliver social protection effectively and efficiently. Based on the principle of universal protection, SPFs are an investment with both immediate and long-term effects on millions of lives, including enhanced political stability and social cohesion. Social protection and SPFs also contribute to economic growth by supporting household incomes and thus domestic consumption. They also enhance human capital and productivity, and empower people to find decent jobs. Consequently, they are a critical policy tool for supporting transformational national and regional development and the formalisation of economies.

https://www.ilo.org/wcmsp5/groups/public/---ed_norm/---

relconf/documents/meetingdocument/wcms_531409.pdf (accessed 8 June 2020).

Sources: Olivier, M. (2018), Social Protection for Migrant Workers in ASEAN: Developments, Challenges, and Prospects. Geneva: International Labor Organization (ILO); and ILO (2016), Building an Inclusive Future with Decent Work: Towards Sustainable Development in Asia and the Pacific. Report of the Director General to the 16th Regional Meeting of Asia and Pacific, Held in Indonesia. Geneva: ILO.

5. Association of Southeast Asian Nations Commitment to International Treaties

Over the years, AMS have failed to ratify or acknowledge several important international labour and human rights instruments, with some notable exceptions, such as the 1966 UN International Covenant on Economic, Social and Cultural Rights. To date, no AMS has ratified the Social Security (Minimum Standards) Convention, 1952 (No. 102), which is the main ILO social security instrument.

As of early 2020, only two countries, Indonesia and the Philippines, had ratified the global Convention on the Protection of the Rights of All Migrant Workers and Members of Their Families, which states that migrant workers' rights are to be respected without undermining state laws.¹² The convention recognises several different groups of migrant workers (frontier, seasonal, self-employed, seafarer, and itinerant), and comprises nine parts: scope and definitions; non-discrimination with respect to rights; human rights of all migrants; other rights of documented or regular migrants; provisions applicable to particular categories of migrants; the promotion of sound, equitable, humane, and lawful conditions in connection with international migration; the application of the convention; general provisions; and final provisions.

Even when ratified, the norms and standards embedded in these instruments have been poorly implemented. In fact, no bilateral social security agreements exist between AMS, and the level of ratification of international standards is not uniform, resulting in a persistent lack of adequate instruments or appropriate enforcement mechanisms that jeopardises the 'greater integration of the ASEAN region'.

¹² On 18 December 1990, the UN General Assembly adopted a resolution guaranteeing dignity and equality to migrant workers in an era of globalisation. The Committee on the Protection of the Rights of All Migrant Workers and Members of Their Families is a mechanism to monitor and measure the national agencies and actors responsible for implementing the International Convention on the Protection of the Rights of All Migrant Workers and Members of Their Families articles. For an update and commentary on the status of the enforcement of this convention in ASEAN, please see Olivier (2018). The fact that only two AMS have ratified the convention is unsurprising since AMS have also failed to endorse many other similar international conventions and treaties. They likely see these conventions as a needless burden to carry and prefer to handle domestic and regional migrant workers issues in a more informal, self-regulated, and at times arbitrary manner subject to national priorities or even prejudices. In general, AMS do not appear eager to sign conventions that they do not wish to implement.

| Country | | Migrant workers' rig | ghts | Equal treatment in social protection | | | | | | |
|-------------|---------------------|----------------------|----------------------|--------------------------------------|-----------------------|-----------------------|--|--|--|--|
| | Migration for | Migrant Workers | International | Equality of | Equality of | Maintenance of Social | | | | |
| | Employment | (Supplementary | Convention on the | Treatment | Treatment (Social | Security Rights | | | | |
| | Convention | Provisions) | Protection of the | (Accident | Security) Convention, | Convention 1982 (No. | | | | |
| | (Revised), 1949 | Convention, 1975 | Rights of All | Compensation) | 1982 (No. 118) | 157) | | | | |
| | (No. 97) | (No. 143) | Migrant Workers | Convention, | | | | | | |
| | | | and Members of | 1925 (No. 19) | | | | | | |
| | | | Their Families, 1990 | | | | | | | |
| Brunei | | | | | | | | | | |
| Darussalam | | | | | | | | | | |
| Cambodia | | | 2004 ^(c) | | | | | | | |
| Indonesia | | | 2012 | 1950 | | | | | | |
| Lao PDR | | | | | | | | | | |
| Malaysia | 1964 ^(a) | | | 1964 | | | | | | |
| Myanmar | | | | 1927 | | ••• | | | | |
| Philippines | 2009 ^(b) | 2006 | 1995 | 1994 | 1994 ^(d) | 1994 | | | | |
| Singapore | | | | 1965 | | | | | | |
| Thailand | | | | 1968 | | | | | | |
| Viet Nam | | | | | | | | | | |

Table 1.3: Ratification of Conventions Related to Migrant Workers by Association of Southeast Asian Nations Member States

...: Convention not ratified, Lao PDR = Lao People's Democratic Republic.

^a Only Malaysia-Sabah (also excludes the provisions of Convention No. 97, Annexes I–III).

^b Excludes the provisions of Convention No. 97, Annexes II and III.

^c Year of signature (yet to be ratified).

^d Includes branches a–g only.

Source: Ong, C.B. and C.P. Bista (2015), *The State of Social Protection in ASEAN at the Dawn of Integration*. Bangkok: International Labour Organization Regional Office for Asia and the Pacific. https://www.ilo.org/wcmsp5/groups/public/---asia/---ro-bangkok/---ilo-jakarta/documents/publication/wcms_428982.pdf (accessed 11 August 2020); reproduced from International Labour Organization and Asian Development Bank (2014), ASEAN Community 2015: Managing Integration for Better Jobs and Shared Prosperity, pp.83–100. Bangkok; Manila: International Labour Organization and Asian Development Bank. *p. 95.*

ASEAN has made impressive progress over the decades, both as a cohesive regional grouping and in the way it has managed to improve the living conditions of its populations steadily through policies of peaceful development. However, the treatment of migrant workers in the region still leaves much to be desired. This is partly for economic reasons as businesses in receiving countries want to spend as little on labour as possible. Another reason is political, as governments worry about how domestic populations would respond to large influxes of migrant workers from neighbouring nations. Nevertheless, in a globalised economy in which the region has taken advantage of low-cost labour to boost its own growth, it is necessary to follow global standards of humane treatment by providing social protection and investing in the welfare of the most vulnerable sections of the population.

Chapter 2

Impact of the Coronavirus Disease Pandemic

Southeast Asia was affected early by the COVID-19 pandemic because of its close geographical proximity, business, travel, tourism, and supply chain links to China. However, relative to other regions of the world, the number of confirmed cases in the ASEAN region is much lower and the rate of increase is also slowing. As of 22 October, there were 42,138,128 confirmed COVID-19 cases worldwide (Worldometer, 2020;) and over 869,544 cases in Southeast Asia (Chua, 2020).

By October 2020, many countries in the region such as Malaysia, Singapore, Thailand, and Viet Nam, while still very cautious, had started relaxing their nationwide lockdowns or community quarantine measures, and economic activities had begun to resume. Travel restrictions and quarantine protocols for foreign visitors and incoming labourers are still being implemented.

Even prior to the COVID-19-induced crisis, Southeast Asian economies were already struggling with their growth figures for a variety of reasons, but mainly because of global trade tensions and a general economic slowdown across the continents. It is expected that the region will be severely impacted by a global recession and disruption of economic activities in several key economic sectors after the pandemic.

In response to the crisis, most governments in the region have introduced fiscal stimulus packages mobilising both fiscal and monetary measures to mitigate the economic impact, but the impact of these measures is yet to be seen. ASEAN has also called for collective action by leveraging technology and digital trade and setting up trade facilitation platforms to foster supply chain connectivity and sustainability (ASEAN, 2020a). As part of the Declaration of the Special ASEAN Summit on Coronavirus Disease 2019, AMS leaders also proposed a post-pandemic recovery plan and a COVID-19 ASEAN Response Fund (ASEAN, 2020d).

Overall, however the pan-ASEAN lockdown and curfew has rapidly transformed a public health crisis into an economic and social crisis, and to some extent a mild political crisis too. Intra-regional migrant workers were amongst the hardest hit segments of society, and faced extreme social and economic distress. The spread of COVID-19 has highlighted the weaknesses of ASEAN's national healthcare and social welfare systems, in addition to challenging the labour recruitment and employment practices implemented in region over the years. While informal labourers have become increasingly important for many different economic activities, the pandemic exposed their poor living and working conditions, and their migration status impacted their well-being during the crisis. This is discussed in greater detail below.

1. Impact on Association of Southeast Asian Nations Economies¹³

More broadly, the COVID-19 lockdown measures worsened the economic slowdown of 2018–2019. The crisis has interrupted global value chains, exacerbating the fall in global economic activity and disruption to international trade that was already weakened by the 2019 trade tensions between the US and China. Many migrants have already lost their jobs, are projected to lose their jobs, or are expected to be forced to accept lower wages due to lockdowns or oil price crashes in their destination countries. They may also be prevented from sending remittances as a result of stringent movement restrictions and the exclusion of money transfer service providers from the list of essential services. Furthermore, many intended migrants who had been preparing to depart their home countries have been forced to change their livelihood plans for the coming years. In summary of this situation, the IMF reported in 2020 that 80% of the world's total remittances flowed to low- and middle-income countries in 2019. Elaborating on this, the IOM (2020a) reasoned that the negative impacts of the COVID-19 outbreak may be more serious in developing countries whose citizens depend heavily on remittances from migrant family members. Reports published by multilateral organisations in April-May 2020 (IMF, 2020; World Bank, 2020b; Villafuerte and Takenaka, 2020) concluded that the ASEAN region may experience a decline in remittances, tourism, and foreign direct investment. However, reports in October 2020 suggest that remittances are gradually recovering in countries like the Philippines (Lopez, 2020).¹⁴ However, analysts are reluctant to accept this as a sure sign of recovery, as such positive signals may capture a short-term effect of the pandemic on remittances while the negative impact on declining income and remittances, if any, on household welfare may be more serious in the longer term. Therefore, as Murakami, Shimizutani, and Yamada (2020) suggest, it may be necessary to take a more nuanced approach to the use of data on international restrictions on travel and remittance transactions.

An online report published by the Center for Strategic and International Studies (CSIS) (2020) projected that the economic impact of COVID-19 in ASEAN will be huge, on par with the fallout of the 1997–1998 Asian financial crisis, or perhaps much greater. The IMF projects that ASEAN+5 (Indonesia, Malaysia, the Philippines, Thailand, and Viet Nam) will see their economies shrink by 0.6 percentage points in 2020, in contrast to its prepandemic forecast of 4.8% growth. A World Bank report (2020a) includes both a baseline and a more pessimistic scenario, with the 'worst-case' forecast projecting that the

¹³ A crisis of this nature generally leads to 'distress' sales of land and assets amongst poorer households. At present, no data are available on distress sales of land and assets by migrant worker households, if any, from labour-sending nations in the region. However, in response to COVID-19 concerns, real estate developers and architects will likely reverse the trend of densification and open-plan layouts of urban and peri-urban areas. Public health officials may begin to amend building codes to limit the risk of future pandemics, and may seek more land for built-up areas in the future. In addition, the return of poorer migrant workers to their villages may increasingly compel them to look for cash, leading to the sale of land and assets. Moreover, demand for land is likely to increase, especially in rural areas where more space is available.

¹⁴ The post by CNN cited here suggested that remittances to the Philippines totalled \$3 billion in July 2020, despite the pandemic. Land-based Filipino workers remitted more funds than in 2019, while those working at sea remitted less. Local analysts suggested a 10%–12% decline in 2020; however, these projections were described as lacking evidence as to the likelihood of a recovery.

economies of the major developing ASEAN countries will contract by 0.5–5.0%, with the exception of Viet Nam, which has maintained positive (+1.5%) growth. The global collapse in oil prices is expected to have a sharp impact on economies dependent on fuel exports, in particular Indonesia, where coal and oil comprise nearly 25% of exports; Malaysia, where oil and gas make up about 16% of exports; and of course Brunei Darussalam, where crude and natural gas account for over 90% of all exports (Searight, 2020). The COVID-19 pandemic has also drastically reduced global demand for oil, especially in the transportation sector, and it is unclear to what degree low oil prices will impact ASEAN economies (Taghizadeh-Hesary, 2020; Arezki and Nguyen, 2020).

1.1. Supply Chains

The fallout of COVID-19 has already affected demand and supply chains in both food and non-food sectors because of the contraction of production and distribution networks, and has also impacted the service sectors and commodity exports. With around 40% of its exports relying on global value chains, and its strong linkages to multiple nodes, the region is highly exposed to supply-chain risks (United Nations Economic and Social Commission for Asia and the Pacific [UNESCAP], 2020c). Amongst the ASEAN countries, Malaysia, Singapore, and Thailand are heavily integrated into regional supply chains and have been most severely affected by reduced demand for goods produced therein (Menon, 2020). Indonesia and the Philippines, which have been increasing supply chain engagement, are similarly threatened. In Viet Nam, which has maintained a robust supply chain with China, movement restrictions from late February 2020 created severe supply disruptions that later spread to the Lao PDR, Myanmar, Thailand, and across the region (CSIS, 2020).

Although the global food supply chain remained intact until June, some reports indicate that since mid-June COVID-19 restrictions on movement and activities have affected all segments of food supply chains, simultaneously impacting farm production, food processing, transport, and logistics.¹⁵ Indonesia, Malaysia, and Thailand have reported that farm production has been afflicted by various bottlenecks, with limits on the mobility of people reducing the availability of seasonal workers for planting and harvesting. With travel restrictions continuing, there is also concern over seed and other input shortages. The supply chain disruptions are likely to exacerbate food insecurity in remote places in the Lao PDR, Myanmar, and parts of Indonesia (Food Industry Asia, 2020; Food and Agriculture Organization of the United Nations, 2020a; Kovac, 2020).

1.2. Impact on the Service Sector

In countries like Indonesia, Malaysia, Thailand, and Viet Nam, the tourism sector, which employs a large share of migrant workers, has been badly impacted. In the case of Myanmar, a post-conflict nation in transition, the IMF reasoned that its six growth engines would be affected (De and Nadeem, 2020). Since late March 2020, governments across the region have imposed restrictions on large-scale events, restaurants, and leisure activities, in addition to school closures. These curfew and lockdown measures, although

¹⁵ Some farm subsectors are dependent on seasonal labour, and certain operations, such as vegetables and fruit, are particularly labour-intensive.

differing in scale and scope across the region, have slowed or even stopped economic activities in certain sectors, and had a ripple effect on others (ASEAN, 2020m; Hayat, 2020; Agarwal, 2020). Several migrant workers from Myanmar engaged in the tourism sector in Rayong, Kanchanaburi, and Chiang Mai, Thailand reported non-payment of wages since April 2020; and social benefits, however meagre, were available to only a few (Kaicome, 2020; Sandar, 2020; O'Connor, 2020; Bangkok Post, 30 March 2020). Indonesian workers engaged in the service sector in Malaysia reported similar wage cuts in April 2020. According to reports, more than 2.8 million domestic Indonesian workers had lost their jobs by the end of April 2020, and another 1 million workers (both domestic and returning migrants) were furloughed and placed on paid or unpaid leave (Rahman, 2020).

1.3. Manufacturing

With respect to manufacturing, MSMEs, which tend to have fewer internal resources and more limited access to information, are likely to be particularly affected. This stress is likely to impact economically vulnerable communities disproportionately, as these people are much less likely to be employed in large enterprises. Surveys suggest that roughly two-thirds of ASEAN MSMEs had less than 2 months' worth of cash reserves left in early April, and more than one-third expected to lay off over 40% of their staff (Organisation for Economic Co-operation and Development [OECD], 2020a). However, ADB has also pointed out that ASEAN's resilient nature should help the region overcome the economic impact once the pandemic is contained. ASEAN should ensure that containment measures to rejuvenate the economy cover all parts of society, including marginalised and vulnerable communities (ADB, 2020a; Fernandez, 2020; Virgil and Lie, 2020)



Figure 2.1: Estimates of Gross Domestic Product Loss in the Association of Southeast Asian Nations

Lao PDR = Lao People's Democratic Republic.

Source: World Bank (2020), *East Asia and the Pacific in the Time of COVID-19 – Regional Economic Update.* Washington, DC: World Bank. April. <u>http://hdl.handle.net/10986/33477</u> (accessed 24 May 2020).

Box 2: Association of Southeast Asian Nations Responses and Reactions

In the Association of Southeast Asian Nations (ASEAN), first responses to the coronavirus disease (COVID-19) pandemic have included scattered national policy initiatives leading to severe limitations on freedom of movement. This has also resulted in the increased concentration of power in the hands of executive branches, which have largely turned to govern by decree (or executive order) to legislate curfews and lockdowns.

However, there is considerable variation in the ways that ASEAN governments have responded to the pandemic, as they had to navigate the costs of each measure according to their domestic social, demographic, and economic realities. One major difference can be seen in the speed of response, ranging from Viet Nam's prompt action to monitor and contain the pandemic, to the Philippines and Indonesia's long 'phase of denial' (Dabla-Norris et al., 2020; Abuza, 2020; Lindsey and Mann, 2020). Another major difference has been in the aggressiveness of the response. Some countries have implemented full or partial lockdowns, such as the Philippines' lockdown of Manila, curfew in Thailand, and Malaysia's Movement Control Order. Other countries have opted for less strict measures, such as Myanmar's targeted partial lockdowns. With local economies likely to stumble, most ASEAN countries' initial instinctive reaction to the COVID-19 crisis has been to look inwards and act alone. As a result, borders were abruptly closed, supply chains disrupted, and regional activity virtually came to a halt for a period of time.

As in other parts of the world, several ASEAN governments presented their dilemma as being between stopping the spread of COVID-19 and saving jobs. In some countries, the consequences of initially excluding marginalised groups from COVID-19 response measures, or paying less attention to them, has been dire. For example, in Singapore, the second wave of the virus was linked to packed dormitories for migrant workers, who were apparently not granted access to the same degree of medical care and contact-tracing measures as the general population (Noel, 2020; Karmini, 2020; Lindsey and Mann, 2020).¹⁶ Such measures have invariably impacted migrant workers in their home and host countries in different ways.

Even where commendable ambitious national responses were implemented, it has become clear that the immediate needs of the broader economy and migrants were not addressed, and most workers were allowed to return home without adequate medical screening or advisory support. This ambiguous response from host nations led to mild discontent amongst labour-sending nations, and there is uncertainty as to whether workers will return when normalcy is reestablished.

During the pandemic's initial stages, many host governments also lacked clarity as to whether migrant workers could access health and welfare systems in their host countries (Global Forum on Migration and Development, 2020). Civil society networks were sceptical of government assurances of material support, and screening and testing, which were viewed with suspicion (ATUC, 2020). Meanwhile, police forces were deployed in several cities and along travel routes across the region to ensure 'law and order', ignoring the public health nature of the pandemic (Migrante International, 2020; Human Rights Watch, 2020b; Palatino, 2020; World Law Group, 2020). At the same time, governments in origin countries (e.g. Singapore and Thailand) also made a feeble attempt to halt migrants from returning home, ostensibly for fear of further spreading the virus (ILO, 2020f; ILO, 2020g; ILO, 2020m; Yee, 2020; Gu, 2020).¹⁷

¹⁶ Although workers were paid during the lockdown, continued uncertainty regarding their jobs and the possibility of returning home affected their movement and mental health.

¹⁷ For example, the Government of Myanmar initially resisted opening checkpoints for workers to return home from Bangkok.

While the ASEAN nations all provided relief packages to mitigate the negative impact of the pandemic, the delivery and impact of such efforts remains to be seen. The relief packages were provided independently, without much coordination across the ASEAN countries to ensure that they mitigated the sufferings of migrant workers. If there had been more coordination in the earlier stages of the pandemic, the relief packages could have been better targeted to mitigate the pandemic's negative effects, both within and across countries. Kimura et al. (2020) reason that a coordinated policy response would have been best. However, as Djalante et al. (2020) highlighted, such an effort requires political will to an extent that remains challenging. Bilateral and multilateral diplomacy was instituted late in Southeast Asia: ASEAN ministers only began to confer virtually in mid-April 2020, and ASEAN leaders held a summit in June 2020 to provide a framework for coordinated action ('ASEAN Holds Special Summit', 2020, 2020; ASEAN, 2020k). Prior to this, coordination amongst ASEAN governments remained elusive and was frequently scattered, impacting supply chains, migrants, and the regional economy.

Source: Authors.



Figure 2.2: Coronavirus Disease Containment and Mitigation Measures in Southeast Asia

Source: United Nations Economic and Social Commission for Asia and the Pacific (2020), *Policy Brief: Impact of COVID-19 on Southeast Asia*. July. Bangkok: United Nations Economic and Social Commission for Asia and the Pacific.

2. Immediate Impact on Migrant Workers¹⁸

In the early days of the pandemic, particularly between March and mid-June 2020, the shutdown of economic activities and curfew imposed on the movement of persons left migrant workers with the choice of either starving in the countries where they had been working, with no job or money and unpredictable access to healthcare and welfare systems; or return to their home countries (United Nations Capital Development Fund, 2020).¹⁹ Despite facing the same poverty that compelled them to leave in the first place,

¹⁸ The ASEAN Trade Union Council periodically disseminates information through its website on the status of migrant workers in the region (www.aseantuc.org).

¹⁹ Before the onset of COVID-19, the major destination countries for Myanmar's migrants were China, Japan,

media reports and various government statements suggest that many believed they would have a better chance of survival amongst familiar surroundings and familial networks in their countries of origin. Highlighting the plight of refugees, who are also mostly irregular migrant workers in their host countries, analysts argued that 'services dealing with migration, both governmental and non-governmental, are currently heavily stretched in terms of money, manpower and facilities leaving not only migrants but also refugees scrambling for advice on the pandemic and immigration services'.²⁰

2.1. Closed Borders

When Thailand announced measures in March to close their land borders temporarily, thousands of jobless migrant workers streamed over the borders to their homes. However, for some of these, 'home' means one of nine camps along the Thai–Myanmar border that shelter more than 90,000 people (Nanthini, 2020; Engblom, Lephilibert, and Baruah, 2020; Asadullah, 2020; Bismonte, 2020). Current border control measures have limited their freedom of movement across the border and left them unable to travel for informal labour or to return home (to Myanmar, in this case), while political uncertainty persists. This has eroded their income and left them almost entirely dependent on external humanitarian assistance. Not satisfied with living in the refugee camps for long, they will often take up jobs below their skill level in their host countries. Thus, despite their credentials, they are unable to integrate easily into their host countries.

The suspension of public transport services and restrictions placed on the movement of private vehicles drove tens of thousands of migrant workers to return to their hometowns on foot. Others decided to travel by road using local vehicles, such as cargo vans, and crossed international checkpoints in small groups. Thailand reported large movements of migrants out of Bangkok and other metropolitan areas into Cambodia, the Lao PDR, Myanmar, and other provinces during the first week of April (ILO, 2020o).

2.2. Rising Xenophobia

The lack of coronavirus testing during the early stages of the pandemic and the virus's long incubation period made it difficult to estimate how many migrant workers left their host countries (e.g. Brunei Darussalam, Malaysia, Singapore, or Thailand) carrying the COVID-19 virus. What is known, however, is that after the migrant workers returned to Myanmar, the Lao PDR, and Cambodia, those countries experienced spikes in COVID-19 cases, exacerbating the strain on their limited health resources. As a result, local media and governments claimed that they had kept their countries safe from the pandemic, but that returning migrants 'brought the virus and spread it' (Nyein, 2020; Weng, 2020; Htet, 2020). To address this, precautionary quarantine measures were implemented, but little

Korea, Malaysia, Saudi Arabia, Singapore, and Thailand. At least 71,000 migrants are thought to have returned to Myanmar by the end of May 2020, primarily from Thailand and China. However, the final figure is uncertain because there were many informal returnees. Returning migrants are required to quarantine for 21 days in community-based quarantine facilities, and a further 7 days at home before they can rejoin their communities.

²⁰ Although refugees are in a different category from migrants, they tend to work in their host countries, as in Thailand, often without formal documentation.

effort was made to support workers returning home or to prevent the spread of wrong or exaggerated information on the spread of virus via returning workers. This led to occasional acts of violence and discrimination linked to the virus and migrant workers. Acknowledging this situation, in May 2020, the UN Secretary General issued a statement appealing to governments to prevent the spread of the 'virus of hate' (Human Rights Watch, 2020a).

In the middle of May 2020, a surprising second wave of coronavirus infections emerged in several ASEAN countries, and continued for a while. In September 2020, Viet Nam reported around 2,000 new infections in Danang province, indicating a sudden spike; meanwhile, the Philippines reported around 1,800 new cases per day (on average) while Indonesia reported 1,400. In contrast, in the same month Singapore only witnessed around 533 new infections, of which about 400 occurred in migrant dormitories (Singapore Ministry of Health, 2020). Following the political turmoil in Thailand and Malaysia in September–October, a higher number of new infections was reported: Sabah in Malaysia reported 2,600 new infections, second only to Kuala Lumpur. Periodic spikes in Thailand, Malaysia, and Singapore were attributed to the conditions of migrant worker dormitories, a fact that was overlooked in the initial stages. For example, while Singapore was originally hailed for its quick COVID-19 response, by May-June blind spots had emerged in its execution (Bismonte, 2020; Ng, 2020; Minter, 2020). As of June 2020, the country had more than 1,351,800 foreign workers in a population of 5.6 million, with lowskilled migrant workers making up roughly 40% of the total (Singapore Ministry of Manpower, 2020). Most COVID-19 cases occurred amongst these low-skilled migrant workers living in crowded foreign worker dormitories (Koh, 2020).

On 18 March, Malaysia issued a Movement Control Order, which restricted the movement of people. While some relaxations were allowed in May 2020, several trade unions have criticised this order for placing migrant workers in a precarious situation involving not only job losses but also the inability to move, making their futures uncertain. The ILO (2020o; see also ILO, 2020j) reported that, as of June, almost 50,000 Thai and 100,000 Indonesian workers had been repatriated from Malaysia. The Malaysian Trade Union Congress reported that in March-June 2020 there were instances of unfair termination of employment, unpaid wages, workers forced to live in poor conditions, employers forcing workers to continue working in nonessential jobs, and uncertainty surrounding workers' employment status, as governments frequently modified their migration policies. Commentators and civil society organisations (CSOs) feared that migrant workers would be overlooked in COVID-19 response programmes supporting people's access to spaces for isolation and decent housing and living conditions, such as dormitories. On this subject, the Malaysian Bar, a coalition of 62 local CSOs and many other groups issued statements following the large-scale arrests of irregular migrant workers on 1 May 2020. In addition, a statement issued by the UN Resident Representatives' Office in Malaysia noted that fear of arrest and detention may drive migrant groups further into hiding and prevent them from seeking treatment ('Sacked and Abandoned', 2020; Sri Priya, 2020; Thomas, 2020; Bismonte, 2020).

2.3. Chaotic Exodus

This situation suggests that, although Malaysia, Singapore, and Thailand (the main worker-receiving countries in the region) have robust domestic welfare and health care systems owing to their investment in social protection and health security over the years, their quick shutdown and restrictive migration policies have led to a sudden movement of large numbers of people and contributed to the spike in COVID-19 cases in the region (Yayboke, 2020). This precipitous flight also caused tremendous economic stress to the already traumatised migrant population, depriving them of the economic resources they would have had if given more time to evacuate, and making them even more susceptible to COVID-19. While the banning of cross-border travel across all countries in the region may have served domestic ends, it harmed the greater ASEAN region, particularly in its execution, by displacing thousands of migrant workers who hurriedly returned home without adequate testing, health protection (such as face masks), or food support, triggering concern in their home countries as well (Taskforce on ASEAN Migrant Workers, 2020).

3. Medium and Long-Term Impact on Migrant Workers

Several studies that investigated the impacts of the COVID-19 outbreak on regional migrant workers generally examined the prevalence of migrant workers in two types of jobs: (i) jobs in essential industries and occupations where workers are unlikely to lose their jobs but are more likely to face health risks; and (ii) jobs in nonessential industries and occupations that cannot be done from home in which workers are more likely to lose their jobs (Djalante et al., 2020; ILO, 2020i; OECD, 2020b; ILO 2020q).²¹

Although regional migrant workers have supported local businesses for decades by filling domestic labour shortages, they were amongst the first to lose their jobs as employers downsized their work forces. Initial projections suggested that close to one-third of those in the construction sector and more than 50% of those in the tourism sector temporarily lost their jobs during March–June. The lockdowns also impacted the domestic labour market. Country-level assessments (e.g. ILO, 2020g on Malaysia dated 8 May 2020; ILO, 2020l on Myanmar dated 22 June 2020; ILO, 2020o on Thailand dated 3 July 2020) and regional reviews (CSIS, 2020) highlighted that the construction and tourism sectors in ASEAN were massively hit during March–June, and most regional migrant workers were unsure whether their employment would continue.

²¹ ADB (April 2020) estimated that employment in Asia and the Pacific will fall by as much as 167 million jobs in 2020 should containment measures last 6 months from when the outbreak first intensified in the region. In turn, wage incomes in the region are projected to fall from \$359 billion to \$550 billion. Migrant workers, many of whom have limited job security and access to social assistances, are amongst the hardest hit groups as workplace closures and border control restrictions are being put in place to abate further outbreaks. The crucial remittances they send home to their families are expected to decline dramatically. In understanding job and income losses in ASEAN (estimated to be around \$300 billion and 90 million jobs lost between February and December 2020), a proportional estimate for the region will be helpful. The ILO's initial assessment in May 2020 indicated that close to one in six young people is likely to lose their job due to COVID-19 (ILO Newsroom, 2020). This estimate was marginally revised in September 2020 but job losses on a broader scale were retained.
The Government of Malaysia estimated that every 10 new migrant workers helped create five new jobs for locals and increased GDP by 1.1%. Without these migrant workers, many MSMEs – particularly in the plantation, construction, and service sectors – were likely to face long closures (Asadullah, 2020).

Since April 2020, the ILO has published several country briefing notes, a regional briefing report, and country reports documenting the experiences of migrant labourers during the ongoing crisis. These reports indicate that the consequences of the crisis on intra-ASEAN migrant workers have been harshest in Malaysia, Singapore, and Thailand where those employed in construction, services, and manufacturing were the first to lose their jobs. Countries like Indonesia, Myanmar, and the Philippines, which supply migrant workers for sectors such as fishing and shipping, have also been affected, with boats and vessels being grounded in early April 2020 (Engblom, Lephilibert, and Baruah, 2020; Manlangit, 2020).²²

The Migrant Working Group, a member-based organisation that advocates for the rights of migrant workers in Thailand, estimates that as many as 700,000 migrants had lost their jobs by the end of March ('Migrant Workers on Virus Front Line', 2020; Knight, 2020). These workers were reportedly in a very grim situation by May as they had difficulty finding new jobs, limited options to return home, and very little access to government aid in Thailand. Many of these workers were women employed in either the domestic or tourism sectors. Later reports estimated that almost 30% of the 21 million migrant workers had lost their jobs since March 2020, and may have been unable to return to work until at least the end of 2020 (Fernandez, 2020).

The impact of the COVID-19 crisis has been greater on certain sub-groups of migrant workers, such as women, indigenous peoples, older people, and irregular migrants, all of whom face systemic barriers to integrating themselves into the formal labour market. One in every five young people in the ASEAN region is expected to be unemployed in 2020, twice the overall rate and three times the adult rate (Fernandez, 2020). In addition, 20% of young people aged 15–24 are neither studying nor working. This proportion is even higher for women, at nearly 30%. More than 80% of women in the domestic sector are employed informally, and many women are exposed to gender-based violence as a result of the confinement measures (UNESCAP, 2020c; ILO, 2020j).

Many of those fortunate enough to remain employed during the crisis across the region have reported employment-related problems. A rapid assessment interviewing migrant workers in several countries, including Thailand, during April–May 2020 found that 32% reported work-related problems or abuses such as arbitrary wage-cuts, inability to refuse work during lockdown, being pushed to take unpaid leave, having their personal documents kept by employers, being threatened with the termination of their contacts, or other forms of harassment and violence. Anecdotal reports indicate that most used their meagre savings to survive during the lockdown period. Most of the respondents in this assessment were women (ILO, 2020n; Thubchumpon, 2020).

²² It is estimated that approximately 350,000 Filipino overseas workers had returned home by 1 May 2020.

Even before the pandemic, migrant workers in ASEAN already faced many challenges, including employment-related and human rights abuses. The absence of more effective, timely, and coordinated government responses to contain the pandemic and a lack of measures to protect the rights of migrant workers and their families has obviously exacerbated their precarious condition (Thubchumpon, 2020). According to the Task Force on ASEAN Migrant Workers (TF-AMW), 'The COVID-19 pandemic exposed the existing discrimination, inequality in access to decent work, health care, and better job opportunities. ASEAN's vulnerable people, migrant workers, are at higher risk of contracting COVID-19, as they are on the front lines, workers in low-wage, high-contact, essential jobs in sectors such as health care, retail, and government services. In addition, they may be less likely to have access to medical testing' (2020: para. 3). The ASEAN ministers acknowledged the lack of adequate coordination amongst AMS as a challenge in meetings held in April and June 2020.

In the past, CSOs and trade unions have demanded absolute protection and promotion of the rights and welfare of migrant workers, having outlined more than 140 recommendations (TF-AMW, 2018). Key strategic recommendations include the protection of human and labour rights, a complaint mechanism, pre-departure and postarrival training, the reintegration of returnees, monitoring recruitment agencies, social protections for migrant workers, labour inspection and work safety, and the promotion of decent work for all, including migrant domestic workers (Serrano, Marasigan, and Pupos, 2014).

Persistently depressed economic conditions could lower demand for migrant workers and lessen support for cross-border migrant workers and permissive migration policies. For example, in construction and entertainment (major sectors that absorb migrant labour), lower demand for migrant workers is reportedly emerging in Thailand where no new construction work has taken place, and hotels and entertainment places closed in March 2020 (National Geographic, 2020).²³ Alhough ongoing construction projects have resumed and hotels reopened in June 2020, there is little hope for a 'return to normal', resulting in a lessened demand for labour. Moreover, the Government of Thailand has imposed additional scrutiny measures for the recruitment of migrant workers by contractors, creating further hurdles and possibly incentivising the illegal deployment of migrant workers ('Bangkok Tightens Scrutiny', 2020). Yusof, Muuti, and Ariffin (2020) and other media reports suggest that growing negative social attitudes towards migrants are already evident in some settings.²⁴ There is also widespread concern that prolonged travel restrictions across national borders may lead to labour shortages and speed up the mechanisation of sectors like agriculture, further lowering demand for migrant workers in the future.

²³ According to National Geographic, 'The unemployment rate reached 9.6% in May for the greater Bangkok area and 8.4 million workers across the country are at risk of losing their jobs, according to one government estimate. More people are homeless because they can't afford to pay rent' (2020).

²⁴ See earlier discussions on reported discrimination of migrant workers in Malaysia, Singapore, and Thailand.

Box 3: Association of Southeast Asian Nations Stimulus Packages

To deal with the fallout of the coronavirus disease pandemic, Association of Southeast Asian Nations (ASEAN) governments and central banks have announced fiscal and monetary policy packages on an unprecedented scale. Monetary policies adopted by several ASEAN member countries were meant 'to ensure adequate liquidity and boost confidence in the economy when initial concerns on broken supply chains and travel cancellations emerged' (Zulkhibri and Sinay, 2020: 4). Okamura, Nguyen, and Doi (2020) noted that governments in the region eased interest rates and provided government guarantees on select bank lending activities, market interventions to improve temporary credit lines, purchases of corporate bonds, and even temporary relaxation of regulatory measures for financial institutions.

Some measures to ensure monetary stability were at the regional level, building on financial cooperation initiatives first adopted in 2000 after the Asian financial crisis. Finance ministries and central banks of ASEAN, together with those from China, Japan, and the Republic of Korea (ASEAN+3) established the Chiang Mai Initiative, a network of bilateral swap agreements designed to protect member countries against future crises. In 2010, this was revamped as the Chiang Mai Initiative Multilateralisation (CMIM), and a corpus of \$240 billion was set up to provide dollar liquidity support to reduce vulnerability to short-term external shocks.

In response to the coronavirus disease pandemic, ASEAN+3 finance ministers and central bank governors reached agreements to enhance the CMIM further, making it more effective and operationally ready for countries in need. These included institutionalising the use of local currencies in addition to the dollar for CMIM financing, broadening financing options for members in times of need.

Fiscal policy measures in the region have been mostly aimed at encouraging household consumption and helping companies survive the severe economic downturn. Governments implemented fiscal stimuli to boost the capacity of the health sector and cushion the impact on the broader economy, particularly in hard-hit sectors such as tourism and micro, small, and medium-sized enterprises.

Sources: Okamura, K., N. Thi Hong, and T. Doi (2020), 'Boosting ASEAN+3's Economic Resilience and Recovery during COVID-19', *Nikkei Asia*. <u>https://asia.nikkei.com/Opinion/Boosting-ASEAN-3-s-economic-resilience-and-recovery-during-COVID-19</u> (accessed 29 October 2020); and Zulkhibri, M. and J.B. Sinay (2020), 'Assessing ASEAN Economy Policy Responses in a Pandemic', *ASEAN Policy Brief*, No. 2. 28 May. <u>https://asean.org/storage/2020/06/ASEAN-Policy-Brief-02_31-May-2020.pdf</u> (accessed 28 October 2020).

The sizes of the stimulus packages have ranged from 10% of GDP to 40% of GDP. Collectively, fiscal packages adopted by AMS amount to a \$318.2 billion, equivalent to 10.1% of regional GDP in 2019. Thailand has the largest total stimulus package with three instalments reaching \$88.8 billion, followed by Indonesia with \$81.8 billion.

While the fiscal stimulus packages adopted by AMS are primarily aimed at helping affected industries, a significant part of the measures are for the health sector and subsidies to households, including cash allowances, electricity subsidies, and subsidies for social security contributions and pensions. These subsidies are crucial for the daily needs of workers and their households, especially those in the low-income and vulnerable categories.

In most countries in the region, food packages have also been provided to those in need. For example, the Philippines announced the reallocation of around \$3.9 billion for subsidies in cash and basic needs to 18 million low-income households over 2 months. In Singapore, citizens aged 21 and older will each be given \$\$900 (\$634.90), while selfemployed persons will be paid \$\$3,000 (roughly \$2,116.20) in three tranches to supplement their incomes. Innovative measures such as the 'rice ATMs' in Indonesia and Viet Nam, or pay-it-forward food coupons in Thailand, were also introduced. Beyond support for basic needs, other subsidies, such as for electricity, fuel, pensions, and social security contributions have helped lessen the financial burden on households. However, despite such short-term measures, for most migrant workers, maintaining their livelihoods during the COVID-19 crisis and coping amidst vulnerabilities and marginalisation remains a challenge.

4. Coping Strategies of Migrant Workers After Job Losses: The Case of Myanmar

COVID-19 created a crisis within a crisis, that is, a health crisis led to an economic and survival crisis amongst migrant workers returning from different parts of the ASEAN region. In the case of Myanmar, reports indicate that around 250,000 workers, both domestic and abroad, have lost their jobs since March 2020 following the closure of microenterprises, tourism sites, and restaurants and hotels (Zaw Zaw Htwe, 2020).²⁵ Migrant workers from Myanmar typically work as temporary labourers in Thailand (and therefore lack the flexibility to switch jobs), and job loss often means a loss of work and residence permits, pushing them into irregular status without protection. There are serious obstacles barring them from accessing food supplies and health services in Thailand (for example, a Thai bank account is needed to access social protections). Several banks in Yangon reported an approximately 40% decline in external remittances during April–June 2020.

In interviews, migrant workers returning from Thailand to their homes in Myanmar's Ayeyarwaddy region reported that the most stressful types of situations they had experienced since March 2020 were related to health, jobs, economic conditions (of the family and self), social and emotional issues, and the death of a friend or relative.

²⁵ This subsection is based on interviews and discussions with CSOs in Myanmar held on 21 July 2020. Some of the findings were validated by the ILO report of 2 June 2020 on Myanmar (ILO, 2020j).

Figure 2.3: Impacts of Job Loss on Migrant Workers from Myanmar



NGO = nongovernment agency.

Sources: Zaw Zaw Htwe (2020), 'Quarter of a Million Myanmar Workers Left Jobless Due to COVID-19', *The Irrawaddy*. 26 June. <u>https://www.irrawaddy.com/news/burma/quarter-million-myanmar-workers-left-jobless-due-covid-19.html</u> (accessed 20 July 2020); and authors' personal communication with local civil society organisations on 21 July 2020.

In Myanmar, those migrating from interior rural areas to Thailand, Malaysia, or Singapore experienced more income shocks than those coming from other regions, as agriculture and related activities had stagnated in these areas long before, and could offer the returnees little (ILO, 2020I). For many households, loss of income in their villages pushed members to Thailand for work. Job opportunities for informal workers within Myanmar have also declined. For example, since the onset of the pandemic, several garment factories in the country slowed production and retrenched workers.²⁶ There are reports that major importers like Korea and Europe have cancelled their orders. Myanmar's Garment Manufacturers Association issued a media statement to this effect in early July 2020 (Myo Pa Pa San, 2020). This has left those returning home with fewer options for survival.

²⁶ As of December 2019, garment factories in Myanmar were estimated to employ around 100,000 workers, with a profit of around \$4.5 billion.

Box 4: Civil Society Organisation Response to the Coronavirus Disease Pandemic

Following widespread national lockdowns and curfews across the region, many migrant workers and daily wage earners found themselves stranded without money, food, or advance payment from their employers. They also faced a lack of information on the possibility of travelling and their migration status if they were to return home. Many CSOs directed their efforts towards addressing these immediate concerns, which dominated the discourse in the weeks following the lockdowns and curfews. The civil society response was varied, ranging from the provision of immediate relief to advice on managing the crisis and connecting people to various government schemes and facilities arranged by CSOs.

At the start of the pandemic, maintaining hygiene and safety was promoted as a precautionary measure: people were encouraged to remain safe, maintain physical distancing, and use masks and sanitisers. Families were provided with health kits consisting of masks and sanitisers and/or soap.

This was followed by the dissemination of coronavirus disease information in several different languages adapted to the context in which migrants were living and working, and hotlines were later provided to track gender-based violence and harassment, offer legal services and advice on migration procedures for migrant workers desiring to return home, and monitor human rights (largely related to curfew and movement control orders in Malaysia and Singapore). By June 2020, many CSOs were offering mental health support, training, advocacy, and campaign support. They have created solidarity networks and provide support to migrants, including food, water, essential medicines, shelter, personal protective equipment, and economic assistance. They have also established relief funds for farm workers, domestic workers, and others who have lost their livelihoods as a result of the pandemic. Worker and employer organisations are promoting equal treatment, decent work, and respect for fundamental principles and rights at work through social dialogue and in coordination with local authorities. CSOs and other stakeholders are also working to include migrants in the planning of policy responses to the pandemic. The concerted action of governments and stakeholders in developing coronavirus disease policy responses is key to ensure that migrants' rights and contributions are addressed and fostered.

Source: United Nations Network on Migration (2020), *Standing in Solidarity with Migrants*. 8 July. Vienna: United Nations Office on Drugs and Crime.

https://www.unodc.org/documents/press/releases/2020/July/NW_Statement_for_CSOs_final_version.pdf (accessed 25 August 2020).

Given the extent and scale of the disruptions that have occurred since March 2020, the impact of COVID-19 is likely to be much worse than that of the 1997 Asian financial crisis or the global financial crisis of 2008–2009. While these previous economic crises led to job losses and considerable distress amongst migrant workers, this proved to be temporary, as remittances and influxes of migrants have subsequently risen steadily (Abubakar, 2002; Abella and Ducanes, 2009). The long-term economic impact of the COVID-19 pandemic is not yet clear. While there are some grounds for optimism that ASEAN economies could recover quickly once the pandemic has ended, there is much uncertainty about how long the crisis itself is likely to last. Meanwhile, it is critical that governments focus on alleviating the economic and other suffering of their own citizens as well as that of migrant workers, who are amongst the most vulnerable sections of the region's population.

Chapter 3

Impact of the Coronavirus Disease Pandemic on the Sustainable Development Goals

In 2015, all ASEAN governments adopted the Agenda 2030 for Sustainable Development, including 17 goals and 169 targets, which were designed to shape action over the next 15 years to realise the socioeconomic and cultural rights of all and to balance economic, social, and environmental development (UN DESA, 2016b). This agenda marks a milestone by mainstreaming migration, migrant workers, and the issues of inequality and decent work as integral components of development policy into SDGs 8 and 10 (Foresti and Hagen-Zanker, 2018; Long et al., 2017). In paragraph 29 of the declaration accompanying the adoption of the SDGs, ASEAN governments also committed to recognise the positive contribution of migrants to 'inclusive growth and sustainable development'.

The ASEAN governments also committed to 'cooperate internationally to ensure safe, orderly and regular migration involving full respect for human rights and the humane treatment of migrants regardless of migration status' (UN General Assembly, 2015). Given the wide range of factors driving the phenomenon of migration, from poverty to climate change, it is seen by international agencies (IOM, 2018) and by ASEAN as a crosscutting issue relevant to all 17 of the SDGs and most of the 169 targets in the agenda. The paragraphs below review the impact of the COVID-19 crisis on select SDGs and their targets with respect to migrant workers, based on currently available information.

1. Baseline and Progress

A snapshot analysis of where the ASEAN region stood in 2015 (the baseline year for measuring progress towards the SDGs to be achieved by 2030) revealed that the region as a whole made significant progress in four goal areas from 2000 to 2015: poverty eradication (SDG 1), quality education (SDG 4), affordable and clean energy (SDG 7), and life below water (SDG 14) (UNESCAP, 2017b). For example, extreme income poverty (earning less than \$1.90 a day in 2011 purchasing power parity) more than halved from 26% in 2000 to 9% in 2015.

However, progress was slow or stagnant in six goal areas: good health and well-being (SDG 3); gender equality (SDG 5); clean water and sanitation (SDG 6); industry, innovation, and infrastructure (SDG 9); responsible consumption and production (SDG 12); and peace, justice, and strong institutions (SDG 16).

Figure 3.1: Links between the Sustainable Development Goals and Migration

SDGs related to **MIGRANT WORKERS** POVERTY. Migration can help families to improve their well-being through increased income, consumption, and resilience. HEALTH. Migrants can be especially vulnerable to implementation. health risks. Migrants having reached their CLIMATE ACTION. Migrants are often among the destination may not have access to healthcare because of their migration or contract status. EDUCATION. While migration helps improve both account. education access and quality for their families in origin countries, migrant workers can improve their skills and will have better opportunities to attain decent work and higher income GENDER EQUALITY. Migrant women often lack regularized status or access to social and welfare systems through their employer or host governments. PARTNERSHIPS. Having specific and current data They are especially vulnerable to human trafficking pertaining to migrant workers specific needs, both in and gender-based violence. Supporting women's sending and host countries, helps to increase the participation and leadership reduces the risk of capacity to deliver migrant services. gender-based violence and ineaualities. DECENT WORK. The importance of migrants in the global economy continues to grow. Ethical recruitment, humane working conditions and safe migration helps enhance the imoact of migrants on development.

Source: International Organization for Migration (2018), Migration and the 2030 Agenda: A Guide for Practitioners. Geneva: International Organization for Migration.

SOURCE

Since 2015, ASEAN has been making progress in several target areas under SDG 3 (good health and well-being); SDG 5 (gender equality); SDG 7 (affordable and clean energy); SDG 8 (decent work and economic growth); and SDG 9 (industry, innovation, and infrastructure). The UNESCAP analysis (2017a) cautioned that, to achieve the SDGs by 2030, ASEAN as a whole must to maintain its current rate of progress in reducing under-5 and maternal mortality, ensuring equal opportunities for leadership for women, increasing access to renewable energy, enhancing employment, increasing access to mobile networks, and reducing carbon dioxide emissions.

REDUCE INEQUALITIES. Awareness brings changes. Every migrant worker is a migrant hero.

SUSTAINABLE CITIES. Migrants and migrant workers help cities to thrive and become more vibrant, succesful centers of economy and life. Local governemnts are encouraged to include migration and migrants in the urban development planning and

most harshly affected by climate-related disasters. Financial planning need to take these patterns into

PEACEFUL SOCIETY. Preventing trafficking in personas and offering survivors of trafficking safe and sustainable reintegration support efforts to eliminate all forms of violence against women and girls.



Figure 3.2: The Association of Southeast Asian Nations' Performance on the Sustainable Development Goals, 2015

Source: United Nations Economic and Social Commission for Asia and the Pacific (2017), *Southeast Asia Subregion Challenges and Priorities for SDG Implementation*. 17 May. Bangkok: United Nations Economic and Social Commission for Asia and the Pacific. <u>https://www.unescap.org/resources/southeast-asia-subregion-challenges-and-priorities-sdg-implementation</u> (accessed 12 February 2021).

ASEAN would also have to reverse the trend in several other target areas, including increasing investment in agriculture, reducing overweight and wasting amongst children under 5, reducing adolescent fertility, increasing the supply of qualified primary school teachers, promoting inclusive and sustainable industrialisation, achieving the sustainable management and efficient use of natural resources, conserving natural forests, and reducing all forms of violence and related death rates.

It has been predicted that the COVID-19 pandemic will have both short- and long-term impacts on many of the SDGs (Evetts, n.d.). In the ASEAN region the COVID-19 pandemic is expected to significantly impact the ability of least developed countries like Cambodia, the Lao PDR, and Myanmar in the Lower Mekong Region (currently the main labour-

supplying nations to the ASEAN region) to meet the SDGs by 2030, as well as reversing gains made in recent years (UN, 2020a).

The section below presents an overview of the possible channels of impact of COVID-19 on different SDGs and specific targets related to migration and the welfare of migrant workers. Although the paucity of reliable and up-to-date region-wide data limits this analysis, the discussion below indicates the overall trends and provides useful insights on appropriate mid-course corrective action for governments and policy makers, and future research.

2. Impact of the Coronavirus Disease Pandemic on Migration-Related Sustainable Development Goals

2.1. Sustainable Development Goal 1: End Poverty in All its Forms Everywhere

Historically, an important driver of migration has been the quest of people to improve their lives and those of their families materially (Murrugarra, Larrison, and Sasin, n.d.). Large differences in income within and between countries motivate individuals to escape poverty through migration. Migrants typically not only improve the economic status of their own families, but through remittances can make an impact on sending countries at the national level. They also acquire new skills and education that can make a lasting impact on poverty alleviation.

In implementing the SDG targets related to poverty, governments can help integrate the rights and interests of migrant groups, including asylum seekers, refugees, and internally displaced persons, in local and national poverty reduction policies and programming.

Migration relevant targets to be achieved by 2030 under SDG 1 include (i) eradicating extreme poverty (currently measured as earning less than \$1.25 a day) for all people everywhere (Target 1.1); and (ii) implementing nationally appropriate social protection systems and measures for all, including floors, and achieve substantial coverage of the poor and the vulnerable (Target 1.3).

Impact of the Coronavirus Disease Pandemic

According to a UN assessment of the global impact of COVID-19, the crisis risks 'reversing decades of progress in the fight against poverty and exacerbating already high levels of inequality within and between countries' (UN, 2020a). The UN report warned that, unless adequate measures are promptly put in place, the disruptions imposed by the pandemic and the measures adopted to suppress the virus will dramatically worsen the situation. Another analysis asserted that 'the socio-economic impacts of the pandemic are predicted to hit vulnerable populations in the region – which include urban and rural poor, migrants, informal workers, refugees, indigenous and ethnic groups, religious minorities, and women – harder. This is due to a complex intersection of a lack of social protections, limited infrastructure for hygiene and sanitation, difficulties in implementing social distancing, and overall low resilience to shocks like these due to underinvestment in the necessary supports' (Open Development Mekong, 2020).

Based on a survey, the University of Chicago has estimated that globally two out of every five jobs lost during the pandemic may not come back, and that close to 40% of people are likely to face pandemic-induced layoffs from work (Barrero, Bloom, and Davis, 2020).

What is said about the global trend is equally relevant for ASEAN, and probably more relevant because a higher proportion of the population is dependent on the informal sector (Nortajuddin, 2020b).

A major negative impact of the COVID-19 pandemic in the ASEAN region will likely be reduced employment, in both the formal and informal sector, which in turn will considerably drive up the number of people living in poverty in the region ('ADB Says Cambodia to Lose', 2020; ILO, 2020b; 2020c). While official figures are not yet available, ADB's Asian Development Outlook 2020 concluded that, overall, ASEAN countries stand to lose around 16 million jobs in 2020, and will take 2–3 years to recover.

In the context of Southeast Asia, according to a UN policy brief, the COVID-19 crisis also threatens to destroy the livelihoods of the region's 218 million informal workers, who represent 51–90% of the national non-agricultural workforce in the subregion. Without alternative incomes, formal social protection systems, or savings to buffer these shocks, workers and their families will be pushed into poverty, reversing decades of poverty reduction (UNESCAP, 2020c).

In the ASEAN region, economic shocks wrought by COVID-19 are already impacting poverty and welfare indicators, especially amongst the vulnerable and workers in the informal economy. Virgil and Lie argue that 'COVID-19 impacted all sectors in ASEAN, from health to the economy. It is predicted that 60 million people in East Asia and the Pacific may be pushed into poverty as a result of a 20% loss in income' (2020). While the region has accomplished considerable progress in poverty alleviation in the last few decades, approximately one in seven residents across the region were still living on less than \$2 per day as of 2015 ('The Fourth Industrial Revolution', 2020).



Figure 3.3: Post-Coronavirus Disease Pandemic Job Loss amongst Intra-Association of Southeast Asian Nations Migrant Workers

Lao PDR = Lao People's Democratic Republic.

Source: Myat Thura (2020), *Myanmar: Over 60,000 Workers Lost Jobs Due to Factory Shutdowns. Caused by COVID-19.* 28 April. London: Business and Human Rights Resource Centre (republished from Myanmar Times). <u>https://www.business-humanrights.org/en/myanmar-over-60000-workers-lost-jobs-due-to-factory-shutdowns-caused-by-covid-19</u> (accessed 25 September 2020).

At the country level, between 2015 and 2019 the Philippines made progress in combating poverty; the poverty rate decreased from 21.6% in 2015 to 16.6% in 2018, and was projected to decline further during 2020–2025. A report in 2019 showed that about one in five of the country's 106 million people lived in extreme poverty and, of these, around 50% were reportedly undernourished. The ongoing public health crisis is expected to come as an additional burden for impoverished families and those living in urban slums, many of whom are informal workers. Similarly, Nortajuddin observes that things have never looked bleaker in Indonesia, where 24.79 million people are considered impoverished (2020b).

Figure 3.4: Employment in the Tourism Sector as a Share of Total Employment and Share of Informality in Tourism Employment, Latest Available Year



Note: The informal employment share is not available for Fiji, the Philippines, or Thailand. Source: Calculations based on national labour force surveys in the International Labour Organization (ILO) repository of micro datasets (<u>www.ilo.org/ilostat</u>). ILO (2020e), *Thematic Brief: COVID-19 and Employment in the Tourism Sector: Impact and Response in Asia and the Pacific.* 24 April. Bangkok: ILO. https://www.ilo.org/wcmsp5/groups/public/---asia/---ro-

bangkok/documents/briefingnote/wcms_742664.pdf (accessed 20 July 2020).

Box 5: Collapse of the Travel and Tourism Industry

The travel and tourism sector is fundamental to the economic health of several Association of Southeast Asian Nations (ASEAN) member states, accounting for about 12.0% of the region's gross domestic product in 2019 and 13.3% of employment (World Travel and Tourism Council, 2020). Online travel booking services, the bedrock of thousands of startups in the region, are now a \$34 billion industry and saw annual growth of 15% in the region from 2015 to 2019.

The collapse of the tourism sector (a major source of revenue and jobs in the ASEAN region) induced by the coronavirus disease (COVID-19) pandemic is an important factor behind the International Monetary Fund projections of stalled economic growth for 2020 (International

Monetary Fund, 2020). The impact on the region's tourism sector was felt at an early stage when the number of tourists from China declined abruptly. Now, with the virus spread across the globe, as many as 96% of the world's destinations are imposing some form of travel restrictions according to the United Nations World Tourism Organization.

According to an International Labour Organization report in April 2020, Asia and the Pacific stands to lose approximately 63 million jobs and \$1 trillion in gross domestic product, as international tourism plunged by about 80% in 2020 compared to 2019. This is by far the worst result for tourism since 1950 and puts an abrupt end to a 10-year period of sustained growth since the 2009 financial crisis (International Labour Organization, 2020e). Amongst countries with available data, those with the highest share of employment in tourism are Cambodia (with 6.7%), Thailand (with 9.0%), and Viet Nam (with 6.9%). During peak travel months, the percentage of workers employed in tourism can reach as high as 12%–15% in some countries. Tourism-dependent economies like Thailand – where nearly one in four jobs are in the hotel and restaurant sector, and migrant workers (both domestic and regional) account for two-thirds of those engaged in tourism sector – stand to lose 1.8 million jobs (ADB, 2020a).

In many ASEAN countries, more than three in four workers in the tourism sector are in informal jobs, leaving them especially vulnerable to the negative impacts of the COVID-19 crisis.²⁷ Workers who continue to work for hotels, airlines, or other hospitality industries typically do not have the option to work remotely, and therefore have a heightened risk of COVID-19 infection. Those who do fall ill may be disadvantaged in accessing healthcare services as informal workers and have no way of replacing their incomes if they stop working because of sickness or lockdowns.

Source: Authors.



Figure 3.5: Employment in the Tourism Sector as a Share of Total Employment by Sex, Latest Available Year

Source: Calculations based on national labour force surveys in the International Labour Organization (ILO) repository of micro datasets (www.ilo.org/ilostat). ILO (2020e), *Thematic Brief: COVID-19 and Employment in the Tourism Sector: Impact and Response in Asia and the Pacific.* 24 April. Bangkok: ILO. <u>https://www.ilo.org/wcmsp5/groups/public/---asia/---ro-</u> <u>bangkok/documents/briefingnote/wcms_742664.pdf</u> (accessed 20 July 2020).

²⁷ Informal sector jobs are characterised by a lack of basic protections, including social protection coverage.

Beginning in April 2020, Cambodia, Indonesia, the Lao PDR, the Philippines, Thailand, and Viet Nam reported a spike in unemployment rates that continued into June 2020 (Hutt, 2020). For the last decade, Thailand's jobless rate has hovered around 0.6%, while Cambodia's unemployment rates before the COVID-19 crisis had barely risen above 2% since the early 1990s. Viet Nam's unemployment rate had also been consistently at less than 2%, while the formal unemployment rate in the Lao PDR was less than 1% through the end of 2019. Unemployment rates have risen to a 10-year high in Viet Nam, where the pandemic has cost nearly five million Vietnamese workers their jobs in just the first quarter (Q1) of 2020, according to the country's General Statistics Office. Experts predict that this rate will increase when figures for the more economically debilitating second quarter results are released (OECD, 2020b; 'ASEAN's Virus Dilemma', 2020; Hutt, 2020).

In the case of the Lao PDR, the World Bank observed that a sharp drop in the performance of the travel, tourism, and hospitality sectors – which account for 11% of total employment and 22% of employment in urban areas – has caused widespread job losses (2020c). Sengpaseuth (2020) reported that joblessness in the Lao PDR spiked to around 25% in May alone due to the pandemic. Between 96,000 and 214,000 additional people are projected to fall into working poverty as a result of the pandemic. In all cases, job losses in Thailand are projected to occur in construction, services (hotels and tourism in particular), and seasonal agriculture amongst both domestic workers and migrants.

By June 2020, the situation in Indonesia was equally serious, with around 6 million reported job losses in the construction, textile and garment, and service sectors (Jefriando and Suroyo, 2020). This in turn is expected to impact jobs for returning migrant workers and/or cause competition for jobs, resulting in wage cuts (Soeriaatmadja, 2020). Indonesia's National Planning Board (Bappenas) reasoned that the pandemic will worsen the labour market for Indonesia's young candidates as a result of higher barriers of entry into the job market, long-lasting lower income levels, and worsening labour conditions. These poor employment conditions might, amongst other factors, force desperate young graduates to accept jobs for which they are overqualified with low pay and minimum growth opportunities. Others might even be forced to accept informal jobs, typically characterised by vulnerable contract terms and substandard working conditions (Pradesha et al., 2020).

The situation is no different in Cambodia where layoffs were reported amongst garment workers, while at the same time migrants returned from Thailand in April–May 2020 ('ADB Says Cambodia to Lose', 2020). Given the economic decline, companies have been unable to finance expenses, including their employees' wages; and by the end of June 2020 there were reports of employers seeking 'deferred payment of wages', particularly in the travel and hospitality sector (Bagus Enrico and Partners, 2020). The return of migrant workers and increased competition for local jobs were expected to drive significant wage cuts in the garment sector, which was already bracing itself for the cancellation of export orders and lack of new business proposals. Given this situation, many in Cambodia and Myanmar wanted to return to Thailand (despite travel restrictions and a possible lack of jobs in Thailand too), hoping for a turnaround ('Migrant Workers Poised', 2020).

The COVID-19 shock has been most devastating for discretionary services, including restaurants, hotels, travel, and entertainment, which face a slow recovery but account for a high proportion of employment. One assessment predicted that many of these service businesses would soon have to choose between cutting workers, closing down, or facing bankruptcy (Subbaraman and Varma, 2020). The costs of unemployment for individual migrant workers and their households are not hard to imagine. When a person loses their job, there is often an immediate impact on that person's standard of living, particularly amongst informal sector workers whose savings tend to drift down to zero when a crisis occurs at home. Job losses are also likely to trigger social strife and increased household violence (ILO, 2020i; 2020k).

Despite playing an important role in national economies, providing a link with global structures of agricultural production and trade, and feeding the world, many agricultural workers and their families have long suffered from poverty and periodic and seasonal food insecurity (ILO, 2020c). Post-COVID 19, experts have also predicted that working poverty rates around the world and in ASEAN are expected to increase significantly. A report published by the International Food Policy Research Institute (IFPRI) for Indonesia and Myanmar observed that agricultural workers experienced the highest incidence of working poverty during the early days of COVID-19 as there was prolonged non-activity followed by stiff competition for local jobs that led to arbitrary wage cuts. By June 2020, one-quarter of workers engaged in the sector were estimated to be in extreme poverty (Pradesha et al., 2020; ILO, 2020a). In addition to this, as governments order nonessential business to close, millions of people employed in the gig economy – who are mostly on precarious contracts, with little or limited access to health facilities or health insurance – are put in vulnerable situations and may turn to risky or exploitative employment.

Based on multiple field reports, the ILO expressed concern that casual and migrant wage workers (particularly in agriculture), landless farmers, small-scale traders, and commodity producers whose ability to purchase food grains were most affected. The ILO estimates that, to maintain their standard of living and not fall into poverty, low-paid rural migrant workers in the worst affected countries (like Myanmar, Indonesia, and the Philippines) would have had to find an additional week of employment every month (ILO, 2020c). Given widespread curfew and the closure of economic activities, replacement work has not been possible for many who have lost their jobs.

The social costs of job losses are difficult to calculate but are more real. Past crises have shown that increased unemployment (both domestic and migration-related) leads to domestic violence; social strife within communities; and, politically, nationalistic rhetoric, protectionism, and severe restrictions on migration of poorer groups. Moreover, prolonged periods of unemployment amongst migrant workers without a replacement job at home could lead to resentment within the local community, and/or drive unemployed people to crime to meet their immediate economic needs (ILO, 2020i; UNESCAP, 2020c).

Social Protections

SDG 1.3 calls for the implementation of appropriate social protection systems nationally, including support for different floors,²⁸ and achieving substantial coverage of the poor and vulnerable by 2030. A UN report observed that social protection or assistance has a positive impact on migrant workers and should be systematically enforced to secure SDG 1.3 (2018). The report adds that 'social protection can cushion the adverse social effects of rapid structural change, including those associated with migration, unemployment, rising inequalities and pandemics'. It asserted that better access to social protections and services have been proven to 'enable families to care for and sustain their members and reduce both the costs and time involved in work and other daily activities. They increase the chances that individuals and their families can lift themselves out of poverty and live dignified and productive lives'. This underlines the importance of taking marginalised and vulnerable communities into account in developing the pillars of ASEAN's economy in all circumstances (ATUC, 2016b). Although migrant workers play a key role in ASEAN's economic development, a large proportion of them lack adequate social protection coverage in both their home and host countries, such as proper contracts, unemployment support, retirement funds, accident coverage, sufficient paid leave, and family care. Many labour-sending nations also lack adequate resources for such programmes (Olivier, 2018; ASEAN, 2016a).

The ASEAN community has yet to achieve fair treatment of migrant workers and effective protection for them from abuse, exploitation, and violence (ATUC, 2020). AMS have been reluctant to extend the coverage of the Employment Act and Workmen's Compensation Act to foreign migrant workers, leaving the parties to negotiate benefits individually (ATUC, 2020). Instead, many receiving countries in the region treat the migrant labour market arrangements (including wages and benefits) between employees and employers as a private matter, and refuse to intervene on matters of working hours and rest days for live-in domestic workers.²⁹

Even prior to the pandemic, AMS acknowledged that the absence of coordinated social protection programmes rendered workers vulnerable to discrimination in the laws and practices of both origin and destination countries. They also recognised that undocumented and women migrant workers in particular are left with little or no protection from very low wages and poor working conditions (Orbeta et al., 2013).

In addition, existing social protection programmes have always lacked a comprehensive list of migrant workers employed at any given time, thus weakening the targeting of assistance delivery. The targeting effectiveness of these programmes and registries has also been poor, often excluding over half of the poorest migrants such as undocumented workers or women (who often serve as domestic workers or family care givers).

²⁸ According to the ILO, national social protection floors should comprise at least the following four social security guarantees: (i) access to essential health care, including maternity care; (ii) basic income security for children, providing access to nutrition, education, care, and any other necessary goods and services; (iii) basic income security for persons in active age who are unable to earn sufficient income, in particular in cases of sickness, unemployment, maternity, and disability; and (iv) basic income security for older persons.

²⁹ While Singapore has put in place strict oversight arrangements on the provision of and access to welfare systems for migrant workers, such oversight arrangements are treated flexibly in Malaysia and Thailand.

Some countries in the region have responded to the crisis by increasing social welfare measures for vulnerable populations. For example, the Philippines is currently providing a cash transfer to 75% of the poorest households across the country using its social registry to identify beneficiary households (Rutkowski, 2020). However, this approach likely excludes a considerable portion of intended recipients, particularly migrant workers who may not have registered or renewed their registration. Although Thailand's response was seen as efficient, reaching almost two-thirds of households, few schemes targeted the social protection of documented regional migrant workers, and a large number of undocumented, irregular workers were excluded (ILO, 2020i; 2020o).

The COVID-19 crisis is also gradually reducing access to safe and reliable employment opportunities for migrant workers and informal sector in general.³⁰ Informal sector workers carry obvious risks, like substandard safety provisions and working with dangerous goods and machinery, on construction sites for example. During the COVID-19 crisis, migrant workers have reportedly been prepared to engage in unsafe tasks to avoid job losses (Satursayang, 2020). For domestic workers and those engaged in service sector, there is no process in place for resolving workplace-related safety issues, and helplines provided by civil society organisations only extend support at times. Moreover, women (particularly those in domestic work and elderly care) are sometimes unable to access these services because of their work location and conditions imposed by their employers.

Box 6: Importance of Equal Treatment

Many migrant workers in the Association of Southeast Asian Nations (ASEAN) region are vulnerable to discrimination and exclusion in their destination countries (Olivier, 2018; ATUC, 2019b). Two overlapping reasons for this are their migration status (irregular or undocumented) and the nature of their employment (e.g. informal work or the informal economy). Even regularised migrant workers in the formal economy are not subject to equal treatment across all four social protection guarantees. Since early 2000s, the main receiving ASEAN countries – Brunei Darussalam, Malaysia, Singapore, and Thailand – have also reduced hospital subsidies for non-nationals or obliged them to take up private insurance.

Six of the 10 ASEAN Member States – Indonesia, Malaysia, Myanmar, the Philippines, Thailand, and Singapore – have ratified the Equality of Treatment (Accident Compensation) Convention, 1925 (No. 19) to ensure some occupational injury protection for non-national workers. Under the Work Injury Compensation Act 2009, private insurance for work injury and illnesses is compulsory in Singapore for both migrant and non-migrant workers engaging in manual work or earning less than \$\$1,600 per month for non-manual work. In contrast, although Thailand recognises equality of treatment for accident compensation under its 1994 Workmen's Compensation Act B.E. 2537, in reality, most migrant workers are uninsured for occupational injury and diseases. This is because of their undocumented status, non-compliance of employers, migrants' lack of awareness of their rights, language barriers, onerous administrative procedures, and other factors. In Malaysia, non-permanent resident migrant workers do not qualify for work injury and invalidity protection under its Social Security Organization, and are instead obliged to be insured under the lesser Foreign Workers Compensation Scheme. Alhough not a signatory of Convention No. 19, the Government of Viet Nam recently amended its Law on Social Insurance to permit the coverage of migrant workers with valid work permits as of January 2018.

Source: Authors.

³⁰ Workplace shelter-related issues are discussed under different SDGs.

2.2. Sustainable Development Goal 2: End Hunger, Achieve Food Security and Improved Nutrition, and Promote Sustainable Agriculture

Low-paying employment in rural areas and low productivity of small-scale food producers drive migration to urban centres and overseas, as workers search for better income opportunities. At the same time, agricultural communities are often affected by climate change, extreme weather, drought, flooding, and other disasters, necessitating adaptation strategies to boost livelihoods and help prevent forced environmental migration.

Impact of the Coronavirus Disease Pandemic

A large migrant labour force that depends on the food supply chain sector for their livelihoods remains significantly affected by the COVID-19 crisis. ASEAN's food and beverage sector not only accounts for 116 million jobs (about 35% of the labour force) but also ensures a continued food supply. By the end of July 2020 this labour force was confronting numerous challenges (Chan, 2020). Apart from the threat to lives and livelihoods of millions of those employed in the sector, the pandemic initially raised additional concerns about the durability of the subregion's food supply systems, a complex matrix involving farmers, fishers, labourers, drivers, cold storage, food processors, retailers, and consumers (UN Food and Agriculture Organization [FAO], 2020a). In a briefing note on ASEAN dated 2 June 2020, the OECD observed that '...the COVID-19 pandemic has placed unprecedented stresses on food supply chains, with bottlenecks in farm labour, processing, transport and logistics, as well as momentous shifts in demand. Most of these disruptions are a result of policies adopted to contain the spread of the virus'.

At the onset of COVID-19, initial lockdown measures resulted in panic buying that left many migrant workers and their households in distress as they either could not access supplies or lacked cash on hand to purchase them. Safety nets are essential to avoid hunger and food insecurity for migrant workers and help them overcome temporary shocks. Border closures and export restrictions have limited the availability and affordability of certain food items for countries that rely on imports. Domestically, disruptions in upstream food supply chains have arisen from both mobility restrictions and worker illness during planting and harvesting, in addition to hindered operations in processing, trucking, logistics, and trading ('ADB Says Cambodia to Lose', 2020). The sudden closure of economic activities and international borders quickly halted the transport of essential commodities for some time, leading to public anxiety. Disruptions to domestic and regional food supply chains (including the delayed start of the farming season in Cambodia, the Lao PDR, Myanmar, Thailand, and Viet Nam) caused by major transport and travel restrictions undermined food availability and accessibility (Ellis, 2020; Food Industry Asia, 2020).³¹ At the same time, a host of new problems brought on by the pandemic has stunted both in-country and cross-border supply chains, leaving essentials

³¹ Southeast Asia's food supply chain was already at risk of serious disruption before the COVID-19 pandemic, which has underscored existing frailties and brought them to the fore.

accessible only to a few (Ellis, 2020).

Rapid regional surveys and anecdotal reports indicate that sudden job and income losses are triggering reduced food consumption amongst migrant workers as they try to save money and survive during the lockdown period, leaving them at risk of hunger and malnutrition (ADB, 2020b). Given the lack of income and concerns about the impossibility of returning to work in the near future, households have started 'saving', impacting food consumption and nutrition amongst informal sector workers. Basic food handouts provided by state agencies or charities to compensate for the losses incurred are often limited, and may not meet the nutritional needs of children and pregnant women (Tantau, 2020).

An ADB briefing note found that household food consumption and nutrition have been significantly affected by the loss of jobs and income and by limited access to food. Informal sector workers in particular, most of whom are migrants (either domestic or regional), have been at a higher risk of food insecurity since April 2020. Shortages of labour and input supplies resulting from prolonged lockdowns can reduce the scale of crop production while disrupted logistics limit the options of smallholder farmers to access better priced markets. A parallel report noted distress sales of vegetables during May–June in several parts of ASEAN (ADB, 2020b).

Potential Long-Term Impact

Although food supplies currently appear plentiful and the initial panic buying has subsided, worries persist about the availability of sufficient food at affordable prices in the future, particularly in the context of job and income losses amongst migrant workers. Approximately 61 million people in Southeast Asia are currently undernourished; this number may increase following the pandemic and is likely to moderate only when the economic situation stabilises (FAO, 2019).

Rice is the staple food of Southeast Asia, which produces, trades, and consumes a large portion of the global rice supply. In 2018, Southeast Asia produced more than 220 million tonnes of rice. While both FAO and the IFPRI have projected sufficient rice stocks for the rest of 2020 in ASEAN and Asia and the Pacific more widely, the measures needed to curb further COVID-19 outbreaks could disrupt supply chains critical for food security (ADB, 2020b; Diao and Wang, 2020; Pradesha et al., 2020).

A participant in a webinar on the future of food systems in Southeast Asia post COVID-19 convened by the International Rice Research Institute along with other development institutions asserted that, 'the impacts of the pandemic pose immense threats to the health of communities already struggling with hunger and on the livelihood of vulnerable groups including farmers'. The webinar participants considered various ways that AMS could help smallholders, landless people, and other poor groups who migrate for work across borders and have lost their jobs since March 2020. In the short term, the continued lack of economic access to food is a major challenge that must be dealt with. This could mean cash disbursements in some cases and free distribution of food for a period of time in others.

Studies and anecdotal reports have identified multiple impacts of the pandemic felt by different groups in society, including primary food producers such as smallholder farmers, landless people, and those who have now been without an income for a significant period of time (ADB, 2020b). Field studies and reports from Indonesia, Myanmar, the Philippines, and Thailand observe that curfew forced the closure of operations, removed travel and transport facilities, and prevented workers (both locals and migrants) from working in farms and processing and packaging facilities. In many rural areas, access to farm inputs such as seeds, fertilisers, and crop protection products became challenging. Farming activities virtually came to a halt during April–May and gradually opened only in late June (Pradesha et al., 2020; UNESCAP, 2020b; FAO, 2020b). In the medium term (through early 2021), this non-activity is likely to have significant impacts on labour-intensive food crops, including fruit, vegetables, dairy products, and meat processing (Dzulfikar, 2020).

Acknowledging this situation, the June 2020 ASEAN Summit stated, 'We note that the COVID-19 outbreak has drawn our attention on the immediate danger of food shortage and its adverse effect on nutrition, given a sudden spike in demand and disruption in supply chains, ASEAN needs to also continue its efforts to ensure stable and sustainable food sources' (ASEAN, 2020e). In parallel, the ASEAN Ministers' Summit on Agriculture and Forestry recognised the threat to food security, food safety, and nutrition; and called for pertinent actions (ASEAN, 2020e).

In response, in addition to credit support for the agriculture sector, some ASEAN governments have implemented measures to ensure that supply chains from farms to markets remain open. For example, Malaysia allotted RM1 billion (\$231.6 million) to its Food Security Fund, and provided additional support to farmers and fishers for agricultural production (Prime Minister's Office of Malaysia, 2020). A budget was allocated for food storage facilities and distribution to safeguard supply, and for agro-food projects. Myanmar also provided substantial support for its agriculture sector in the form of cash or loans to smallholder farmers to support production (French-Myanmar Chamber of Commerce and Industry, 2020).

Several AMS have complemented these measures by supporting access to markets and productivity enhancement, and facilitating export processes including for rice. Rural cash-for-work programmes for the economic recovery period were also included in overall measures to ensure food supply chain connectivity during the pandemic (ASEAN, 2020c).

2.3. Sustainable Development Goal 3: Ensure Healthy Lives and Promote Well-Being for All Ages

It is well recognised that to achieve the vision of the SDGs – to leave no one behind –the health needs of refugees and migrants must be adequately addressed (World Health Organization [WHO], 2018). Unfortunately, productive migrant populations have been perceived as a burden on countries' health services rather than an asset. Achieving the health-related targets of SDG 3 in the ASEAN region will require a special focus on migrants and refugees, to help both the cause of these vulnerable groups as well as larger national health goals.

Migrants face a variety of obstacles in accessing quality healthcare, which is often denied to them due to their legal and administrative status. There is a lack of comprehensive national health policies and strategies for migrants in many Southeast Asian countries. With the exception of a few countries in ASEAN, undocumented migrants have been prevented from accessing local health services, resulting in great hardship for them and also contributing to a failure to eliminate vaccine-preventable and other communicable diseases.

Migration relevant targets to be achieved by 2030 under SDG 3 include the following:

- Reduce the global maternal mortality ratio to less than 70 per 100,000 live births (Target 3.1).
- (ii) End preventable deaths of neonates and children under 5 years of age (Target 3.2).
- (iii) End the epidemics of AIDS, tuberculosis, malaria, and neglected tropical diseases; and combat hepatitis, waterborne diseases, and other communicable diseases (Target 3.3).
- (iv) Reduce by one-third premature mortality from non-communicable diseases through prevention and treatment, and promote mental health and well-being (Target 3.4).
- (v) Substantially increase health financing and the recruitment, development, training, and retention of the health workforce in developing countries, especially in least developed countries and small island developing states (Target 3.8).

Impact of the Coronavirus Disease Pandemic

People in vulnerable situations are particularly at risk from the COVID-19 outbreak because of their health and overall social and economic circumstances. Intra-ASEAN migrant workers face greater risks because of their often solitary status and living conditions in workplace shelters or dormitories, which are ideal settings for rapid transmission. They may also find it difficult to social distance. In the case of female domestic workers or caregivers, there is a possibility of further risk from their work, in addition to sociocultural and financial barriers. Working mothers have little recourse to social protection when public health measures such as school closures are imposed (UN Women, 2020b; ILO, 2020n).

Countries such as Cambodia, Indonesia, the Lao PDR, and Myanmar where COVID-19 is likely to exacerbate multiple existing vulnerabilities are of particular concern. Myanmar, for example, has relatively poor public health infrastructure and domestic opportunities for livelihoods. Of the four million Myanmar citizens currently working abroad, it is estimated that 700,000–900,000 may have returned home after losing their jobs, and another 500,000 may have lost their jobs but have yet to return home (Diao and Michael Wang, 2020).³² Almost all have likely lost their incomes and are unable to remit money home, leaving their families to cope with complex livelihood issues, especially health concerns, during a pandemic with less cash on hand. Even before the pandemic, many households in southeast Myanmar were highly vulnerable as a result of the effects of protracted conflict, a lack of sufficient investments in local development, and stagnating

³² This report estimated that remittances declined by 14% during the first 2 weeks of May 2020.

employment opportunities. The COVID-19 crisis has further exacerbated the vulnerability of these households (United Nations Capital Development Fund, 2020). As migrants lose income and remittances correspondingly decrease, families back home dealing with their own pandemic-related challenges will suffer. Over the years, various studies have shown that health care crises can drive people to make risky labour market decisions that can make them more vulnerable, either because they cannot afford to pay for medical care or because they have lost their job as a result of the crisis. This can heighten the risk of forced labour or enslavement, further impacting their health (Human Rights Watch, 2018).³³ Moreover, large numbers of migrant workers fall into employment 'grey areas' such as the entertainment and related service areas, which are associated with higher levels of poverty (Buller et al., 2020).

In Malaysia, Thailand, and Singapore – despite statements to the contrary – there is also serious concern about the risks of infection amongst migrant workers accommodated in densely packed, often poorly sanitised dormitories.³⁴ As analysts have pointed out, with no vaccine and continued high demands placed on the public health system and services, the level of care available to regional migrant workers is likely to deteriorate in the coming months, and the situation could obviously be worse in the case of undocumented workers (US State Department, 2014; Human Rights Watch, 2018).³⁵ In all of these countries, undocumented migrant workers often lack ready access to local health care systems, and that access may be further hampered as spiralling demands on those systems force governments to limit who receives health care.³⁶ The post-COVID-19 surge of nativist and nationalist political rhetoric (e.g. Malaysia, as discussed below) have also made migrant workers easy targets for exclusion from further access to services or, worse, stigmatisation as an infection risk ('Locked up in Malaysia's Lockdown', 2020). As a result, migrant workers have had to use their savings, if any, to pay medical bills.

Long-Term Health Impact

Even before the pandemic, healthcare in the ASEAN region was of variable quality. Maternal mortality in some countries (e.g. the Lao PDR and Myanmar) was 10 times higher than in others (e.g. Brunei Darussalam, Thailand, and Singapore). While Southeast Asia had made considerable progress in its fight against malaria, it lagged in eliminating tuberculosis, with new and relapse cases of tuberculosis still affecting 300 in 100,000 people, double the average rate in Asia and the Pacific. Neglected tropical diseases constituted a burden on the region's development, with 31.3% of the population (more than 200 million people) requiring interventions in the form of large-scale preventive drug treatment or individual treatment. Child malnutrition in the region is also a complicated issue, with more than one-third of children in some countries malnourished, while the

³³ Most migrant workers engaged in the illegal fishing sector in Thailand moved out of their homes largely out of concern about heightened risks of poor health, either to themselves or someone in the household.

³⁴ Thailand allows migrants to access health care like all citizens; however, this situation is challenging in the case of undocumented migrant workers. Similarly, in Malaysia the high number of undocumented migrant workers and refugees (who may also work illegally) makes the provision of health care difficult.

³⁵ Thailand is known to be a source, transit, and destination country for forced labour, particularly in the fishing, service, and sex sectors.

³⁶ Migrants include irregular migrants and those trafficked into other countries.

proportion of overweight children has tripled since 2000. In this context, COVID-19 has the potential to disrupt health services in several different ways; through the health system becoming overwhelmed with COVID-19 patients, through interventions used to slow the transmission of COVID-19 inhibiting access to preventative interventions and services, and through the interruption of medical supplies (Hogan et al., 2020).

According to a 2020 study by the Imperial College London COVID-19 response team, such service disruptions in high-burden settings worldwide could increase deaths related to HIV, tuberculosis, and malaria over 5 years by up to 10% (for HIV), 20% (for tuberculosis), and 36% (for malaria) (Hogan et al., 2020). The greatest impact on HIV deaths will likely be from interruptions to antiretroviral therapy, which may occur at times of high or extremely high demand on health systems; and for tuberculosis, the greatest impact is likely to be from reductions in the timely diagnosis and treatment of new cases, which may result from a long period of COVID-19-suppression interventions. New modelling on HIV by the WHO and Joint United Nations Programme on HIV/AIDS also highlights the importance of taking immediate steps to minimise interruptions in health services and supplies of antiretroviral drugs during the pandemic (WHO, 2020a).³⁷

While there is limited data on COVID-19 infection in tuberculosis patients, those ill with both will likely have poorer treatment outcomes, especially if tuberculosis treatment is interrupted. Tuberculosis patients should take precautions against COVID-19 as advised by health authorities, and continue their treatment as prescribed (WHO, 2020b).

Maintaining critical prevention activities and health care services for HIV, tuberculosis, and malaria could significantly reduce the overall impact of the pandemic and contribute towards the progress of this SDG and its targets.

Box 7: Noncommunicable Diseases

According to a 2019 World Health Organization (WHO) report on noncommunicable diseases (NCDs) and the coronavirus disease (COVID-19) (September), COVID-19 poses a high risk to both older people and people with pre-existing NCDs, include cardiovascular diseases, chronic respiratory disease (e.g. chronic obstructive pulmonary disease), diabetes, and cancer. According to global data, death rates in COVID-19 patients with pre-existing conditions such as cardiovascular disease (13.2%), diabetes (9.2%), chronic respiratory disease (8%), hypertension (8.4%), and cancer (7.6%) exceed that of patients without coexisting conditions (0.9%) (Guan et al., 2020).

Noncommunicable diseases – mainly cardiovascular diseases, chronic respiratory diseases, diabetes, and cancer – are top killers in the WHO South-East Asia Region,³⁸ claiming an estimated 8.5 million lives each year (WHO, n.d.a.). One-third of these deaths are premature and occur before the age of 70, thus affecting economically productive individuals. The four most common NCDs are largely caused by four modifiable behavioural risk factors: tobacco use, unhealthy diet, insufficient physical activity, and harmful use of alcohol. NCDs disproportionately affect the poor, impoverish families, and place a growing burden on healthcare systems.

According to Thakur et al. (2020), many people with NCDs are expected to experience restricted

³⁷ Service disruptions related to COVID-19 could cause hundreds of thousands of extra deaths from HIV.

³⁸ The WHO South-East Asia Region includes only Indonesia, Myanmar, and Thailand.

mobility due to lockdowns or a lack of transportation, which will affect their ability to access health services during the COVID-19 pandemic. In addition, restrictive measures such as lockdowns, social distancing, and travel restrictions to reduce the spread of infection in many countries may also impact people living with NCDs by limiting their activity, hampering their ability to secure healthy foods and access preventive or health promotion services. Without proper management, chronic medical conditions can worsen due to stressful situations resulting from restrictions, insecure economic situations, and changes in normal health behaviours. Disruptions in routine health services and medical supplies can also increase morbidity, disability, and avoidable mortality amongst NCD patients (Thakur et al., 2020).

Source: Authors.

2.4. Sustainable Development Goal 4: Ensure Inclusive and Equitable Quality Education and Promote Lifelong Opportunities for All

Migrants often have difficulty providing their children with quality education because of a variety of barriers, including cost, access, and discrimination. Deprived of a good education, these children grow up with several disadvantages that in turn keep them in the same conditions of poverty as their parents. Ensuring the education of these children is essential to meet the targets of SDG 4 in the ASEAN region.

Migration-relevant targets to be achieved by 2030 include the following:

- By 2030, ensure that all girls and boys complete free, equitable, and quality primary and secondary education, leading to relevant and effective learning outcomes (Target 4.1).
- (ii) By 2030, ensure equal access for all women and men to affordable and quality technical, vocational, and tertiary education, including university (Target 4.3).
- (iii) By 2030, ensure that all youth and a substantial proportion of adults (both men and women) achieve literacy and numeracy (Target 4.6).
- (iv) By 2020, substantially expand globally the number of scholarships available to developing countries (especially least developed countries, small island developing states, and African countries) for enrolment in higher education, including vocational training; information and communication technology; and technical, engineering, and scientific programmes in developed countries and other developing countries (Target 4.b).

Impact of the Coronavirus Disease Pandemic

A study of AMS showed that, while the impact of worker migration on children is not always negative, it often results in them being deprived of a normal schooling and childhood (Mortensen, 2016). Migrating parents often leave school-aged children with elderly grandparents or other extended family members, and children therefore may lack the necessary supervision to attend school. These children often must work to meet shortfalls in household income (resulting in increased child labour), and girls especially are required to take on caring roles previously done by the migrant mother. Children not in school are often easy targets for trafficking. Children left behind in a sending country also have high rates of non-attendance at school between the ages of 10 and 17 years, and those in school often show no substantial improvement in educational performance. Children who accompany their parents to their destination countries may find it difficult to access educational infrastructure as they are often not included in school enrolment campaigns by local authorities, for example, because of a lack of proficiency in the local language or non-recognition of grades completed in their home towns. Children who do enrol in a state school may be unable to learn in their native language, and households may find it difficult to bear the costs of uniforms and/or books. (Altbach and de Wit, 2018).

The COVID-19 crisis is likely to exacerbate existing inequalities in access to quality education (e.g. through the lack of adequate information technology infrastructure and facilities to engage in online learning) in the ASEAN region, and children of vulnerable groups such as migrant workers are likely to be impacted most by the COVID-19 crisis (Jalli, 2020). This reality will affect populations from low-income, labour-exporting countries like Cambodia, the Lao PDR, Myanmar, the Philippines, and Viet Nam (and to an extent Indonesia), where access to good-quality learning institutions has been a problem for some time (Altbach and de Wit, 2018). In ASEAN's developing countries, primary school-age children from households in the poorest quintile are almost three times more likely to be out of school than those in the richest quintile, and the childhood mortality rate for the poorest quintile is two to three times higher than for the richest quintile.

During the COVID-19 crisis many of these countries observed significant increases in school drop-out rates and the incidence of child labour because of unaffordable school costs and a lack of adequate food (Nortajuddin, 2020c).³⁹ In addition to the impact on their growth and development, children not in school are easy targets for trafficking (both domestic and across borders) (US State Department, 2020; US Department of Justice, n.d.).⁴⁰

School Closures

The Declaration of the Special ASEAN Summit on COVID-19 (virtual) held on 14 April 2020 reaffirmed ASEAN's commitment to take collective action and coordinate policies in the fight against the pandemic. In doing so, ASEAN leaders acknowledged that hard-won gains in expanded access to education in the region since the 1990s could stagnate or reverse as school closures are extended, and access to alternative options like online or distance learning remain out of reach for those without means to connect. This may cause further losses in human capital and diminish economic opportunities. The statement included special action plans for school-going children and education ('The Fourth Industrial Revolution', 2020). Nonetheless, schools were closed for a period of time, sparking concern about the education sector in general and rural schools in particular.

³⁹ The World Bank has stated that Cambodia's economy is likely to shrink by 1.0%–2.9% this year, the worst performance for the country in a quarter of a century. The COVID-19 crisis will put 1.76 million jobs at risk as a result of losses in tourism, manufacturing, and construction, which together account for more than 70% of growth and 40% of employment (World Bank, 2020a).

⁴⁰See <u>www.ecpat.org</u>. The US Department's Trafficking in Persons Report 2020, which cites different issues that occurred during 2020, shows that child trafficking flourished despite the pandemic.

While school closures seem to be a logical solution to enforce social distancing within communities during the pandemic, educationists and activists assert that prolonged closures tend to have a disproportionately negative impact on the most vulnerable students, particularly those living in rural areas and those with disabilities. They have fewer opportunities for learning at home, and their time out of school may present economic burdens for parents who may face challenges finding prolonged childcare, or even adequate food in the absence of school meals.

After some initial hesitation, countries like Singapore, Malaysia, and Thailand (and to a lesser extent the Philippines and Indonesia) continued with internet (online) and distance learning, regardless of school closures ('Schools May Reopen', 2020; 'Thailand Schools Reopen', 2020; Su and Daga, 2020). Institutions of higher learning in these countries were also generally quite prepared to adapt to this new reality because of the rapid progress of technology since the mid-2000s and distance learning methods. However, rural areas faced technical problems along with difficulties providing the necessary household infrastructure for children to engage in online classes. This gap further increased as a result of the job and income losses experienced by households that depend on remittances. In Myanmar and the Philippines, there were reports of distress sales of items like mobile phones and laptops by households in both urban and rural areas.⁴¹

Most other countries and school systems (e.g. Indonesia, the Lao PDR, Myanmar, and the Philippines) were much less prepared to introduce e-learning methods, particularly for secondary education and below. Governments acknowledged that access to technology in most households varied, with shortfalls seen amongst poorer groups and especially amongst migrants. Programmes to target those most in need were lacking, and children from these households were likely to drop silently out of school (Jalli, 2020).

By August, schools in Malaysia, Singapore, and Thailand reopened with varying degrees of caution, but those in other countries remained closed. In countries like Myanmar, where health facilities are scarce, schools were turned into makeshift holding centres during the initial period of the crisis. The extent to which this temporary closure of schools impacted education is unknown. These issues must all be factored into planning, particularly during the coping and recovery phases when schools reopen post-COVID-19, and in addressing the education needs of children of migrant workers.

⁴¹ Discussions with U Maung Soe in Yangon (21 July 2020) and Ramon Bultron in Manila (21 August 2020).

Box 8: Risks of Increasing Child Labour

The crisis induced by the coronavirus disease (COVID-19) pandemic has already caused enormous geographical and social dislocation, loss of income, and disruption in the lives of migrants; and is expected to have a very negative impact on the education of their children, only compounding existing problems. In recognition of this, at the onset of the pandemic, the International Centre for Missing and Exploited Children and the Joining Forces and Child Rights Coalition issued the following collective statement: 'The COVID-19 pandemic is a child rights crisis in Asia. Due to school closures across the region, tens of millions of children have been forced into potentially unsafe home environments for weeks or months on end. We have received extremely worrying reports from several countries that domestic violence is on the rise. Governments of ASEAN and SAARC must put children's well-being at the centre of the pandemic response. Child protection services must be designated as essential and be given adequate resources to respond to reports of abuse'.⁴²

A particular concern was that prolonged closure of rural schools could result in large numbers of children dropping out of the system and becoming child labourers (Nortadujjin, 2020a). In early May 2020, the International Labour Organization and United Nations International Children's Emergency Fund (UNICEF) warned that the COVID-19 outbreak could create the first increase in child labour in the Association of Southeast Asian Nations region in more than 20 years unless urgent action were taken. This concern was particularly widespread amongst educationists and civil society groups engaged in protecting child rights (UNICEF, 2020).

Integrating child labour concerns across broader policies for education, social protection, justice, labour markets, and international human and labour rights makes a critical difference. According to a UNICEF brief, COVID-19 could lead to a rise in poverty and in child labour as a result, as households use every available means to survive (UNICEF, 2020). Some studies show that a one percentage point rise in poverty leads to at least a 0.7% increase in child labour in certain countries (Idris, 2020). This again underscores the importance of providing social protection to the most vulnerable in times of crisis (The Human Capital and Education for Asian Development Foundation, 2020).

Source: Authors.

2.5. Sustainable Development Goal 5: Achieve Gender Equality and Empower All Women and Girls

Female migrant workers not only face the same problems that male workers face, but are also subjected to gender-based discrimination. Migrant women and girls are also often subject to violence and exploitation at all stages of the migration cycle. This includes physical, sexual, or psychological violence both during transit (e.g. while travelling or in refugee camps) and at the final destination (e.g. by an employer).

Achieving SDG 5, which calls for gender equality and empowering all women and girls, should involve protecting the rights of migrant women, including domestic workers and combating all forms of trafficking of women and girls (Inter-Agency Coordination Group against Trafficking in Persons, 2018).

⁴² Statements by Amihan V. Abueva, Regional Executive Director, Child Rights Coalition Asia.

Migration-relevant targets include Target 5.5: ensure women's full and effective participation and equal opportunities for leadership at all levels of decision-making in political, economic, and public life.

Impact of the Coronavirus Disease Pandemic

Women account for nearly half of the intra-ASEAN migrant working-age population (UN Women, 2017), and form a substantial proportion of the workforce in sectors such as agriculture, personal care and healthcare, garbage collection, and cleaning services where employment conditions are poor.⁴³ Prior to the pandemic, undocumented women workers accounted for almost one-third of intraregional migrants. At least 30% of women migrant workers in Malaysia and Thailand are girls aged 15–24 (UN Women, 2017; Piper, 2011).

Although migration brings many socioeconomic benefits and some freedom from social norms that women normally experience in their home countries, women migrant workers confront restrictive social norms and laws, gender and racial discrimination, and gender-specific vulnerabilities in their host countries as well, limiting their opportunities for personal growth and access to benefits. Many migrant women are also employed in highly feminised sectors such as healthcare, domestic work, entertainment, manufacturing, and textiles in destination countries such as Malaysia, Singapore, and Thailand (Piper, 2011). Thus, women migrant workers are vulnerable not just because of their youth, which prevents them from being vocal about their rights because of constraints imposed by regressive cultural norms, but also because of the type of work in which they are engaged (e.g. domestic work, home-based health care, entertainment, and services) and a lack of adequate protection when they face challenges.⁴⁴

The nature of their employment, inability to work remotely, limited access to private transportation, physical proximity with co-workers and customers, and lack of adequate protective equipment and hygiene options make some of these 'women migrant-centric' occupations particularly risky. Many women workers who lacked decent working conditions before the COVID-19 crisis are now at high risk of getting sick because of a lack of access to health care, job loss, or substantial decreases in income (as employers tend to cut wages citing the pandemic). They also often lack safety and privacy, and face specific privacy concerns such as accommodation arrangements for live-in domestic workers.⁴⁵ At the same time they have limited access to social protection, as employment arrangements rarely follow prescribed standards (Guadagno, 2020). Several reports highlighted acts of overt and covert deception, coercion, and exploitation experienced by

⁴³ASEAN countries with ageing populations or high women's labour force participation (e.g. Malaysia, Singapore, and Thailand) have higher demand for domestic work and care for children and the elderly. Although more women are participating in the labour force, social gender norms still dictate that women care for the household, both in domestic work and caregiving. This presumption leads to a higher demand for migrant women, creating a 'global care chain' consisting of women migrant workers.

⁴⁴ Despite skilled female migrants being the fastest growing category of migrants, little attention has been paid to their experiences, particularly those working in the personal care and health care sectors.

⁴⁵ See, for example, Committee on the Elimination of Discrimination against Women Article 2: legal guarantees and penalties for violation. GR. No. 26, para. 17: lack of privacy and hygiene in working and living conditions, especially related to health.

women migrant workers (e.g. forced sex, noncompliance with contract provisions, extended work hours, and lack of social distancing) (ILO, 2020n).



Figure 3.6: Association of Southeast Asian Nations Women Migrants at Mid-Year

Lao PDR = Lao People's Democratic Republic.

Source: United Nations Department of Economic and Social Affairs (2017), *Trends in International Migrant Stock: The 2017 Revision* [Data Set]. New York, NY: United Nations Department of Economic and Social Affairs. <u>https://www.un.org/en/development/desa/population/migration/data/estimates2/estimates17.asp</u> (accessed 12 February 2021).

COVID-19 has exposed the influence of social norms on women's migration and decisions to take up certain kinds of work (Fleury, 2016).⁴⁶ Several studies have shown that many migrant women, often accompanying their spouses, start off as temporary workers before moving to seasonal or circular (during off-farm months) employment, quasi-permanent jobs, then 'better' or more socially valued types of employment (Women's Legal and Human Rights Bureau, 2017; ASEAN, 2012). For example, women migrants are often employed in domestic work either because they accompany their spouses to a host country and then seek local employment that is largely undocumented (and hence lack benefits and protection), or because they have moved from other jobs such as entertainment seeking 'better', or more socially valued types of work; this is reflected in lower pay and fewer labour regulations compared with other sectors (Fleury, 2016). For example, several ASEAN countries do not offer adequate protection for domestic workers within national labour laws. Some countries (e.g. Malaysia and Singapore) may include protections in national labour laws but invest little or no resources in enforcement. Enforcing the rights of domestic workers is particularly difficult since many live in their employer's home, away from public view (UN Women, 2017; 2020a; O'Neil et al., 2016). Gender-based discrimination of women migrant workers and structural inequalities were

⁴⁶Women are more likely to make migration decisions based on their family rather than individually. On the other hand, single mothers, widows, or divorcees who experience discrimination at home may use migration to escape social stigma. Domestic workers in the latter category reportedly confronted more workplace abuses during the pandemic and curfew period as they had little incentive or option to return home, leaving them at the mercy of their employers (Statement by Indonesian Migrant Workers Union, 24 May 2020).

widely seen during the COVID-19 pandemic. The pandemic has been especially challenging for female migrant workers, who have had to face gender barriers (e.g. access to health, privacy, and childcare facilities) on top of other general impacts and discrimination in terms of language, race, and cultural barriers (Fleury, 2016). With economic activity at a halt during the pandemic, employers across ASEAN initially laid off women workers in the informal sector in large numbers (around 33% of women workers between April and June 2020), resulting in a dramatic decline in their capacity to earn a living (ILO, 2020n; UN, 2020a; Women's Legal and Human Rights Bureau, 2017).⁴⁷ Secondly, rural women are more likely to migrate across borders if social networks are in place or if recruitment agencies can facilitate the process. With social networks virtually broken during the early days of the pandemic and lockdown, gender-based risks reportedly increased in the case of undocumented women workers (UN Women, 2020c; Peterman et al., 2020).

Between March and June 2020, there was no shortage of news reporting on migrant workers on the road, with the media describing them as the mass intraregional migrant worker exodus. Many migrant workers traveled on foot or different road transport facilities to reach the border. However, women migrant workers were invisible in these discussions, either because many stayed with their employers and did not return, or because their trauma is even less obvious than that of other categories of workers (Lavietes, 2020; United Nations Development Programme, 2020).

Surveys also show that fewer women than men are receiving information to prepare for COVID-19 (IOM, 2020b). For instance, in the Philippines, 79% of female respondents indicated they did not receive any information on the virus, compared to 57% of men. Assessments indicate that women experience increased barriers in accessing health care (especially amongst undocumented and sex workers who fear seeking help will lead to deportation) (UN Women, 2020c). As health care systems have been overwhelmed, the crisis has hit pregnant women and women with infants and young children particularly hard, disrupting access to health care services, medical supplies, and hygiene products.

⁴⁷ This estimate was presented during the early days of the pandemic, and was later affirmed by data from the ILO (3 July 2020).

Figure 3.7: Proportion of People Who Experienced Job Losses or Decreased Paid Work Hours since the Spread of Coronavirus Disease, by Sex (January–June 2020)



Source: United Nations Economic and Social Commission for Asia and the Pacific (2020), *Policy Brief: Impact of COVID-19 on Southeast Asia*. 30 July. Bangkok: United Nations Economic and Social Commission for Asia and the Pacific. https://www.unescap.org/sites/default/d8files/2020-07/SG-Policybrief-COVID-19-and-South-East-Asia-30-July-2020.pdf (accessed 30 August 2020).

Box 9: Impact on Trafficking

Prior to the onset of the coronavirus disease (COVID-19) pandemic, several regional studies concluded that migrant women domestic workers in the region are often victims of trafficking and forced labour. Most are poor, especially those who are single or widowed, and therefore particularly vulnerable to distress migration. Some factors underpinning this vulnerability are their lack of skills, awareness, income-generating opportunities, land, and assets, in addition to illiteracy and social inequalities due to nationality or ethnicity. National labour laws still do not protect many domestic workers, who are instead exposed to overly restrictive immigration laws and policies. Their isolation in private homes and a lack of information and support can lead to exploitation.⁴⁸ Human rights groups have largely blamed the heightened abuse and exploitation of migrant workers on outsourcing firms, who are also involved in the trafficking of persons for labour in Malaysia, Singapore, and Thailand (ECPAT International, 2016; Giammarinaro, 2020). Based on available reports from the International Labour Organization, other multilateral institutions, and civil society, it is evident that COVID-19 has impacted those forcefully trafficked into ASEAN in at least three ways: (i) heightened risks for those already working under exploitative conditions, (ii) increased risk of exploitation, including through child labour and sex work, and (iii) disruption of some of the affirmative actions and response efforts in progress prior to the onset of the pandemic.

⁴⁸ The SDGs' central reference to migration is made in Target 10.7: to facilitate orderly, safe, regular, and responsible migration and mobility of people, including through the implementation of planned and wellmanaged migration policies. This appears under SDG 10: to reduce inequality within and amongst countries. Other targets that directly reference migration mention trafficking, remittances, and mobility, amongst other things.

In Thailand and the Philippines, for example, civil society voices have expressed concern for sex workers as nightclubs shut and sex work is pushed underground, making workers increasingly vulnerable to lower pay, poorer working conditions, and exploitation. In the construction and tourism sectors, with economic activities contracting, an increased supply of workers could result in lower wages and exploitation. Past experiences indicate that the growing informalisation of women migrants heightens the risk of modern slavery, further reversing previous socioeconomic progress.

Finally, the social and economic disruptions caused by COVID-19 will fragment anti-trafficking response efforts in numerous ways. Many government and civil society response organisations have been impeded, and are finding it difficult to obtain the resources needed to sustain anti-trafficking efforts (including funding and attention). On the other hand, civil society organisations that provide critical protection, advocacy, and advisory support to those trafficked (such as shelters and reintegration programmes) are likely to be adversely affected as donors turn their attention elsewhere. This could further expose migrant women to abuse and exploitation.

Source: Authors.

2.6. Promote Sustained, Inclusive, and Sustainable Economic Growth; Full and Productive Employment, and Decent Work for All

The concept of 'decent work' in the 2030 Agenda is an important and positive move forward, steering policy debates beyond their focus on the quantitative aspects of job creation towards considering how the quality of new and existing work opportunities might also be enhanced (Võ Hải Minh, 2012; Umanath, 2020).⁴⁹ It features most centrally in SDG 8, which sets targets for job creation, social protection, social dialogue, and workers' rights, including those of migrant workers, in a manner that is consistent with environmental, social, and even economic imperatives. Migrants in particular, especially irregular migrants, should be given access to social protections, and efforts should be made to ensure the transferability and portability of these benefits.

Targets relevant to migration under SDG 8 include the following:

- (i) Promote development-oriented policies that support productive activities, decent job creation, entrepreneurship, creativity, and innovation; and encourage the formalisation and growth of MSMEs, including through access to financial services (Target 8.3).
- (ii) By 2030, achieve full and productive employment and decent work for all women and men, including for young people and persons with disabilities, and equal pay for work of equal value (Target 8.5).
- (iii) By 2020, substantially reduce the proportion of youth not in employment, education, or training (Target 8.6).
- (iv) Take immediate and effective measures to eradicate forced labour, end modern slavery and human trafficking, prohibit and eliminate the worst forms of child

⁴⁹ It is important to note that, within the policy context, the relationship between migration and employment tends to be discussed in terms of how job availability influences migration movements (and vice versa). Recent mass displacements have given further weight to this focus on the numbers. In countries such as Myanmar, the Philippines, and Viet Nam, new jobs are being created to counter domestic political pressures.

labour (including the recruitment and use of child soldiers), and end all forms of child labour by 2025 (Target 8.7).

- (v) Protect labour rights and promote safe and secure working environments for all workers, including migrant workers and women migrants in particular, as well as those in precarious employment (Target 8.8).
- (vi) Strengthen the capacity of domestic financial institutions to encourage and expand access to banking, insurance, and financial services for all (Target 8.5).

Impact of the Coronavirus Disease Pandemic

Worsening Work Conditions

Current unemployment rates and a prevailing pessimistic outlook on job creation in the short- to medium-term in ASEAN is a cause for concern.⁵⁰ The current phase of growing unemployment is not cyclical, and different estimates have asserted that unemployment will continue to increase over the next 5 years while many jobs lost during the pandemic may not return. The sectors that are expected to be affected most are tourism, services (including health care tourism), garments, and, to an extent, supply chains (logistics) ('ADB Says Cambodia to Lose', 2020). These are all sectors that employ a large proportion of migrant workers, who are likely to lose their jobs with no new investments or job opportunities in sight. This sudden loss of jobs for migrant workers not only has resulted in lost income, but also is likely to make competition for jobs more deeply entrenched, with significant wage cuts and social costs (Virgil and Lie, 2020; Diao and Wang, 2020).

Better recruitment and labour market practices built over the years, along with sustained advocacy and investments, have gradually eroded since the pandemic. Even prior to the pandemic, because of the high costs, long duration, and considerable complexity of navigating the existing bilateral channels for migration, many intra-ASEAN migrants were precariously employed in an irregular or quasi-regular status (Wickramarasekara, 2002). The pandemic has provided an additional opportunity for employers and agents to erode and undermine labour market practices deliberately (Satrusayang, 2020). Migrant workers who returned to work in June 2020 reported not only wage cuts (at times more cuts than those imposed on domestic workers), but also being asked to work more hours (against employment norms) to compensate for the losses incurred in April–May, without additional payment (Clean Clothes Campaign, 2020). Migrant workers in Thailand and Malaysia have reported that this situation had already resulted in wage cuts in June–July 2020 (ILO, 2020d). This situation was reportedly prevalent in export-dependent industries (e.g. textiles) where merchants could not execute pending orders (because of cancellations or delayed processing of documentation) and insisted on suspending minimum wages and certain work-related benefits, such as free meals (Business and Human Rights Resource Centre, 2020c; Clean Clothes Campaign, 2020). According to

In ASEAN, more young people than older people are unemployed, with five young people per one adult out of work, on average. Young people, moreover, are vulnerable to losing their jobs in volatile, short-cycle industries. The high proportion of young, unemployed workers creates disturbing problems for society, such as increased crime, increased numbers of drug addicts, and the cost to the state of dealing with these problems.

workers, such measures were subtly imposed on them, such as through the threat of massive layoffs, to avoid legal action (Business and Human Rights Resource Centre, 2020b). The ILO and UN Women have reported more such suffering amongst domestic workers and health care providers. In a related blog post, based on information received from different network members, the Business and Human Rights Resource Centre argued that migrant workers in supply chains faced further risks from inadequate and crowded living conditions, harsh virus containment measures, and discrimination. The situation was reportedly more dire in the garment, food, and beverage sectors, which employ proportionally more women (2020b).

Box 10: Irregular Workers

Irregular labour migration has emerged as a major issue affecting the management of intra-Association of Southeast Asian Nations migration, and more so during times of economic or political crises. Many workers become irregular in their host countries because they have overstayed their visas. Typical examples of irregular migrant workers are those who overstay their tourist visas and engage in work, students engaged in employment, trainees overstaying their visas, regular migrants continuing beyond the contract period, regular migrants running away from their designated employers before the expiry of their contracts, and persons trafficked into the sex industry. The seriousness of this problem led the Governments of Malaysia and Thailand to revise their visa policies several times since the 1990s. At present, no verifiable data are available on the regular versus irregular migration of workers. However, anecdotal data and estimates indicate that close to one-third of migrant workers could be treated as 'irregular' (Kassim and Zin, 2011). The irregular situation of such migrants, who comprise 30%–40% of all migrants in the region, puts them at the mercy of unscrupulous agents, employers, and officials and weakens their ability to seek redress (Orbeta and Gonzales, 2013).

Source: Authors.

Immigration status worsens the sufferings of migrant workers. Given the rise of outsourcing and off-shoring approaches, migrant workers across ASEAN often work as part of a complicated system of subcontracting and intermediaries, which hurts their rights as workers as well as the progress of the 'decent work' agenda. Recruitment agencies, who are more active in construction and domestic work, did not turn up to offer advice, information, or travel support to workers when the pandemic unfolded in March-April 2020 (ILO, 2018b). Despite several governments issuing statements on the automatic extension of work visas or arranging transport to return to border towns, the information networks on which workers rely failed to function, aggravating anxieties and concerns (Kuentak, 2020; Lindsey and Mann, 2020; Rajah and Yihan, 2020). On the other hand, considering the high level of bureaucracy and 'service fees' often demanded when making use of 'official' channels for labourers to work in another country, seeking employment via social networks might seem like a rational choice from a migrant's perspective. However, during the pandemic such arrangements seem to have partly made workers more vulnerable to abuse and exploitation (Orbeta and Gonzales, 2013).

The pandemic provided an additional opportunity for employers and outsource agencies to modify existing employment arrangements with irregular workers with less pay and fewer benefits, if any, using the threat of deportation in the case of resistance (Reich, 2020; United Nations Office on Drugs and Crime, 2015). There were unverified reports of unscrupulous manpower agencies engaging in coercive practices, including forced isolation, surveillance, withholding of payments, and threats of violence and of denunciation to authorities. In many cases, especially in the domestic and entertainment sectors, recruiters or employers reportedly withheld workers' identification documents and passports for fear of losing them (The Interpreter, 2020). The retention or confiscation of workers' identity documents typically affects international migrants, but can also involve those that do not cross international borders. This can restrict workers' freedom of movement, and be used as a means to bind them to a particular job or employer, forcing them to do work that they may not have originally consented to for fear of losing their documentsly, and consequently being deported to their country of origin (TF-AMW, 2020; 2018).

Labour agents thrived even during the pandemic, but have neglected to prioritise migrant workers' survival needs. In Indonesia in May–June, labour brokers reportedly deceived workers about the characteristics and terms of their potential employment, charging them fees of \$600–\$1,200 to secure them employment (US State Department, 2014). The Indonesian Migrant Workers Union asserted that recruitment agencies usually work through individual brokers who often operate in rural areas and lure people by offering well-paid jobs. In such cases, recruitment agencies' abusive practices are generally hidden under the surface, and the agencies may claim that they are not accountable for local brokers' noncompliance with agreements made with the workers. Such abuses were rampant as news of workers returning and the revival of economic activities surfaced in May 2020, highlighting the demand for labour. Fearing demands from workers, by early April most brokers' phones were either 'switched off' or busy.⁵¹

The work of labour institutions during the early phase of the crisis exposed serious limitations in labour governance (ILO, 2020m). As curfew measures were relaxed in mid-June, stranded migrant workers prepared to return to their home countries, whose economies were already fragile and could not absorb additional labour. The Indonesian Migrant Workers Union reported that labour institutions were completely unprepared and seriously under-resourced to help returning workers reintegrate, and those who remained in their host countries to seek work received little information on available opportunities. There was no recognition that helping returning migrants reintegrate could reduce socioeconomic tensions in their home countries, where some communities fear the transport of virus by returning workers. In addition, analysts have also noted that 'there was no scheme to reintegrate migrants so that returnees are not forced into accepting wage-cuts or efforts to re-migrate through illegal means' (Walden and Wijaya,

⁵¹ In an interview in early September, Dr Dominggus Li in Indonesia confirmed that labour agents were still 'hiding', fearing demands for lost wages by workers sent to work overseas. Dr Dominggus Elcid Li is the Executive Director of the Institute of Resource Governance and Social Change (2013 to present), an Indonesian think tank based in Kupang, East Nusa Tenggara, Indonesia.

2020; Chew, 2020; 'Region's Migrants Struggle', 2020). Poor or dysfunctional labour institutions in both sending and receiving countries impacted the capacity of labourers to express their grievances and undermined the strength and voices of labourers in every possible arena (ATUC, 2016b). According to the International Trade Union Conference's statement, immigrant workers entering through brokers faced extremely harsh conditions during April–June and later received lower wages than promised (Ang, 2020; Medina, 2020).

Even migrants who follow regular programmes may not find the protection of their guaranteed, since certain destinations such as Singapore group them into different categories: high-skilled labour is actively catered for and such workers may gain the right to abode, while migrants in low-skilled jobs can only get temporary contracts, and domestic workers are completely excluded from labour laws, since their occupation is not recognised as actual work (Kaur-Gill, 2020).

The private sector has a significant responsibility in progressing SDG 8. If the targets under this goal are to be achieved, the private sector must fundamentally change the way it operates. A move away from a model built upon maximising profit through the exploitation of labour to one built upon rights-based principles and approaches, as well as adherence to the ILO conventions will be key.⁵²

Private sector involvement in humanitarian and health crises is not a new phenomenon. For example, in the aftermath of the 2004 Indian Ocean earthquake and tsunami, several corporate groups provided financial aid to relief organisations and affected populations. Indeed, the private sector has almost always been willing to provide in-kind and cash donations in times of crisis. However, the impact of COVID-19 is on a much larger scale than previous outbreaks as it has seriously impacted every aspect of life and livelihoods. It is not geographically isolated, instead spreading across all of the AMS and beyond. The scale and reach of the disease have created a global supply chain crisis, with many AMS facing shortages of medical equipment such as surgical masks and ventilators. In this context, the private sector was expected to make efforts to manage the pandemic. While most medical companies are repurposing their factories and leveraging their comparative advantages and resources to help fill gaps in medical supply chains, others were forced to 'rethink' their plans and strategies. Regrettably, the private sector response during the crisis was not viewed favourably overall. Initially, several large corporations announced cash donations and medical supplies to governments as relief funds (CSIS, 2020; Kirschner, 2020; ILO, 2020k; TF-AMW, 2020). However, most factories and private sector operations announced the closure of activities and laid off workers without much warning, while cutting the wages of those

⁵² A discussion of the role played by the private health sector is outside the scope of this report. However, we note that ASEAN's private health sector is largely unregulated and was visibly absent in the initial weeks of the pandemic, leaving the provision of health services to state hospitals. By the end of April, the regional WHO office reported that the private sector was handling only one-tenth of the COVID-19 patient load, although it had two-thirds of the region's hospital beds and ventilators. This was a result of several factors, from restrictions implemented through government policy, to these hospitals 'playing it safe'. According to local media reports, private hospitals only began to provide emergency and patient care for those affected by COVID-19 in May 2020.
who continued to work (ILO, 2020k; TF-AMW, 2020).

There were also reports that the private sector in Thailand and Malaysia was not properly and timely distributing relief funds provided by the government or local charities meant for migrant workers, for example, by disallowing entry or timely distribution at construction sites or worker dormitories. There were also reports of pilferage and nondistribution (Uy, 2020). It is clear that, given their fragile immigration status, workers receive neither fair compensation for their efforts and productivity, nor relief in times of crisis to which they are entitled. Consequently, self-reporting and self-assessment of social responsibility by the private sector must end and be replaced by mandatory and transparent agency-by-agency reporting. The current model of global and regional labour supply chains is based on low wages and insecure – often unsafe – work. The challenge facing governments and civil society is how to push the private sector to do more to promote social and fiscal accountability and transparency in times of crisis.

Regular dialogue amongst stakeholders while undertaking relief efforts is essential to improve quality of outcomes (ILO, 2020d).⁵³ Between April and October 2020, AMS made several statements indicating the need for national- and regional-level dialogue and action (ASEAN, 2020e; 2020h; 2020l; 2020m). However, no multi-stakeholder social dialogue was held between the ASEAN governments and civil society groups; instead, governments issued 'orders and instructions' to be complied with. The absence of a multistakeholder platform during the early stages of the pandemic deprived the public of a better understanding of the evolving situation, restricted the exchange of information, and prevented the building of responsive networks to support migrant workers in the region.⁵⁴ Some countries with reasonably good public health systems (e.g. Malaysia, Singapore, and Thailand) should have responded to the crisis more effectively, but poorly designed crisis-redressal mechanisms and a lack of monitoring resulted in weak targeting and the circulation of misinformation, leading to panic buying and workers hurriedly traveling out of their work areas, amongst other things (Migrant Forum in Asia [MFA], 2020c).⁵⁵ Regional and local CSOs asserted that the absence of a clear crisis monitoring and stakeholder dialogue platform left governments ignorant as to what was happening on the ground, and governments continuing to treat the pandemic as a mere 'law and order' issue also did not help provide any good solutions (Migrante International, 2020). Better communication with civil society and local communities would have mitigated the forcible displacement of workers within a short timespan and non-collection of wages for work performed, while reducing deaths and the suffering of workers (MFA, 2020b; 2020c; World Council of Churches, 2020). On a positive note, prior to the pandemic, the ILO's work through the Triangle Program for ASEAN established a platform known as the ASEAN

⁵³ It is important to note that, of the four pillars of the Decent Work Agenda, social dialogue is the only one not explicitly recognised amongst the targets and indicators of SDG 8. Since the four pillars are equally important and mutually reinforcing, this omission is a notable oversight. with regard not only to SDG 8 but also to other goals, including ending poverty (SDG 1), advancing gender equality (SDG 5), reducing inequalities (SDG 10), and building more just and inclusive societies (SDG 16).

⁵⁴ On the other hand, ASEAN countries convened a couple of virtual gatherings to review the situation and issued solidarity statements.

⁵⁵ See Migrant Forum in Asia (2020c) for instances of 'wage theft' during the pandemic, based on anecdotal evidence.

Migrant Labour Forum, which meets annually to review progress on the implementation of compliance with international labour standards and treaties.⁵⁶ During the pandemic, such a forum can play valuable role in encouraging stakeholder engagement. One can only hope that going forward such platforms and efforts will offer a space to develop and strengthen dialogue amongst governments, the private sector, and civil society for a better future for migrant workers.

In sum, to achieve the targets under SDG 8, comprehensive national employment policy frameworks along with functioning labour institutions are needed to ensure that standards of employment of regional migrants in ASEAN match global requirements. The COVID-19 pandemic has exposed the poor progress made in policymaking and strengthening labour institutions in the region to support the provision of 'decent work' for migrant workers. Despite numerous efforts to engage in tripartite consultations including governments and civil society partners (e.g. employers and workers' representatives), strong and functioning labour market policies and institutions have not yet emerged in any meaningful way. Arbitrary wage cuts, extended work hours, poor benefits, and a lack of social protection programmes for migrant workers remain the norm.

This calls attention to the fact that workers in general and migrant workers more specifically have not emerged in the region as a powerful enough lobby to have their voices heard and demands fulfilled by those in government. Most AMS, with some exceptions, follow economic and social policies heavily skewed in favour of businesses and investors, often to the detriment of employees. For just and humane policies on migrant workers to be implemented with sincerity, international conventions alone may not be enough and in the long term, change will have to come from within ASEAN by assigning workers their rightful place as important contributors to economies and nation building.

2.7. Sustainable Development Goal 10: Reduce Inequality within and amongst Countries

SDG 10 intends to tackle inequality both within and amongst countries with respect to income inequality; social, political, and economic exclusion; discrimination; inequalities of opportunity and outcome; and reform of global governance.

With regard to migrant workers, SDG 10 wants governments to facilitate the orderly, safe, regular, and responsible migration and mobility of people, including through the implementation of planned and well-managed migration policies. Specifically, SDG 10.c aims to reduce the transaction costs of migrant remittances to less than 3% and eliminate remittance corridors with costs higher than 5% by 2030.

The targets set by SDG 10 also emphasise human rights, and operate as a lever to combat 'horizontal' inequality and the exclusion of particular groups, including women and racial

⁵⁶ The annual ASEAN Forum on Migrant Labour hosted by the ILO regional office is an open platform for the review, discussion, and exchange of good practices and ideas amongst governments, workers' and employers' organizations, and civil society stakeholders on key issues facing women and men migrant workers in Southeast Asia; and to develop recommendations to advance the implementation of the principles of the 2007 ASEAN Declaration on the Protection and Promotion of the Rights of Migrant Workers.

or ethnic minorities, as well as overall levels of economic inequality (i.e. disparities of income and wealth). For example, Target 10.1 calls for progressively achieving and sustaining income growth for the bottom 40% of the population at a rate higher than the national average by 2030. SDG 10.2 aims to empower and promote the social, economic, and political inclusion of all, irrespective of age, sex, disability, race, ethnicity, origin, religion, or economic or other status.

In early 2020, when COVID-19 hit ASEAN, many AMS were not on track to achieve the 17 SDGs by 2030, but were lagging behind in most of them (UNESCAP, 2020a). The 2020 UNESCAP Report on this subject observed that, although the region has done well in terms of economic growth over the last few decades (especially since 2010), it has neglected human development issues such as persistent high levels of inequality in select regions and amongst social groups, continued lack of access to social safety nets and welfare systems amongst poorer groups, low levels of social protection, and the continued growth of a large informal sector, as well as weak institutions of justice and ecological sustainability.

The preparation of medium- and long-term responses to the COVID-19 crisis provides AMS with an opportunity to rethink their priorities and to centre the welfare of poorer sections of the society and ordinary people, both domestic workers and migrants, in developmental policies. This will not only enable a speedy recovery by mobilising the contributions of all citizens, but will also ensure the sustainability of social safety nets and investments in the long term.

Migration-relevant targets under SDG 10 include the following:

- By 2030, empower and promote the social, economic, and political inclusion of all, irrespective of age, sex, disability, race, ethnicity, origin, religion, or economic or other status (Target 10.2).
- (ii) Ensure equal opportunity and reduce inequalities of outcome, including by eliminating discriminatory laws, policies, and practices; and promoting appropriate legislation, policies, and action in this regard (Target 10.3).
- (iii) Facilitate orderly, safe, regular, and responsible migration and mobility of people, including through the implementation of planned and well-managed migration policies (Target 10.7).
- (iv) By 2030, reduce the transaction costs of migrant remittances to less than 3%, and eliminate remittance corridors with costs higher than 5% (Target 10.c).

Impact of the Coronavirus Disease Pandemic

In the AMS, as in other regions, wide wage and income disparities and wealth inequality, in addition to poor living conditions, amongst sections of the population are the single major reason why many poorer and less educated sections migrate for employment, both within and outside the country.⁵⁷ Typically, many leave their homes because of a lack of sufficient fertile pastures, arable land, food, or water, as well as a surplus household labour force and lack of local employment opportunities and other fundamental

⁵⁷ In 1960, the income of the richest fifth of the world's population was, on average, 30 times higher than that of the poorest fifth. In 2015, it was 90 times higher.

requirements (Singapore Ministry of Manpower, 2020). The poorest often lack the means to escape war and poverty or obtain support from community networks. Once in a new country, migrants prefer to engage in a job right away and undergo a difficult and often conflict-driven process in the host community (Hein et al., 2015).

The slow economic progress across the AMS has also given rise to distorted social progress. Serious development problems, including poverty, inequality, and unemployment, lingered even in economically advanced Singapore (Võ Hải Minh, 2012). The COVID-19 crisis exacerbated these problems, which were prominently exposed in the sufferings of migrant workers.

ASEAN's 2001 Hanoi Declaration on Narrowing Development Gap for Closer Economic Integration acknowledged the uneven distribution of income and wealth both within the AMS and across the region, and resolved to promote effective measures to address these growing challenges (ASEAN, 2012). As of 2019, despite economic progress made by ASEAN since the 1990s, inequalities in income and wealth were severe and had been widening in both worker-sending and -receiving countries (see Table 3.1). Economic prosperity within the region is significantly uneven, and poverty remains an enduring challenge both within and amongst ASEAN countries.

However, regional economic gains have fallen short of erasing significant differences amongst the AMS. The AMS span a wide spectrum of income levels, from Singapore's GDP per capita of \$57,714 (as of 2018) to Cambodia's \$1,298. In recent years, lower-income states have made important gains. With respect to financial inclusion, the 2017 edition of the World Bank's Global Findex showed that, while 98% of adults in Singapore and 85% in Malaysia had a bank account, just 22% of Cambodian adults and 26% of Myanmar adults did (Demirgüç-Kunt et al., 2018).

Today, roughly two-thirds of the people in the region live in economically poor countries such as Cambodia, Indonesia, the Lao PDR, Myanmar, the Philippines, and Viet Nam. Women, indigenous peoples, ethnic groups, and rural communities in particular have not benefitted to the same extent from the region's macroeconomic gains or progress of their respective countries (ASEANStats, 2018). Large wage differentials also exist between AMS, with workers in Myanmar earning an average of \$91 per month and those in Singapore earning \$2,859 (Harkins and Lindgren, 2018). Given the wage cuts imposed after the pandemic, these gaps are set to increase rather than decline, which will negatively impact progress towards the targets under this SDG.

Available data indicate that, between 1990 and 2019, intra-ASEAN migration of workers, of whom 85% are either unskilled or low-skilled, increased many times over. Those migrating from one part of ASEAN to another obviously seek better terms of employment and wages.⁵⁸

⁵⁸ In general, the labour-receiving nations are Brunei Darussalam, Malaysia, Singapore, and Thailand; and the labour-sending nations are Cambodia, Indonesia, the Lao PDR, Myanmar, the Philippines, and Viet Nam. Thailand is also a labour-sending nation, with people traveling to work in the Middle East, Korea, Taiwan, and Japan.

| Country | Last | Previous | Reference | Unit |
|-------------|------|----------|---------------|------------------|
| Cambodia | 190 | 182 | January 2020 | \$/month |
| Indonesia | 3.94 | 3.6 | January 2019 | Rp million/month |
| Malaysia | 1200 | 1100 | February 2020 | RM/month |
| Myanmar | 4800 | 3600 | December 2018 | MK/day |
| Philippines | 537 | 537 | December 2019 | ₱/day |
| Thailand | 331 | 325 | December 2019 | B/day |

 Table 3.1: Minimum Wages in Association of Southeast Asian Nations

 Member Countries

Source: Trading Economics (2020), Minimum Wages – Asia [Data Set].

https://tradingeconomics.com/country-list/minimum-wages?continent=asia (accessed 1 September 2020).

The Asian Development Bank Institute's August 2020 briefing note on the consequences of the COVID-19 pandemic observed that 'the overall economic conditions of the host countries and their sectoral affiliations determine the impact, with severe losses of migrant jobs reported in the retail trade, manufacturing, hospitality and recreation, and accommodation and food service sectors. Workers have been laid off and, in many cases, stranded in host economies as strict quarantines have been imposed, borders closed, and air travel halted. Worse still, departures of new migrant workers from Asia have been put off until further notice' (Takenaka et al., 2020a).

In the long term, income gaps between countries are the most important driver of migration pressures, and the present crisis is not expected to lower the income gap sufficiently to reduce this. Instead, the crisis is likely to increase income inequality between low- and high-skilled workers (Ratha et al., 2020). Similarly, in the case of migrant workers within a country, the COVID-19 outbreak is likely to result in dire conditions, with many losing their (mostly informal) jobs and unable to return home because of disruptions to public transport services and movement restrictions. Lockdowns, travel bans, and social distancing measures in response to the crisis have disproportionately affected internal migrant workers, who have found themselves stranded, unable to return to either their places of work or their communities of origin. Without adequate access to housing, basic water and sanitation, health facilities, or social safety nets to help them survive such restrictions, these migrants have become even more vulnerable to both ill health and severe poverty.

Unless governments include internal migrants in programmes to provide health services, cash transfers, and other social programmes, the net result may be increased inequality between the rich and poor within AMS.

Box 11: Migration Policies

While Association of Southeast Asian Nations (ASEAN) member states have responded to the problems faced by migrant workers as part of their policy initiatives aimed at the larger economic and social crisis wrought by the coronavirus disease (COVID-19) pandemic, there is a need to maintain a holistic and longer-term perspective on migration governance. The current crisis can be an opportunity for governments to reassess their long-term migration governance and be better prepared for future crises (Giammarinaro and Palumbo, 2020).

In this context it is relevant to note that the Sustainable Development Goal (SDG) target 10.7.2 – which calls on countries to facilitate the orderly, safe, regular, and responsible migration and mobility of people, including through the implementation of planned and well-managed migration policies – is the target most explicitly and directly related to international migration amongst all the targets of the 2030 Agenda for Sustainable Development. The indicator aims to describe the state of national migration policies and how such policies change over time.

The International Organization for Migration (IOM) and the Population Division of the United Nations Department of Economic and Social Affairs have developed a methodology to measure this indicator based on an assessment of six policy domains: migrants' rights, institutional capacity, regional and international cooperation, migrants' socioeconomic well-being, mobility dimensions of crises, and safe and orderly migration (Migration Data Portal, 2020).

The Migration Governance Framework developed is being applied in a growing number of volunteering countries to help them develop baseline assessments and conduct future reviews of their work in the context of the SDGs. This is a gap analysis tool, not meant for ranking countries on their migration policies, but rather aiming to offer insights on policy levers that countries can action to strengthen their migration governance, and identify best practices for future programming.

Based on the available data, an analysis of SDG Indicator 10.7.2 indicates that more than half of all ASEAN governments (54%) report having a wide range of policies to facilitate the orderly, safe, regular, and responsible migration and mobility of people (Migration Data Portal, 2020). Additionally, 91% of governments indicate that they have an inter-ministerial coordination mechanism on migration in place, and 90% have bilateral agreements on migration with other countries.

Of the ASEAN Member States, only Cambodia, the Philippines, and Thailand are participating in the monitoring process, and all of them meet or fully meet the different criteria for achieving the target set by SDG Indicator 10.7.2 (United Nations Department for Economic and Social Affairs and IOM, 2019).

As is the case with most other countries, the analysis shows that these ASEAN members have room for improvement as policies are often not fully aligned with other important relevant policy domains such as sustainable development, disaster management, and climate change mitigation and adaptation (IOM, 2019).

Source: Authors.

Impact of the Coronavirus Disease on Remittances

Remittances form a substantial proportion of many nations' GDP in the region, and have historically been an important tool to help people in developing countries overcome poverty. For example, in 2018 the Philippines, as one of the world's largest recipients of remittances, received roughly 12%–16% of its GDP through this channel (ASEAN, 2020c; Bismonte, 2020); and remittances formed about 12% of Myanmar's GDP. These flows

have become the single most important source of foreign exchange to the economy (Figure 3.8) and a significant source of income for recipient families. Migrant remittances transferred to families in home countries directly become part of household budgets that can be spent on basic needs, serve as extra funds for increased consumption of durable and nondurable goods, or be used for savings. Remittances may also serve as capital for starting businesses. Thus, such cash flows from overseas raise the standard of living of recipient families (Takenaka et al., 2020b).



Figure 3.8: Top Remittance Recipients in the East Asia and Pacific Region (2019)

Source: Ratha, D.K., S. De, E.J. Kim, S. Plaza, G.K. Seshan and N.D. Yameogo (2020), 'COVID-19 Crisis Through a Migration Lens', *Migration and Development Brief*, No. 32. Washington, DC: World Bank Group. http://documents.worldbank.org/curated/en/989721587512418006/COVID-19-Crisis-Through-a-Migration-Lens (accessed 20 August 2020).

Although exact figures might be unknown for some time, it is clear that COVID-19 has already caused many migrant workers to lose their jobs and income while abroad and forced them to use any savings they may have had to survive for a period of time. This situation has, in turn, led to an inevitable decrease in remittances, which will negatively affect families and communities in sending countries acutely (Dang, 2020; Yayboke, 2020; ILO, 2020i).

The World Bank projects that remittance flows will decline across all regions as a result of the COVID-19 crisis, as follows: Europe and Central Asia (-27.5%); Sub-Saharan Africa (-23.1%), South Asia (-22.1%), the Middle East and North Africa (-19.6%), Latin America and the Caribbean (-19.3%), and East Asia and the Pacific (-13%) (Ratha et al., 2020). This decline comes after remittances to low-to-middle-income countries reached a record \$554 billion in 2019, overtaking foreign direct investments. In 2019, in current US dollars, the top five remittance-recipient countries were India (\$83.1 billion), China (\$68.4 billion), Mexico (\$38.5 billion), the Philippines (\$35.2 billion), and the Arab Republic of Egypt (\$26.8 billion) (Ratha et al., 2020). This drop in remittances could have a very significant impact on ASEAN's labour-sending nations. For example, in the case of the Philippines a conservative estimate forecasts that the COVID-19 pandemic will result in a drop of around \$10 billion (around 35 % below 2019 levels) in 2020, an enormous blow to the country's remittance-dependent economy (Moritsugu, 2020).

| - | | |
|---------------------------------------|------------------------|---------------|
| Remittance Recipients | Amount (\$ million) | % of Baseline |
| Global | -108, 617 | -18.3 |
| Asia | -54,255 | -19.8 |
| Australia and New Zealand | -299 | -10.8 |
| Central Asia | -3,366 | -23.8 |
| East Asia (excluding China and Japan) | -1,660 | -16.2 |
| Japan | -497 | -13.3 |
| China | -7,886 | -12.6 |
| Southeast Asia | -11,660 | -18.6 |
| South Asia | -28,621 | -24.7 |
| Pacific | -267 | -13.2 |
| United States | -482 | -7.4 |
| European Union + United Kingdom | -17,889 | -14.4 |

Table 3.2: Impact on Global Remittance Inflows (Worst-Case Scenario)

Note: Data presented for East Asia do not include those for China and Japan. Source: Asian Development Bank estimates.

Another ADB analysis assuming a worst-case scenario projected that ASEAN (including Asia and the Pacific) economies will likely take about a year to get their domestic outbreaks under control and bring economic activities back to normal; and, in parallel, estimated that remittances in Southeast Asia would fall by around \$11.7 billion, an 18.6% decline from 2018 (Takenaka et al., 2020b).

An analysis by the IFPRI's Myanmar Agriculture Policy Support Activity showed a major short-term contraction of remittances of 22%–27% into the country during April–May 2020, compared to the same period in 2019 (Diao and Wang, 2020). The report highlighted that the lockdown and subsequent restrictive measures have had direct and indirect negative impacts on the flow of goods and services, resulting in a decline of 41% in Myanmar's GDP during the 2-week lockdown in April 2020; this decline reportedly continued into May–June. The situation is no different in other countries in the region. For example, in the case of Indonesia, remittances from overseas workers declined by 20% by June 2020, and an IFPRI report suggested that national poverty would increase by 13 percentage points by the end of 2020 (Pradesha et al., 2020; Villafuerte and Takenaka, 2020). A panel discussion on migrant workers, organised by the IFPRI in collaboration with the FAO and others, asserted that this decline in remittances is partly linked to travel bans, lockdowns, and social-distancing rules, which reduce migrants' incomes or lead to unemployment (Paitoonpong and Chalamwong, 2012).

In addition, since the start of the lockdown, as commercial banks and money-transfer offices were not functioning at full scale, marooned migrants were unable either to send money overseas to their families or to receive cash from home to survive. This had a very negative impact on households, many of whom are reeling from job losses and economic hardship and need the money more than ever (Chalamwong, 2011).

Transaction Cost of Remittances

Reducing the cost of remitting funds by migrant workers is an important target under SDG 10. The objective of SDG 10.c is to help bring remittances into the formal economy, enhance financial inclusion, and increase the net income of receiving households. Globally, sending remittances costs 6.75% of the amount sent, on average (World Bank, 2020d). SDG 10 aims to reduce this global average to 3% by 2030 and the transaction costs of between any two corridors to less than 5%. Cutting prices by 5% age points is expected to result in savings of up to \$16 billion.

The cost of remittance services can vary substantially by region and transfer method. According to a World Bank study, the average cost of transferring \$200 to a developing country remained at 7% in Q1 2019, about the same as in previous quarters (Ratha et al., 2020). This is more than double the SDG target of 3% to be achieved by the year 2030. Banks are the most expensive route for sending remittances, with an average cost of 10.9%, while post offices are cheaper at 7.6%, although many poorer migrant workers use informal banking systems where costs are lower (Ahmed, Mughal, and Martinez-Zarzoso, 2020). Overall, based on data from formal remittance methods it is estimated that the transaction cost is the lowest in South Asia, at 4.98%, and the highest in Sub-Saharan Africa, at 8.47% on average (Ratha et al., 2020). According to a database maintained by the World Bank, the average cost of sending \$200 in remittances to East Asia and the Pacific dropped to 7.13% in Q1 2020, compared with 7.21% in Q1 2019 (2020d). The five lowest-cost corridors in the region averaged 2.6% while the five highest-cost corridors averaged 15.4% as of Q4 2019. Money transfer costs from Thailand to neighbouring countries in Southeast Asia were amongst the highest, averaging 12.1% in Q4 2019.

Because of the closure of many brick-and-mortar services, the use of digital channels for sending money is increasing (World Bank, 2020d). However, many migrant workers and their families back home who lack access to proper banking services face challenges in meeting the due diligence requirements of digital channels. In the short term, services that remain available are generally less expensive than those preceding the COVID-19 measures. Some remittance service providers have removed their fees and have been using social media to raise awareness of digital payment instruments. According to the World Bank study, it is important for remittance service providers and authorities to work together to mitigate the effects of the crisis and encourage the adoption of digital payments, greater use of regulated channels, and wider availability of cost-efficient services (World Bank, 2020d).





Source: Ratha, D.K., S. De, E.J. Kim, S. Plaza, G.K. Seshan and N.D. Yameogo (2020), 'COVID-19 Crisis Through a Migration Lens', *Migration and Development Brief*, No. 32. Washington, DC: World Bank Group. http://documents.worldbank.org/curated/en/989721587512418006/COVID-19-Crisis-Through-a-Migration-Lens (accessed 20 August 2020).

Chapter 4

Way Forward

The COVID-19 pandemic has been a global catastrophe, causing enormous damage to economies and severely affecting the day-to-day life of billions. While the number of infections in the ASEAN region is much lower than in many other parts of the world, the pandemic has significantly disrupted lives and there is no certainty as to when it will end. A severe economic contraction as a result of stringent restrictions on travel and regular social and business activities has left millions without work or sources of income. Those already vulnerable have been especially traumatised and will continue to be unless decisive steps are taken to support them.

Against this background, discussions have identified tardy progress in enforcing several labour laws, treaties, and guidelines critical for improved governance and better institutions. Given the extent of regional labour migration, it is inevitable that discussions on labour governance will include concerns about employment practices, labour safety, occupational health, and welfare systems. The ASEAN region needs to develop a response that helps everyone, especially the many migrant workers in the region, to build back better from the conditions before the crisis. Simply put, the challenge is to make policies and regulations work for society and not the other way around. There are many good examples of innovative, practical means of achieving this. However, this calls for the adoption and implementation of reforms at a certain speed and scale. General improvements in governance and institutions in tandem with the implementation of legal and policy guidelines for sustainable labour management could contribute to labour reforms.

Undoubtedly, this process will require large-scale resource mobilisation across the region. To build a cohesive response, greater attention will also have to be paid to several important initiatives undertaken by governments, international agencies, and civil society groups before the pandemic. This includes complying with various treaties to ensure the rights of migrant workers and enabling policies that allow them to live and improve their lives with dignity, as equals with all members of society.

In addition to its impacts on the economy, the pandemic is expected to strain regional cooperation initiatives. Scholars have thus called for sustained regional policy coordination on all socioeconomic matters to mitigate and isolate the pandemic shock (Kimura et al., 2020). Drawing from past experiences, Kimura et al. (2020) indicated the need for timely and greater coordination in ASEAN to mitigate the pending economic shock in terms of unemployment, corporate bankruptcy, and financial market fragility. In the absence of regional cooperation, the cost of the pandemic and economic shocks will increase significantly.

It is also clear that the resources required for recovering from the COVID-19 shock will take the form not only of funds or materials but also of non-tangible things such as trust, cooperation, and solidarity between governments and people of the region. The current

crisis offers an opportunity to build an ASEAN that is both economically vibrant and a world leader in upholding standards of human welfare and environmental protection.

1. Regional Measures to Soften the Impact of the Coronavirus Disease Pandemic

Coordination amongst AMS was slow in the initial stages of the COVID-19 outbreak, but there has since been increased policy convergence and a more united regional response. This is thanks to the functioning of existing mechanisms set up to deal with regionwide crises of any kind, particularly since the 1997 Asian financial crisis. For example, high-level ASEAN officials met as early as January 2020 to prepare a regionwide response, building on existing measures undertaken by individual AMS and regional health forums. The meeting pledged to strengthen coordination of national and regional efforts in ensuring ASEAN's readiness to mitigate and subsequently eliminate the threat of COVID-19 (ASEAN, 2020i).

Similarly, during the 26th ASEAN Economic Ministers' Retreat in Da Nang, Viet Nam on 10 March 2020, the regional group agreed to take collective action to mitigate the impact of the virus by working with external and development partners (ASEAN, 2020a). The ASEAN Economic Ministers issued a joint statement saying that they would continue to facilitate the smooth flow of goods and services in supply chains, and refrain from imposing new and unnecessary non-tariff measures (ASEAN, 2020m). Their statement focused on leveraging technology, digital trade, and trade-facilitation platforms such as the ASEAN Single Window, to foster supply-chain connectivity and allow businesses, especially SMEs, to continue operating during the pandemic.

ASEAN is also collaborating with external partners such as the European Union to mitigate the social and economic impacts of COVID-19, keep supply chains open, and advance relevant scientific research. It has also joined with the US to strengthen cooperation on public health measures. Experiences in tackling the pandemic were shared within the ASEAN+3, which includes China, Korea, and Japan.

In early April 2020, a Special ASEAN Summit and an ASEAN+3 Special Summit on COVID-19 issued a statement (ASEAN, 2020c) calling for a post-pandemic plan for ASEAN to (i) restore connectivity, tourism, normal business, and social activities to prevent potential economic downturns; (ii) ensure that critical infrastructure for trade and trading routes via air, land, and sea ports remain open; and (iii) refrain from imposing unnecessary restrictions on the flow of medical, food, and essential supplies. One result of these summits was the decision to establish the COVID-19 ASEAN Response Fund.

At its virtual meeting in June 2020, ASEAN leaders underscored the need to strengthen regional economic integration and enhance cooperation based on open markets that promote free trade and encourage investment, and an inclusive and rules-based multilateral trading system (ASEAN, 2020i). They reaffirmed their shared commitment to maintaining resilient supply chains for essential goods and to easing restrictions imposed during the COVID-19 pandemic.

The meeting also reaffirmed ASEAN's commitment to sign the RCEP in 2020 (ASEAN, 2016b), sending a clear signal of unwavering support for the multilateral trading system, regional integration, and economic development across the region. The meeting reaffirmed their shared commitment to supporting migrant workers' well-being across ASEAN in the context of the COVID-19 pandemic. An effort was made to chart a course towards early economic recovery from COVID-19 by harnessing potential opportunities and boosting community resilience.

2. Association of Southeast Asian Nations Comprehensive Recovery Framework

At the 37th ASEAN Summit, AMS leaders adopted the ASEAN Comprehensive Recovery Framework (ACRF) and Implementation Plan to serve as an exit strategy for the region to recover and build back better from the COVID-19 pandemic (ASEAN, 2020k). The ACRF, which takes a whole-of-community approach, focuses on key sectors and segments of society that are most affected by the pandemic, setting broad strategies and identifying measures for recovery in line with sectoral and regional priorities. Overall, ASEAN's recovery efforts will focus on five broad strategies that are deemed most impactful to take the region through the recovery process and its aftermath.

Broad Strategy 1 (enhancing health systems) prioritises building and sustaining current health gains and measures, maintaining and strengthening essential health services, and strengthening vaccine security. In addition, the quality of human resources for health will be enhanced by ensuring an adequate supply of trained, well-equipped, and healthy workers. The strategy also calls for enhancing the capacity of public health services to enable health emergency responses, including food safety and nutrition in emergencies. Broad Strategy 2 (strengthening human security) commits ASEAN to develop a recovery framework that puts the welfare of people at its core, by strengthening the protection and empowerment of all people and communities during the COVID-19 recovery period and beyond. To guide these efforts, the immediate priority is to enhance social protection, and strengthen food security, food safety, and nutrition for vulnerable groups. To overcome severe setbacks on human capital accumulation, digital skills and higher education, as well as re-skilling and upskilling for employment, will be promoted. To ensure the well-being of workers and improve the future of work, labour policies for 'the new normal' will be further strengthened, including through social dialogue. Mainstreaming gender equality will also be prioritised throughout the response and recovery process given the gendered impact of the pandemic. Lastly, the strategy calls for safeguarding human rights.

Broad Strategy 3 (maximising the potential of the intra-ASEAN market and broader economic integration) focuses on intensifying intra-ASEAN trade and investment, establishing ASEAN as a competitive market, and further increasing intra-ASEAN trade and investment to strengthen supply-chain resilience and regional value chains. Opportunities to maximise the potential of ASEAN's internal market will be explored while ensuring that the region remains open to opportunities presented by broader regional integration. New initiatives will be promoted, including the setting up of travel bubbles or corridors; strengthening of transport and regional connectivity; and revival of affected sectors such as tourism and MSMEs. The RCEP is expected to provide a strong anchor for achieving all these goals.

Broad Strategy 4 (accelerating inclusive digital transformation) leverages the momentum and imperative of digital transformation, and seizes the enormous opportunities presented by digital technologies to boost the economy and improve society in a post-COVID-19 world. Amongst other important priorities, promoting e-commerce and digital economy will be key, along with initiatives on e-governments and e-services, digital connectivity, the use of information and communication technology in education, and digital transformation of MSMEs. Data governance, cybersecurity, legal framework, and institutional capacity will also be enhanced.

Broad Strategy 5 (advancing towards a more sustainable and resilient future) concerns ASEAN's commitment to develop a recovery framework that is durable, long-lasting, inclusive, and capable of safeguarding the region's natural resources and social fabric, as well as the prosperity of its people. To support the attainment of this strategy, the priorities will focus on achieving sustainability in ASEAN in all dimensions, particularly in investment, energy, agriculture, green infrastructure, disaster management, and sustainable financing.

Given the comprehensive scope of the ACRF, resources will also be needed to support the implementation of initiatives and programmes. Some of this will be met via the newly set up COVID-19 ASEAN Response Fund. Additionally, existing funding programmes in the region will be re-purposed to meet current needs towards COVID-19 response and recovery efforts.

3. Association of Southeast Asian Nations Stimulus Packages

The steep fall in economic growth, accompanied by overall large-scale job losses is, of course, the primary challenge. Unless this is tackled, the region will find it very difficult to solve any of the other problems that have arisen. The good news is that, while 2020 has been a setback, ASEAN economies are projected to rebound sharply in 2021, matching or even outstripping growth in the years prior to the COVID-19 crisis. Much of this regional growth will be driven by the improvement in the global economic outlook, as well as the unprecedented monetary and fiscal stimulus packages announced by various ASEAN governments. Monetary policies adopted by several AMS were meant to ensure adequate liquidity and boost confidence in the economy when initial concerns on broken supply chains and travel cancellations emerged (Zulkhibri and Sinay, 2020). Governments in the region eased interest rates and provided government guarantees on (i) select bank lending activities, (ii) market interventions to improve temporary credit lines, and (iii) purchases of corporate bonds; and even temporarily relaxed regulatory measures for financial institutions (Okamura, Nguyen, and Doi, 2020).

Some measures to ensure monetary stability were at the regional level, building on financial cooperation initiatives first adopted in 2000, following the Asian financial crisis. Back then, finance ministries and central banks of ASEAN, together with those from Japan,

Korea, and China (ASEAN+3) had established the Chiang Mai Initiative, a network of bilateral swap agreements designed to protect member countries from future crises. In 2010, this was revamped as the CMIM, and a corpus of \$240 billion was set up to provide dollar liquidity support to reduce vulnerability to short-term external shocks.

Following the COVID-19 pandemic, ASEAN+3 finance ministers and central bank governors reached agreements to enhance the CMIM further, making it more effective and operationally ready for countries in need. This included the institutionalisation of the use of local currencies in addition to the dollar for CMIM financing, helping to broaden financing options for members in times of need.

| Country | Stimulus package | \$ billion equivalent | Date |
|----------------------|--------------------------|-----------------------|-------------|
| Brunei Darussalam | B\$450 million | 0.319 | 30 March |
| Cambodia | \$420 billion | 0.1 | 13 April |
| | ₹8.1 trillion | 2.0 | 14 April |
| Indonesia | Rp 10.3 trillion | 0.742 | 25 February |
| | Rp 120 trillion | 8.1 | 13 March |
| | Rp 22.9 trillion | 1.5 | 16 March |
| | Rp 62.3 trillion | 3.9 | 18 March |
| | Rp 405.1 trillion | 24.6 | 31 March |
| | Rp 641.2 trillion | 43.0 | 19 May |
| Lao PDR | K N30 billion | 0.005 | 2 April |
| Malaysia | RM20 billion | 4.8 | 27 February |
| | RM230 billion | 55.2 | 27 March |
| | RM10 billion | 2.4 | 6 April |
| Myanmar | MK100 billion | 0.101 | 27 April |
| Philippines | ₱27.1 billion | 0.527 | 16 March |
| | ₽200 billion | 3.8 | 31 March |
| Singapore | S\$6.4 billon | 4.6 | 18 February |
| | S\$48 billion | 33.7 | 26 March |
| | S\$5.1 billion | 3.4 | 6 April |
| | S\$33 billion | 23.3 | 26 May |
| Thailand | \$100 billion | 3.2 | 4 March |
| | \$117 billion | 3.6 | 24 March |
| | ₿1.9 trillion | 82.0 | 7 April |
| Viet Nam | <u>đ</u> 250 trillion | 10.8 | 6 March |
| | <u>đ</u> 62.0 trillion | 2.6 | 10 April |

Table 4.1: Summary of Association of Southeast Asian Nations Member States Stimulus Packages to Counter the Impact of the Pandemic

Lao PDR = Lao People's Democratic Republic.

Note: Based on information from various sources, including government websites and online news sites. Source: Zulkhibri, M. and J.B. Sinay (2020), 'Assessing ASEAN Economy Policy Responses in a Pandemic', *ASEAN Policy Brief*, No. 2. 28 May. <u>https://asean.org/storage/2020/06/ASEAN-Policy-Brief-02_31-May-2020.pdf</u> (accessed 28 October 2020). The size of the stimulus packages has ranged from 10% to 40% of GDP. Collectively, fiscal packages adopted by AMS amount to a combined \$318.2 billion or 10.1% of regional GDP in 2019. Thailand has the largest total stimulus package, with three instalments reaching \$88.8 billion, followed by Indonesia with \$81.8 billion.

While the fiscal stimulus packages adopted by AMS are primarily aimed at helping affected industries, a significant number of the measures are for the health sector and household subsidies, including cash allowances, electricity subsidies, and subsidies for social security contributions and pensions. These subsidies are crucial for workers to meet their daily needs and those of their households, especially those in low-income and vulnerable categories.

In most countries in the region, food packages have been provided to those who need them. The Philippines, for example, announced the reallocation of around \$3.9 billion in subsidies consisting of cash and basic necessities to 18 million low-income households over 2 months. In Singapore, citizens aged 21 and above will be given S\$900 (\$634.90) each, while self-employed persons will be paid S\$3,000 (\$2,116.20) in three tranches to supplement their incomes. Innovative measures such as 'rice ATMs' in Indonesia and Viet Nam and pay-it-forward food coupons in Thailand were also introduced. Beyond support for basic needs, other forms of subsidies, such as for electricity, fuel, pensions, and social security contributions, help lessen the financial burden on households.

| Country | Beneficiary | Origin or Destination Country | Intervention |
|-----------|---------------|-------------------------------------|---|
| Indonesia | International | Sending | The Indonesian migrant workers' protection agency, the National Board for the Placement and Protection of Indonesian Overseas Workers, facilitated the return of Indonesian migrants, both documented and undocumented, owing to lockdowns in Malaysia. |
| Malaysia | International | Receiving | The Government of Malaysia announced the introduction of an RM150 subsidy to employers for each foreign worker sent for coronavirus disease (COVID-19) screening starting 1 June 2020. To be eligible, foreign workers must be documented and contributing to the Social Security Organization. For workers not in the target group, the cost of COVID-19 screening should be borne by the employer to prevent an |

Table 4.2: Migration-Related Responses to the Coronavirus Disease within the Association of Southeast Asian Nations

outbreak amongst workers in the country.

- The government announced a 3-month grace period to allow employers to comply with amendments to the Workers' Minimum Standards of Housing and Amenities Act 1990, which addresses living conditions of migrant workers. Enforcement had been scheduled for 1 June 2020, but was delayed to 31 August. The amendment (passed in July 2019) extends the coverage of the act to housing and accommodation for workers in all sectors across Peninsular Malaysia and Labuan. Previously, the act was only empowered to govern such aspects for plantations of more than 20 acres (8.09 hectares) and the mining sector. Act 446, (amendment) 2019, also hopes to enhance the 2018 guidelines regarding accommodation for foreign workers prepared by the Peninsular Malaysia Labor Department. This covers a minimum standard for space required for workers' accommodations, basic facilities for housing and safety, and hygiene elements that employers must emphasise. At the end of August 2020, Malaysia's Human Resources Ministry published new rules covering minimum standards that employers must follow from September 2020 if they provide housing for their employees. Failure to do so will result in a fine of up to RM50,000 (Lai, 2020).
- Eligibility of undocumented migrants for COVID-19 tests and treatment has been extended.
- The foreign worker levy was cut by 25% for those with a permit ending by 31 December 2020, but the discount is not applicable to domestic helpers.
- Foreign workers can apply online to renew temporary employment visit passes that expire during the movement control order.

| Philippines | International | Sending | The Philippines Department of Labor and Employment recently announced that cash assistance amounting to \$200 (over ₱10,000) would be provided to overseas Filipino workers (OFWs) whose work was affected by the COVID-19 pandemic. The government aims to reach 70,000 OFWs. To access the assistance, OFWs must submit the certificate of employment issued by their agencies. Their applications will be evaluated and processed by the Philippine Overseas Labor Office or Overseas Workers Welfare Administration. Various government agencies have expressed their collective decision to provide ₱5 billion to support migrant workers from the Gulf and other destinations globally. |
|-------------|---------------|---------|--|
| Thailand | International | Sending | Thailand has just initiated a multi-pronged package for Thai workers forced to return from the Republic of Korea due to the outbreak. Those who are members of the Overseas Workers Fund will be entitled to B15,000 in compensation; this is also available to member workers who return from other countries that have announced a COVID-19 outbreak. The Thai Ministry of Labor will coordinate with labour offices in the Republic of Korea to make sure that Thai labourers receive pending wages and benefits. The Department of Employment has identified over 81,562 domestic jobs for Thai labourers returning from overseas. |



Sources: Lai, T. (2020), 'New Rules for Employees' Minimum Housing Standards from September 1, 2020. Employers to comply or be fined RM 50,000', *HRNEWS.my*. 1 September.

http://hrnews.my/2020/09/01/new-rules-for-employees-minimum-housing-standards-from-sept-1employers-to-comply-or-be-fined-rm50000/ (accessed 2 October 2020); Moroz, H., M. Shrestha, and M. Testaverde (2020), 'Potential Responses to the COVID-19 Outbreak in Support of Migrant Workers', *World Bank Working Paper*. Washington, DC: World Bank.

https://openknowledge.worldbank.org/handle/10986/33625 (accessed 12 August 2020).

4. Protecting Migrant Workers

The Joint Statement of ASEAN Labor Ministers on 'Response to the Impact of Coronavirus Disease 2019 (COVID-19) on Labor and Employment' pledges to support the livelihood and health of all workers, including migrant workers, by facilitating access to essential health care services, without discrimination (ASEAN, 2020h). In solidarity, AMS will support migrant workers affected by the pandemic in each other's country or in third countries to implement occupational safety and health standards and social protection systems effectively.

Several AMS have implemented policies that factor in the needs of vulnerable populations. Thailand will provide emergency welfare measures to migrant workers during the pandemic, providing them with free food and health services in select locations. Thailand also suspended immigration enforcement for overstaying migrants through the middle of November while international travel restrictions are in place. Some AMS have introduced economic stimulus plans and workplace measures to protect the health and income of workers during the pandemic. However, effective implementation of these measures will be key, and ASEAN often sees a significant gap between agreements and actual implementation.

Migration Governance Framework

In 2015, the IOM developed a Migration Governance Framework to clarify the purpose of SDG 10.7, which calls for 'well-managed migration policies'. It also developed Migration Governance Indicators (MGIs) to assess national frameworks and help to operationalise the framework. This framework is relevant for monitoring the social protections of migrant workers and tracking progress on certain SDGs.

The MGIs are a tool based on policy inputs that offers insights on policy levers that countries can use to develop their migration governance. Countries with a comprehensive migration governance framework are likely to be better prepared to manage any shock, including COVID-19. The MGI framework also includes information that can help assess countries' preparedness to ensure that no migrant is left behind, the risk of which is heightened during a crisis like the COVID-19 pandemic.

The six dimensions of migration governance included in the Migration Governance Framework and MGIs are outlined in Figure 5.1.

Figure 5.1: Six Dimensions of Migration Governance

Migrants' rights

Indicators in this domain assess the extent to which migrants have the same status as citizens in terms of access to basic social services such as health, education, and social security. This domain also includes the ratification of the main international conventions

Whole of government approach



institutional, legal, and regulatory frameworks related to migration policies. This domain also investigates the extent to which governments collect and use migration data.



Partnerships

This domain focuses on countries' efforts to cooperate on migration-related issues with other states and with relevant nongovernment actors, including civil society organisations and the private sector.

Mobility dimensions of crises This domain studies the type and level of preparedness of countries when they are faced with mobility dimensions of crises, linked to disasters, the environment, and/or conflict.

Well-being of migrants

This includes indicators on countries policies for m<mark>anaging the s</mark>ocioeconomic well-being of migrants, through aspects such as the recognition of migrants' educational and professional qualifications, provisions regulating student migration, and the existence of bilateral labour agreements between countries.

Safe, orderly, and dignified migration

This domain analyses countries' approach to migration management in terms of border contr and enforcement policies, admission criteria for migrants, preparedness, and resilience in the cas of significant and unexpected migration flows, a well as the fight against trafficking in human beings and smuggling of migrants. It also assesse efforts and incentives to help integrate returning citizens

Source: International Organization Migration (2019), Migration Governance Indicators: A Global Perspective. Geneva: International Organization Migration. https://publications.iom.int/books/migration-governanceindicators-global-perspective (accessed 5 August 2020).

5. **Need and Areas for Further Research**

Further research and in-depth documentation are needed to come up with long-term solutions to ensure humane policies that help migrant workers work and live with dignity, and to track the progress of relevant SDG targets and indicators in the ASEAN region. Currently, several grey areas exist because of a paucity of data and the lack of a nuanced understanding of the dynamics of migrant worker flows, the true impact of current policies, obstacles to the implementation of treaties, and a method for developing more effective future strategies.

For example, the effect of the COVID-19 crisis is likely to differ amongst the various subcategories of migrant workers and their families, including long-term, short-term, undocumented, returning, and women migrants. Going forward, AMS will have to work with international agencies and civil society to develop interventions targeting each of these subcategories specifically, to address their problems most effectively.

Such a disaggregated approach will help develop new policies or fine-tune existing ones to help anticipate and solve, ahead of time, many of the problems related to migrant workers that have emerged during the COVID-19 pandemic. For short-term or temporary migrants, who are often very low-skilled and have few social protections, the impact of sudden job losses on their immigration status and income can be drastic. Job loss as a result of the COVID-19 outbreak means loss of income for consumption, to remit home, and to repay the (often large) loans taken out to finance migration (Moroz et al., 2020). Typically, such workers live in shared accommodation or work in environments that are not conducive to social distancing or home-based work, making them more vulnerable to COVID-19. Moreover, loss of employment could also mean loss of legal status and risk of harassment by local authorities.

Undocumented workers, who migrate without legal documents, are another very vulnerable category. Not only are they often employed in jobs without social protection benefits, but their illegal status also makes them ineligible for any benefits that governments may provide. The UN estimates that globally, up to 20% of all migrants could be irregular (UN DESA, 2018). Reliable data on undocumented workers across the ASEAN region are lacking; this is yet another area that needs further research.

Returning migrants are another subcategory about which information is relatively sparse. COVID-19 disrupted economies and livelihoods, forcing many to return home after sustaining losses in terms of unpaid wages and depleted savings, without any great expectation of social support.⁵⁹ Apart from the health challenges posed during a pandemic outbreak by such large and chaotic population movements, such a sudden influx can also exacerbate various social tensions.

⁵⁹ Several ASEAN governments have taken measures to help returning migrants, including Indonesia, Myanmar, the Philippines, and Thailand. For example, Indonesia's National Agency for Placement and Protection of Indonesian Migrant Workers (BNP2TKI) has facilitated the return of both documented and undocumented migrants from Malaysia. In Myanmar, an emergency COVID-19 cash transfer targeting vulnerable groups, including internal and returning international migrants, will be financed by a consortium of international donors.

Figure 5.2: Areas and Themes for Further Research

Areas and Themes for Further R<mark>esearch</mark>

In the coming days, systematic studies will have to be undertaken to examine a range of themes covering social, economic, and political aspects related to regional labour migration and the progress of SDGs as also policies for migrant workers during the COVID-19 pandemic and beyond. Given below are a few possible areas of focus that can be taken up for further enquiry.

Economic

- Examine ways and means to mitigate structural economic inequalities that drive migration within ASEAN
- Develop a methodology and program to enhance cross-border investment programmes for helping migrants
- Examine ways to periodically develop schemes for upgrading the skills of migrants
- Advocate and support resource
 mobilisation by different stakeholders for dealing with migrant issues
- Monitor progress of migrant-specific
 economic stimulus packages and provide recommendations for improvement
- Assess and ensure food security in the context of migration
- Examine and define 'future of work and workplaces' (to monitor progress of, for
- example, SDG 10) Explore opportunities for providing migrant housing and addressing related issues
- Examine current level of safety and occupational health of migrant workers
- and propose ways to improve them

Social

- Examine obstacles to portability of migrant workers' rights and propose measures to overcome them
- Explore class, gender, ethnic, and religious fault lines in the context of labour migration
- Explore ways to deal with xenophobia in the context of migrant workers
- Identify different ways and opportunities to overcome gender discrimination
- Develop a road map for AMSs to implement and monitor compliance with human rights treaties and principles
- Strengthen prevention of trafficking of
- women and children during economic crises

Political

- Explore and map out potential threats to
 ASEAN economic integration from antiimmigrant sentiments
- Examine ways to eliminate anti-immigrant rhetoric likely to come up as part of domestic political competition
- Put in place reliable dispute-resolution mechanisms in the context of migration
- Develop mechanisms and tools to monitor the implementation of regional cooperation
- agreements on labour migration

Policy

 Support launch of thematic studies in all ASEAN countries on regional labour migration (and ensure systematic data collection and analysis) based on MiGOF and MGI. Currently only Thailand, Cambodia, and the Philippines are participating in such a monitoring process.

ASEAN = Association of Southeast Asian Nations, AMS = ASEAN member state, COVID-19 = coronavirus disease, MGI = Migration Governance Index, MiGOF = Migration Governance Framework, SDG = Sustainable Development Goal.

Source: Authors.

6. Looking Ahead

There is growing consensus amongst civil society groups and international organisations working in AMS that the region should adopt and enforce policies and programmes related to migrant workers that ensure respect for all human rights, the rule of law, and international labour standards. Safety and health, employment and decent work, wage protection, social security, and non-discrimination are key principles that must be upheld. Based on several consultations and reports published by concerned groups (MFA, 2020a) and international bodies (UNESCAP, 2020c), a set of recommendations are provided below for ASEAN governments of both origin and destination countries.

General Measures for the Association of Southeast Asian Nations

- (i) Strengthen social protection and welfare systems. Social safety nets, particularly those targeting regional migrant workers, must be reinforced and strengthened to protect migrants from the consequences of the COVID-19 outbreak, including income loss and health impacts.
- (ii) Develop and implement an action plan for rejuvenating livelihoods and reducing immediate economic stress. An action plan must be put in place to restore people's livelihoods and return economic activity to normal levels. Lost income reduces demand and worsens economic conditions, creating social conflict. Special employment-retention policies may be needed to keep all workers, including migrant workers, employed.
- (iii) **Strengthen supply chains.** Sustainable and resilient supply chains are necessary to supplement recovery efforts, both locally and globally. A key immediate step should be to reduce trade tariffs and open borders for goods to restart trade.
- (iv) Invest in targeted stimulus packages. As part of building back better, stimulus measures and long-term policy changes must focus on ameliorating deeply entrenched inequality in the region. Policies should be formulated to reduce inequalities in income, wealth, and access to basic services and social protection, with a special focus on vulnerable groups, such as people in the informal economy, women and girls, persons with disabilities, and migrants. Increased investments to strengthen health systems and accelerate progress towards universal health care will also help reduce inequalities significantly.
- (v) Emphasis on labour-efficient approaches. In the long run, ASEAN's overall productivity framework, which relies on the use of low-wage workers instead of more efficient technology, must be restructured. Such a change will need to be accompanied by measures to upgrade the skills of migrant workers and ameliorate economic and social conditions in origin countries.
- (vi) Advocate rights of workers. Upholding human rights and good governance practices will ensure that recovery in the region benefits all sections of the population and is sustainable. In destination countries, tensions between locals and migrants, exacerbated by the COVID-19 crisis, must be reduced. The rise of hate speech against migrants in particular needs to be tackled systematically before it results in further conflict.

(vii) Reinforce and bolster gender-focused actions. The return of women to the labour force must be supported. All forms of social protections must be extended to them, including measures to address protection from possible violence at the workplace and from unpaid care work in mid- to longer-term economic recovery plans.

Governments of Destination Countries

- (i) Advocate the adoption of inclusive labour policies at national and regional levels. ASEAN governments must be persuaded to include migrant workers' welfare in national policies or economic stimulus packages being implemented to help companies recover and sustain themselves during and post-lockdown. Many industries such as plantation, agriculture, and construction that traditionally depend heavily on migrant workers should therefore have equal access to financial benefits from the government as part of these stimulus packages.
- (ii) Encourage policies centred on rights. Adoption and enforcement of a comprehensive legal framework to protect migrant workers will help orderly progress on SDGs. To address the COVID-19 health crisis, governments should be persuaded to adopt and implement policies that are inclusive, equitable, non-discriminatory, and centred on human rights. This will enable migrant workers and their families, regardless of their status, to access free screening and treatment for COVID-19 without the fear of arrest and detention. The health and safety of migrant workers in the workplace must be ensured in keeping with WHO guidelines.
- (iii) Ensure workers' safety and health standards. Mechanisms must be developed to ensure that employers who provide dormitory accommodation to migrant workers guarantee the health and safety of workers living in that space in accordance with WHO guidelines.
- (iv) Ensure access to health services in case of COVID-19-related issues. Migrant workers who test positive for COVID-19 or are unable to work due to preventive quarantine and forced temporary business closures should continue to receive their full salaries, wages, and benefits. The COVID-19 crisis must not be used as an excuse by unscrupulous employers to lay off workers without just cause and without payment of earned salaries, wages, and benefits.
- (v) Ensure access to fair financial services. Procedures must be developed to guarantee that governments uphold a non-discriminatory approach to loan and credit card payment issues during COVID-19, such as deferring payments for both national and migrant workers. This will include recognising COVID-19 as a disease within labour laws and social security legislation, thus entitling migrant workers to claim compensation, medical, and allied care.
- (vi) Help streamline remittance procedures. As remittances are important to migrant worker households, governments should be encouraged to develop policies to address the fall in remittances likely to result from the COVID-19 outbreak.
- (vii) Disseminate information regularly. Migrant workers must be provided with correct and up-to-date information on government policies and regulations to combat COVID-19 in languages they understand, using various traditional and social media

platforms.

- (viii) **Urge the elimination of unfair practices.** Governments and employers must stop the discriminatory and arbitrary detention and deportation of migrant workers as a means to contain COVID-19. The human and labour rights of migrant workers must be respected at all times, even in the midst of the COVID-19 pandemic.
- (ix) Encourage protection of labour rights. Procedures must be developed to ensure that labour courts protect migrant workers' rights by passing preventive attachment orders against employers' assets. This is to ensure that migrant workers get their service benefits if the company closes or if an employer files a case against migrant workers in court.

Governments of Origin Countries

- (i) Boost migrant labour-related diplomacy and communication across the region. All AMS should use the current situation as an opportunity to ensure that diplomatic missions and consulates perform their duties to protect migrant workers from COVID-19 by disseminating correct and timely information to migrant workers in destination countries. In addition, all missions and consulates should reinforce their communication mechanisms to ensure the protection of the rights of migrant workers to communicate with their family and friends.
- (ii) Develop mechanisms for safe return transport. Governments should ensure the free, immediate, and humane repatriation of all migrant workers and their families stranded and/or deported from abroad. The remains of deceased migrant workers should also be returned to their families without delay, in accordance with WHO guidelines.
- (iii) Assure access to facilities and services. Once workers return home, governments should ensure that they have free and immediate access to health care including testing, quarantine facilities, and treatment for COVID-19. Returned migrant workers should be provided with free transportation, food, temporary housing, and quarantine facilities (if deemed necessary), until their return to their own homes in their origin countries.
- (iv) Recognise COVID-19 as an eligible disease for medical support. COVID-19 should be recognised as a disease in labour laws and social security legislation so that migrant workers can claim compensation, medical, and allied care.
- (v) Develop mechanisms to settle claims and grievances. A fair, accessible, and speedy mechanism must be set up for rescue, repatriation, settlement, and payment of migrant workers' claims, even after they have been repatriated to their origin countries.
- (vi) **Develop opportunities for employment in migrants' home countries**. Decent work opportunities should be provided to all, including returned migrant workers, in their home countries.

References

- Abella, M. and G. Ducanes (2009), 'The Effect of the Global Economic Crisis on Asian Migrant Workers and Governments' Responses', *Asian and Pacific Migration Journal*, 18(1), pp.143–61. doi:10.1177/011719680901800107.
- Abubakar, S.Y. (2002), 'Migrant Labour in Malaysia: Impact and Implications of the Asian Financial Crisis', East Asian Development Network Regional Project on the Social Impact of the Asian Financial Crisis. http://citeseery.ist.psu.edu/viewdoc/download2doi=10.1.1.455.2467&rep=rep18t

http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.455.2467&rep=rep1&t ype=pdf (accessed 7 April 2020).

- Abuza, Z. (2020), 'Explaining Successful (and Unsuccessful) COVID-19 Responses in Southeast Asia', *The Diplomat*. 21 April. <u>https://thediplomat.com/2020/04/explaining-successful-and-unsuccessful-covid-19-responses-in-southeast-asia/ (accessed 19 August 2020).</u>
- 'ADB Says Cambodia to Lose 390,000 Jobs Due to COVID-19' (2020), *The Phnom Penh Post.* 12 July. <u>https://business.inquirer.net/302695/adb-says-cambodia-to-lose-</u> <u>390000-jobs-due-to-covid-19</u> (accessed 19 August 2020).
- Agarwal, S. (2020), *The Political and Economic Impact of Covid-19 in Southeast Asia*. 11 June. Seattle; Washington, DC: National Bureau of Asian Research. <u>www.nbr.org/publication/the-political-and-economic-impact-of-covid-19-in-</u> <u>southeast-asia/</u> (accessed 11 July 2020).
- Ahmed, J., M. Mughal, and I. Martínez-Zarzoso (2020), Sending Money Home: Transaction Cost and Remittances to Developing Countries. Center for European, Governance and Economic Development Research (Discussion Papers 387).
 Göttingen: University of Göttingen, Department of Economics.
- Altbach, P., and H. de Wit (2018), 'Are We Facing a Fundamental Challenge to Higher Education Internationalization?', *International Higher Education*, 2(93), pp.2–4.
- Ang, J. (2020), 'More ASEAN Organisations Considering Pay Cuts and Restructuring Due to COVID-19'. *Human Resources Online*. 25 June.
 <u>https://www.humanresourcesonline.net/more-asean-organisations-considering-pay-cuts-and-restructuring-due-to-covid-19</u> (accessed 19 August 2020).
- Arezki, R. and H. Nguyen (2020), Coping with the Dual Shock: COVID-19 and Oil Prices. Washington, DC: World Bank. 14 April. <u>https://www.worldbank.org/en/region/mena/brief/coping-with-a-dual-shock-coronavirus-covid-19-and-oil-prices</u> (accessed 25 August 2020).
- Arisman, A. and R.K. Jaya (2018), *Protection of Human Rights and Labour Migration for Employment Purpose Across ASEAN.* Jakarta: Center for Southeast Asian Studies.
- Arisman, A. and R.K. Jaya (2020), 'Labour Migration in ASEAN: Indonesian Migrant Workers in Johor Bahru, Malaysia', *Asian Education and Development Studies*, 10(1), pp.27–39. https://doi.org/10.1108/AEDS-02-2019-0034
- Asadullah, M.N. (2020), 'Is ASEAN's COVID Response Leaving Migrant Workers Behind?', *The Bangkok Post.* 30 May.

https://www.bangkokpost.com/opinion/opinion/1926672 (accessed 6 June 2020).

Asian Development Bank (ADB) (2020a), 'Special Topic: The Impact of the Coronavirus

Outbreak – An Update', *Asian Development Outlook 2020*. April. http://dx.doi.org/10.22617/FLS200119-3

ADB (2020b), 'Food Security in Asia and the Pacific amid the COVID-19 Pandemic', ADB Briefs, No. 139. June.

https://www.adb.org/sites/default/files/publication/611671/adb-brief-139-foodsecurity-asia-pacific-covid-19.pdf (accessed 2 Feburary 2021).

- Association of Southeast Asian Nations (ASEAN) (n.d.), ASEAN Free Trade Area (AFTA Council). Jakarta: ASEAN. <u>https://asean.org/asean-economic-community/asean-free-trade-area-afta-council/</u> (accessed 21 September 2020).
- ASEAN (n.d.), Free Trade Agreements with Dialogue Partners. Jakarta: ASEAN. https://asean.org/?static_post=rcep-regional-comprehensive-economicpartnership (accessed 21 September 2020).
- ASEAN (2004), Vientiane Action Programme (VAP) 2004–2010 of 29 November 2004. ASEAN Documents Series of 2004. Jakarta: ASEAN. <u>https://www3.nd.edu/~ggoertz/rei/reidevon.dtBase2/Files.noindex/pdf/d/ASEAN</u> <u>-treaties-2004.pdf</u> (accessed 28 April 2020).
- ASEAN (2008), ASEAN Charter (adopted 15 December 2008). Jakarta: ASEAN. https://asean.org/asean/about-asean/overview/ (accessed 16 April 2020).
- ASEAN (2012), Ha Noi Declaration on Narrowing Development Gap for Closer ASEAN Integration (adopted 23 July 2001 in Ha Noi). 9 July. Jakarta: ASEAN. <u>https://asean.org/storage/2020/02/ASEAN-Chairmans-Statement-on-COVID-19-FINAL.pdf</u> (accessed 7 August 2020).
- ASEAN (2014), ASEAN Community in Figures Special Edition 2014: A Closer Look at Trade Performance and Dependency, and Investment. Jakarta: ASEAN. <u>http://www.asean.org/storage/images/ASEAN_RTK_2014/ACIF_Special_Edition_2_014.pdf</u> (accessed 7 August 2020).
- ASEAN (2016a), *The ASEAN Charter 18th Reprint*. Jakarta: ASEAN. <u>http://www.asean.org/wp-content/uploads/2012/05/11.-October-2015-The-</u> <u>ASEAN-Charter-18th-Reprint-Amended-updated-on-05_-April-2016-IJP.pdf</u> (accessed 7 August 2020).
- ASEAN (2016b), *Regional Comprehensive Economic Partnership.* 3 October. Jakarta: ASEAN. <u>https://asean.org/?static_post=rcep-regional-comprehensive-economic-partnership</u> (accessed 2 November 2020).
- ASEAN (2018a), ASEAN Consensus on the Protection and Promotion of the Rights of Migrant Workers. Jakarta: ASEAN. <u>https://asean.org/storage/2019/01/3.-March-2018-ASEAN-Consensus-on-the-Protection-and-Promotion-of-the-Rights-of-Migrant-Workers.pdf</u> (accessed 6 May 2020).
- ASEAN (2018b), ASEAN Launches Safe Migration Campaign: Top ASEAN, EU Officials Discuss Safe Labour Migration. 18 December. Jakarta: ASEAN. <u>https://asean.org/asean-launches-safe-migration-campaign-top-asean-eu-officials-discuss-safe-labour-migration/</u> (accessed 7 August 2020).
- ASEAN (2019), 'Progressive Realization Towards Universal Health Coverage: ASEAN Member States', *E-Health Bulletin*, 14(December). Jakarta: ASEAN. <u>https://asean.org/storage/2017/02/UHC-Bulletin-Dec-2019-FINAL.pdf</u> (accessed 16 July 2020)

- ASEAN (2020a), Strengthening ASEAN's Economic Resilience in Response to the Outbreak of the Coronavirus Disease (COVID-19). 10 March. Jakarta: ASEAN. <u>https://asean.org/storage/2020/03/AEMR-26-Statement-on-COVID-19-FINAL-</u> <u>10.03.2020.docx.pdf</u> (accessed 18 May 2020).
- ASEAN (2020b), ASEAN Policy Brief: Economic Impact of COVID-19 Outbreak on ASEAN. April. Jakarta: ASEAN. <u>https://asean.org/storage/2020/04/ASEAN-Policy-Brief-April-2020_FINAL.pdf</u> (accessed 18 May 2020).
- ASEAN (2020c), Policy Brief on the Impact of COVID-19 on the Economy. 10 April. Jakarta: ASEAN. <u>https://asean.org/storage/2020/04/ASEAN-Policy-Brief-April-</u> <u>2020 FINAL.pdf</u> (accessed 4 May 2020).
- ASEAN (2020d), Declaration of the Special ASEAN Summit on Coronavirus Disease 2019 (COVID-19). 14 April. Jakarta: ASEAN. https://asean.org/storage/2020/04/FINAL-Declaration-of-the-Special-ASEAN-Summit-on-COVID-19.pdf (accessed 6 June 2020).
- ASEAN (2020e), Joint Statement of the Special ASEAN Plus Three Summit on Coronavirus Disease 2019 (COVID-19). 14 April. Jakarta: ASEAN. <u>https://asean.org/jointstatement-special-asean-plus-three-summit-coronavirus-disease-2019-covid-19/</u> (accessed 4 May 2020).
- ASEAN (2020f), Statement of ASEAN Ministers on Agriculture and Forestry in Response to the Outbreak of the Coronavirus Disease (Covid-19) to Ensure Food Security, Food Safety and Nutrition in ASEAN. 16 April. Jakarta: ASEAN. <u>https://asean.org/statement-asean-ministers-agriculture-forestry-response-</u> <u>outbreak-coronavirus-disease-covid-19-ensure-food-security-food-safety-</u> <u>nutrition-asean/</u> (accessed 4 May 2020).
- ASEAN (2020g), ASEAN Pledges to Ensure Food Security during COVID-19 Outbreak. 17 April. Jakarta: ASEAN. <u>https://asean.org/asean-pledges-ensure-food-security-covid-19-outbreak/</u> (accessed 4 May 2020).
- ASEAN (2020h), Joint Statement of ASEAN Labour Ministers on Response to the Impact of Coronavirus Disease 2019 (COVID-19) on Labour and Employment. 14 May. Jakarta: ASEAN. <u>https://asean.org/joint-statement-asean-labour-ministersresponse-impact-coronavirus-disease-2019-covid-19-labour-employment/</u> (accessed 17 June 2020).
- ASEAN (2020i), ASEAN Leaders' Vision Statement on a Cohesive and Responsive ASEAN: Rising Above Challenges and Sustaining Growth. 26 June. Jakarta: ASEAN. <u>https://asean.org/storage/2020/06/Final-ASEAN-Leaders-Vision-Statement-on-a-</u> <u>Cohesive-and-Responsive-ASEAN-final.pdf</u> (accessed 10 July 2020).
- ASEAN (2020j), Chairman's Statement on ASEAN Collective Response to the Outbreak of Coronavirus Diseases 2019. 26 June. Jakarta: ASEAN. <u>https://asean.org/chairmansstatement-36th-asean-summit-26-june-2020-cohesive-responsive-asean/</u> (accessed 10 July 2020).
- ASEAN (2020k), Chairman's Statement of the 36th ASEAN Summit, 26 June: Cohesive and Responsive ASEAN. 27 June. Jakarta: ASEAN. <u>https://asean.org/chairmansstatement-36th-asean-summit-26-june-2020-cohesive-responsive-asean/</u> (accessed 10 July 2020).
- ASEAN (2020l), Joint Statement of the ASEAN Ministers Responsible for Information to

Minimize the Negative Effects of Coronavirus Disease 2019 (COVID-19). 7 August. Jakarta: ASEAN. <u>https://asean.org/joint-statement-asean-ministers-responsible-information-minimise-negative-effects-coronavirus-disease-2019-covid-19/</u> (accessed 20 September 2020).

- ASEAN (2020m), Joint Media Statement of the Thirty Fifth ASEAN Economic Ministers Meeting. 25 August. Jakarta: ASEAN. <u>https://asean.org/joint-media-statement-of-the-thirty-fifth-asean-economic-ministers-meeting-phnom-penh/</u> (accessed 20 September 2020).
- ASEAN (2020p), ASEAN Comprehensive Recovery Framework and its Implementation Plan. 12 November. Jakarta: ASEAN. https://asean.org/asean-comprehensiverecovery-framework-implementation-plan/ (accessed 10 December 2020).
- ASEAN and United Nations Entity for Gender Equality and the Empowerment of Women (UN Women) (2017), Women Migrant Workers in the ASEAN Economic Community. Jakarta: ASEAN and UN Women. <u>https://asiapacific.unwomen.org/-/media/field%20office%20eseasia/docs/publications/2017/06/aec-women-migration-study.pdf?la=en&vs=4122 (accessed 4 April 2020).</u>
- ASEAN Economic Community Strategy Center (2020), *Minimum Wages in ASEAN: How Are They Calculated?* Bangkok: University of the Thai Chamber of Commerce. 12 February. <u>http://aec.utcc.ac.th/minimum-wages-in-asean-how-are-they-</u> <u>calculated/</u> (accessed 7 May 2020).
- 'ASEAN Holds Special Summit on Response to COVID-19 Pandemic' (2020), *Business Standard.* 14 April. <u>https://www.business-standard.com/article/news-ani/asean-holds-special-summit-on-response-to-covid-19-pandemic-120041400460_1.html</u> (accessed 7 May 2020).
- ASEAN Single Window (2020), ASW&NSW News and Updates. Jakarta: ASEAN. https://asw.asean.org (accessed 2 November 2020).
- 'ASEAN's Virus Dilemma' (2020), *Bangkok Post.* 3 August. Jakarta: ASEAN Secretariat. <u>https://www.bangkokpost.com/opinion/opinion/1961579/aseans-virus-dilemma</u> (accessed 21 August 2020).
- ASEANStats (2018), ASEAN Statistical Highlights 2018. Jakarta: ASEAN Secretariat. <u>https://www.aseanstats.org/publication/asean-statistical-highlights-2018/</u> (accessed 7 September 2020).
- ASEANStats (2019), ASEAN Key Figures 2019. Jakarta: ASEAN Secretariat. <u>https://www.aseanstats.org/wp-</u> <u>content/uploads/2019/11/ASEAN_Key_Figures_2019.pdf</u> (accessed 7 September 2020).
- ASEANStats (2019), ASEAN Statistical Yearbook 2019. Jakarta: ASEAN Secretariat. <u>https://www.aseanstats.org/wp-content/uploads/2020/10/ASYB_2019-</u> <u>rev20201031.pdf</u> (accessed 24 February 2021)
- ASEAN Secretariat (2019), ASEAN Integration Report 2019. November 2019. Jakarta: ASEAN Secretariat. <u>https://asean.org/storage/2019/11/ASEAN-integration-report-</u>2019.pdf (accessed 7 May 2020).
- ASEAN Trade Union Council (ATUC) (2016a), *ASEAN Social Charter*. Kuala Lumpur: ATUC. <u>http://www.asean-socialcharter.net</u> (accessed 8 September 2020).

- ATUC (2016b), Strengthening and Harmonizing Labour Standards in ASEAN: A Framework for Union Advocacy. Kuala Lumpur: ATUC. <u>http://aseantuc.org/wpcontent/uploads/2017/08/atuc-asean-labor-standards-online.pdf</u> (accessed 4 October 2020).
- ATUC (2019a), ASEAN Consensus on the Protection and Promotion of the Rights of Migrant Workers. 5 April. Kuala Lumpur: ATUC. http://aseantuc.org/wpcontent/uploads/2014/12/ATUC-ACE-Joint_Statement_FV_05-Apr-2019.pdf (accessed 10 May 2020).
- ATUC (2019b), Joint Statement of the ASEAN Trade Union Council and the ASEAN Confederation of Employers on the ASEAN Consensus on the Protection and Promotion of the Rights of Migrant Workers. 5 April. Kuala Lumpur: ATUC. <u>http://aseantuc.org/wp-content/uploads/2014/12/ATUC-ACE-</u> Joint Statement FV 05-Apr-2019.pdf (accessed 7 May 2020).
- ATUC (2020), *Realizing the Rights of Migrant Workers to Social Protection.* Kuala Lumpur: ATUC. <u>http://aseantuc.org/wp-content/uploads/2020/06/realizing-rights-of-migrant-workers-to-social-protection-online-2020.pdf</u> (accessed 4 October 2020).
- Bagus Enrico and Partners (2020), *Impact of COVID-19 upon Employment Sector in Indonesia*. International Comparative Legal Guides. 10 June. Jakarta: Bagus Enrico and Partners. https://iclg.com/briefing/13306-impact-of-covid-19-upon-employment-sector-in-indonesia (accessed 12 July 2020).

Bahadur, A., E. Lovell, E. Wilkinson, and T. Tanner (2015), *Resilience in the SDGs.* August. London: Overseas Development Institute (ODI). <u>https://www.odi.org/sites/odi.org.uk/files/odi-assets/publications-opinion-files/9780.pdf</u> (accessed 7 September 2020).

- 'Bangkok Tightens Scrutiny of Construction Site Workers for COVID-19', (2020), *The Nation Thailand*. 29 April. <u>https://www.nationthailand.com/news/30386977</u> (accessed 7 May 2020).
- Barrero, J., N. Bloom, and S. Davis (2020), 'COVID-19 Is also a Reallocation Shock', *Becker Friedman Institute for Economics Working Paper*, No. 2020-59). May. Chicago, IL:
 Becker Friedman Institute for Economics. https://bfi.uchicago.edu/wp-content/uploads/BFI_WP_202059.pdf (accessed 5 September 2020).
- Behar, A. (2020), 'A Pandemic Is not a Law and Order Problem', India Development Review. 30 May. https://idronline.org/a-pandemic-is-not-a-law-and-orderproblem/ (accessed 14 July 2020).
- Bismonte, C. (2020), 'The Disproportionate Effect of COVID-19 on Migrant Workers in ASEAN', *The Diplomat*. 22 May. <u>https://thediplomat.com/2020/05/the-disproportionate-effect-of-covid-19-on-migrant-workers-in-asean/</u> (accessed 16 July 2020).
- Buller, A.M., V. Vaca, H. Stoklosa, R. Borland, and C. Zimmerman (2015), Labour Exploitation, Trafficking and Migrant Health: Multi-Country Findings on the Health Risks and Consequences of Migrant and Trafficked Workers. Geneva: International Organization for Migration (IOM) and London School of Hygiene and Tropical Medicine.

https://publications.iom.int/system/files/pdf/labour_exploitation_trafficking_en.p

df (accessed 7 May 2020).

- Business and Human Rights Resource Centre (BHRRC) (2020a), *Cambodia and Myanmar: The Outbreak of the COVID-19 Affects Jobs in the Global Supply Chain, with Cambodia and Myanmar Being Most Affected.* 13 March. London: BHRRC. <u>https://www.business-humanrights.org/en/latest-news/cambodia-myanmar-the-outbreak-of-the-covid-19-affects-jobs-in-the-global-supply-chain-with-cambodiaand-myanmar-being-most-affected/ (accessed 7 May 2020).</u>
- BHRRC (2020b), Cambodia: Over 2000 Workers of Chinese Garment and Footwear Companies Protest to Demand Unpaid Wages. 15 June. London: BHRRC. <u>https://www.business-humanrights.org/en/cambodia-over-2000-workers-ofchinese-garment-and-footwear-companies-protest-to-demand-unpaid-wages</u> (accessed 7 July 2020).
- BHRRC (2020c), COVID-19: Supply Chain Workers. London: BHRRC. https://www.business-humanrights.org/en/big-issues/covid-19-coronavirusoutbreak/supply-chain-workers/ (accessed 14 September 2020). Also see https://www.business-humanrights.org/en/big-issues/covid-19-coronavirusoutbreak/covid-19-food-and-beverage-supply-chain-workers/ (accessed 3 February 2021).
- Cambodianess (2020), Coming Home in COVID-19 Times: Migrant Workers Follow Cambodia's Directives. 11 April. <u>https://cambodianess.com/article/coming-home-in-covid-19-times-migrant-workers-follow-cambodias-directives</u> (accessed 25 June 2020).
- Center for Preventative Action (2020), 'Peace, Conflict, and COVID-19', *Council of Foreign Relations (CFR)*, 6 August. <u>https://www.cfr.org/article/peace-conflict-and-covid-19</u> (accessed 14 September 2020).
- Center for Strategic and International Studies (CSIS) (2020), Southeast Asia COVID-19 Tracker [Data Set]. Washington, DC: CSIS. <u>https://www.csis.org/programs/southeast-asia-program/southeast-asia-covid-19-</u> tracker-0 (accessed 23 August 2020).
- Center on International Cooperation (2020), *Perspectives on a Pandemic: COVID-19 Analysis from CIC.* November 2. New York, NY: New York University, Center on International Cooperation. https://cic.nyu.edu/publications/perspectivespandemic-covid-19-analysis-cic (accessed 18 November 2020).
- Chandra, A., I. Muhajid, and R. Mahyassari (2020), 'Trade Measure in the Time of COVID-19: The Case of ASEAN', *ASEAN Policy Brief*, No. 3. July. https://asean.org/storage/2020/07/ASEAN-Policy-Brief-3_FINAL_.pdf (accessed 4 September 2020).
- Chapman, P. (2020), Understanding Access to Justice During COVID-19, CIC-NYU, September. 6 October. New York, NY: New York University, Center on International Cooperation. <u>https://medium.com/sdg16plus/understanding-access-to-justice-in-the-covid-19-era-b69f15508bb1</u> (accessed 18 November 2020).
- Chew, A. (2020), 'Coronavirus: Indonesia Braces for Exodus of Millions of Workers', *The South China Morning Post*. 10 April. <u>https://www.scmp.com/week-asia/health-</u> <u>environment/article/3079295/coronavirus-indonesia-braces-exodus-millions-</u> <u>workers</u> (accessed 8 August 2020).

- Chalamwong, Y. (2011), 'Different Stream, Different Needs, and Impact: Managing International Labor Migration in ASEAN: Thailand (Emigration)', *Philippine Institute for Development Studies (PIDS) Discussion Papers*, No. 27. Makati: PIDS.
- Chan, J. (2020), 'COVID-19 Puts Southeast Asia's Food Supply to the Test', ASEAN Today. 4 August. <u>https://www.aseantoday.com/2020/04/covid-19-puts-southeast-asias-food-supply-to-the-test/</u> (accessed 10 October 2020).
- Chantanusornsiri, W. (2020), 'Rejected Handouts Back on Track', *Bangkok Post*. 15 April. <u>https://www.bangkokpost.com/business/1899980/rejected-handouts-back-on-track</u> (accessed 7 May 2020).
- Chua, Y. (2020), 'COVID-19 Pandemic in Southeast Asia: The Numbers, March–Present [Data Set]', *Reporting ASEAN*. https://www.reportingasean.net/covid-19-casessoutheast-asia/ (accessed 16 September 2020).
- Chugh, A. (2020), *Will COVID-19 Change How We Think about Migration and Migrant Workers?* May 22. Vienna: World Economic Forum. <u>https://www.weforum.org/agenda/2020/05/covid-19-coronavirus-migration-migrant-workers-immigration-policy-health-securitization-risk-travel-bubbles/</u> (accessed 14 June 2020).
- Clean Clothes Campaign (2020), Coronavirus and Garment Workers in Supply Chains, 1 April–24 August [Data Set]. Amsterdam: Clean Clothes Campaign. <u>https://cleanclothes.org/news/2020/april-2020-covid19-blog</u> (accessed 11 May 2020).
- 'Coronavirus Epidemic Seen Impacting Thai Economy' (2020), *The Nation*. 25 January. <u>https://www.nationthailand.com/national/30337149</u>) (accessed 7 May 2020).
- 'Coronavirus: Myanmar Braces for Surge in Cases after Migrant Worker Exodus from Thailand' (2020), *The South China Morning Post*. 30 March (republished from Kyodo and Reuters). <u>https://www.scmp.com/news/asia/southeastasia/article/3077630/coronavirus-myanmar-braces-surge-cases-after-migrant</u> (accessed 21 May 2020).
- CFR (2019), What Is ASEAN? Council on Foreign Relations. New York, NY: CFR. https://www.cfr.org/backgrounder/what-asean (accessed 14 September 2020).
- Dabla-Norris, E. et al. (2020), *Vietnam's Success in Containing COVID-19 Offers Roadmap* for Other Developing Countries. 29 June. <u>https://www.imf.org/en/News/Articles/2020/06/29/na062920-vietnams-success-</u> in-containing-covid19-offers-roadmap-for-other-developing-countries (accessed
 - in-containing-covid19-offers-roadmap-for-other-developing-countries (accessed 18 August 2020).
- Dang, D.N. (2020), 'How COVID-19 Complicates Vietnam's South China Sea Efforts', *The Diplomat*. 7 August. <u>https://thediplomat.com/2020/08/how-covid-19-</u> <u>complicates-vietnams-south-china-sea-efforts/</u> (accessed 10 September 2020).
- De, J. and S. Nadeem (2020), *Six Charts on Myanmar's Economy in the Time of COVID-19*. 7 July. Washington, DC: International Monetary Fund Asia and Pacific Department. <u>https://www.imf.org/en/News/Articles/2020/07/07/na70720-myanmars-</u> <u>economy-in-the-time-of-covid19-six-charts</u> (accessed 12 August 2020).
- Demirgüç-Kunt, A. et al. (2018), *The Global Findex Database 2017*. Measuring Financial Inclusion and the Fintech Revolution.

https://globalfindex.worldbank.org/sites/globalfindex/files/2018-

04/2017%20Findex%20full%20report_0.pdf (accessed 11 September 2020).

- Diao, X. and M. Wang (2020), *Significant Economic Impacts Due to COVID-19 and Declining Remittances in Myanmar*. Washington, DC: International Food Policy Research Institute Report. May.
- Djalante R. et al. (2020), *The ASEAN's Responses to COVID-19: A Policy Sciences Analysis*. DOI: 10.31234/osf.io/8347d. (accessed 4 September 2020).
- Ducanes, G. (2013), Labour Shortages, Foreign Migrant Recruitment and the Portability of Qualifications in East and South-East Asia. Bangkok: International Labour Organization (ILO) Regional Office for Asia and the Pacific.
- Dzulfikar, L.T. (2020), 'How the COVID-19 Pandemic Will Leave More Young People Unemployed in Indonesia', *The Jakarta Post*. 5 July. <u>https://www.thejakartapost.com/academia/2020/07/05/how-the-covid-19-pandemic-will-leave-more-young-people-unemployed-in-indonesia.html</u> (accessed 4 August 2020).
- ECPAT International (2016), *End Child Prostitution and Trafficking International.* Bangkok: ECPAT International. <u>https://www.ecpat.org</u> (accessed 11 August 2020).
- Ellis, J. (2020), 'COVID-19 Will Compound ASEAN's Long-Term Food Security Challenges New Report', AgFunderNews. 14 April. <u>https://agfundernews.com/covid-19-willcompound-aseans-long-term-food-security-challenges-new-report.html</u> (accessed 1 July 2020).
- 'Employing Caution' (2020), The Bangkok Post. 30 March.
 <u>www.bangkokpost.com/business/1889140/employing-caution</u> (accessed 21 July 2020).
- Engblom, A., N. Lephilibert, and N. Baruah (2020), COVID-19 Impact on Migrant Workers and Country Response in Thailand. 3 July. Bankok: ILO. https://www.ilo.org/wcmsp5/groups/public/---asia/---ro-bangkok/---srobangkok/documents/briefingnote/wcms_741920.pdf (accessed 20 August 2020).
- Economic Research Institute for ASEAN and East Asia (ERIA) (2015), 'ASEAN beyond 2015: The Imperatives for Further Institutional Changes', *ERIA Discussion Paper Series*. Jakarta: ERIA. <u>https://www.eria.org/ERIA-DP-2014-01.pdf</u> (accessed 4 September 2020).
- Evetts, J. (n.d.), *SDGs.* SciDev.Net Asia and Pacific. <u>https://www.scidev.net/asia-pacific/governance/sdgs/</u> (accessed 2 November 2020).
- 'Exodus of 1,500 Migrant Workers Daily Sparks Worry of Thai Labour Shortage' (2020), *The Nation Thailand*. 9 June. https://www.nationthailand.com/news/30389324?utm_source=category&utm_m

<u>edium=internal_referral</u> (accessed 29 June 2020).

- Food and Agriculture Organization of the United Nations (FAO) (2019), *State of Food Security and Nutrition in the World 2019.* Rome: FAO. <u>http://www.fao.org/3/ca5162en/ca5162en.pdf</u> (accessed 14 August 2020).
- FAO (2020a), COVID-19's Long Shadow Darkens the Future of Southeast Asia's Food Security. 6 May. Rome: FAO. <u>http://www.fao.org/asiapacific/news/detail-</u> <u>events/zh/c/1273822/</u> (accessed 22 May 2020).
- FAO (2020b), Southeast Asian Nations Examine the State of Food Systems to Ensure Recovery and Resilience in a post-COVID-19 Era. 24 June. Rome: FAO.

http://www.fao.org/asiapacific/news/detail-events/en/c/1295300/ (accessed 7 July 2020).

- Fernandez, H.A. (2020), 'ADB: Asia Stands to Lose 68 Million Jobs if Coronavirus Is not Contained in Six Months', *Eco Business*. 7 April. https://www.ecobusiness.com/news/adb-asia-stands-to-lose-68-million-jobs-if-coronavirus-is-notcontained-in-six-months/ (accessed 4 June 2020).
- Fleury, A. (2016), 'Understanding Women and Migration: A Literature Review', World Bank-Knomad Working Paper, No. 8. Washington, DC: World Bank, Knomad. https://www.knomad.org/publication/understanding-women-and-migrationliterature-review-annex-annotated-bibliography (accessed 7 May 2020).
- Food Industry Asia (2020), Food Supply Chain Disruptions Caused by COVID-19 May Lead to Increased Food Security Challenges in ASEAN. 9 April. Singapore: Food Industry Asia. https://foodindustry.asia/news/news-release/food-supply-chain-disruptionscaused-by-covid-19-may-lead-to-increased-food-security-challenges-in-asean (accessed 7 July 2020).
- Foresti, M. and J. Hagen-Zanker (2018), *Migration and the 2030 Agenda for Sustainable Development*. September. Publication Series. London: ODI.
- French-Myanmar Chamber of Commerce and Industry (2020), Overcoming as One: COVID-19 Economic Relief Plan-CERP. 27 April. Yangon: French-Myanmar Chamber of Commerce and Industry. <u>https://www.ccifrance-</u> <u>myanmar.org/en/newsandresources/overcoming-as-one-covid-19-economic-</u> <u>relief-plan-cerp</u> (accessed 12 June 2020).
- Giammarinaro, M.G. (2020), *The Impact and Consequences of COVID-19 Pandemic on Trafficked and Exploited Persons.* United Nations Human Rights Special Procedures. 8 June. Geneva: Office of the High Commissioner for Human Rights. https://www.ohchr.org/Documents/Issues/Trafficking/COVID-19-Impacttrafficking.pdf (accessed 4 August 2020).
- Giammarinaro, M.G. and L. Palumbo (2020), 'COVID-19 and Inequalities: Protecting the Human Rights of Migrants in a Time of Pandemic', *Migration Policy Practice*, 10(2), April–June.
- Global Forum on Migration and Development (GFMD) (2019), *Recommendations of the High-Level Political Forum on Protecting the Rights of Migrant Workers*. Geneva: GFMD. <u>https://gfmd.org/pfp/ppd/11056</u> (accessed 7 June 2020).
- GFMD (2020), COVID-19 Information Hub (refer to several postings since January 2020). Geneva: GFMD.
- Guan, W.J. et al. (2020), 'Comorbidity and its Impact on 1590 Patients with COVID-19 in China: A Nationwide Analysis', *The European Respiratory Journal*, 55(5), 2000547. https://doi.org/10.1183/13993003.00547-2020.
- Gu, V. (2020), 'Virus or Hunger? Asia's Migrant Workers' COVID-19 Dilemma. CGTN. 5 April. <u>https://news.cgtn.com/news/2020-04-05/Virus-or-hunger-Asia-s-migrant-workers-COVID-19-dilemma-Pq0vBXgZoI/index.html</u> (accessed 12 August 2020).
- Guadagno, L. (2020), 'Migrants and the COVID-19 Pandemic: An Initial Analysis', *Migration Research Series*, No. 60. Geneva: IOM.
- Harkins, B. and D. Lindgren (2018), *Labour Migration in the ASEAN Region: Assessing the Social and Economic Outcomes for Migrant Workers.*

http://www.migratingoutofpoverty.org/files/file.php?name=harkins-labourmigration-in-asean-update.pdf&site=354 (accessed 29 May 2020).

- Harkins. B., D. Lindgren, and T. Suravoranon (2017), *Risks and Rewards: Outcomes of Labour Migration in Southeast Asia*. 15 December. ILO, IOM, and
 Rapid Asia. ISBN 978-92-2-131411-0
- Hartigan, R. (2020), 'A Look inside Thailand, Which Prevented Coronavirus from Gaining a Foothold', National Geographic. 18 June.
 <u>https://www.nationalgeographic.com/history/2020/06/look-inside-thailand-prevented-coronavirus-gaining-foothold/</u> (accessed 11 July 2020).
- Hayat, R. (2020), 'How COVID-19 Will Impact ASEAN: Deep Recessions and a Weak Recovery', Rabobank. 19 May. <u>https://economics.rabobank.com/publications/2020/may/impact-covid-19-asean-</u>

recessions-and-weak-recovery/ (accessed 22 July 2020). Health Information System Development in Thailand (2015), *Thai Health 2015: Health Frauds: When Health is Commoditized, Drugs Become Profit-Making Tools.* https://www.hiso.or.th/hiso/picture/reportHealth/ThaiHealth2015/eng2015 4.pd

- <u>f</u> (accessed 29 August 2020).
- Hein, A., P. Minoletti, A. Paing, N. Win, and B. Harkins (2015), *Safe Migration Knowledge, Attitudes and Practices in Myanmar.* Bangkok: ILO.
- Hogan, A.B. et al. (2020), The Potential Impact of the COVID-19 Epidemic on HIV, TB and Malaria in Low- and Middle-Income Countries. 1 May. London: Imperial College London. <u>https://doi.org/10.25561/78670</u>.
- Hopkins, E., F. Bastagli, and J. Hagen-Zanker (2016), 'Internal Migrants and Social Protection: A Review of Eligibility and Take-up', *Overseas Development Institute Working Paper*, No. 436. London: ODL
- Htet, K.S. (2020), 'COVID-19 Cases in Myanmar Ich up to 121', *Myanmar Times*. 21 April. <u>https://www.mmtimes.com/news/covid-19-cases-myanmar-inch-121.html</u> (accessed 21 August 2020).
- Human Rights Watch (HRW) (2018), Hidden Chains: Rights Abuses and Forced Labour in Thailand's Fishing Industry. 23 January. New York, NY: HRW.
 https://www.hrw.org/report/2018/01/23/hidden-chains/rights-abuses-andforced-labor-thailands-fishing-industry (accessed 21 September 2020).
- HRW (2020a), COVID-19 Spreading Anti-Asian Racism and Xenophobia Worldwide. 8
 May. New York, NY: HRW. https://www.hrw.org/news/2020/05/12/covid-19fueling-anti-asian-racism-and-xenophobia-worldwide (accessed 12 July 2020).
- Human Rights Watch (2020b), *Philippines: Curfew Violators Abused*. 26 March. <u>https://www.hrw.org/news/2020/03/26/philippines-curfew-violators-abused</u> (accessed 11 September 2020).
- Hutt, D. (2020), 'Mass Unemployment the New Normal in SE Asia', *Asia Times*. 25 May. https://asiatimes.com/2020/05/mass-unemployment-the-new-normal-in-se-asia/ (accessed 1 August 2020).
- Idris, I. (2020), Impact of COVID-19 on Child Labour in South Asia. K4D Helpdesk Report 819. Brighton: Institute of Development Studies. <u>https://resourcecentre.savethechildren.net/library/impact-covid-19-child-labour-south-asia</u> (accessed 22 July 2020).
Imperial College London (2020), Adapting Hospital Capacity to Meet Changing Demands During the COVID-19 Pandemic, Report No. 27. 15 June. London: Imperial College London. https://www.imperial.ac.uk/mrc-global-infectious-disease-analysis/covid-19/report-27-hospital-capacity/ (as of February 2021, Imperial College had published 42 reports on different aspects of public health management and the coronavirus disease).

- International Labour Organization (ILO) (n.d.), COVID-19 and the World of Work: Regional and Country Impact and Policies. Geneva: ILO. <u>https://www.ilo.org/global/topics/coronavirus/regional-country/lang--</u> <u>en/index.htm</u> (accessed 27 August 2020).
- ILO (n.d.), Promoting and Building Social Protection in Asia (3rd Phase): Extending Social Security Coverage in ASEAN. Bangkok: ILO in Asia and the Pacific: Programmes and Projects. <u>https://www.ilo.org/asia/projects/WCMS_507729/lang--ja/index.htm</u> (accessed 21 July 2020).
- ILO (1999a), ILC87 Report III (1B) Migrant Workers. June. Geneva: ILO. https://www.ilo.org/public/english/standards/relm/ilc/ilc87/r3-1b2.htm (accessed 21 September 2020).
- ILO (2015a), Analytical Report on International Labour Migration Statistics Database in ASEAN: Improving Data Collection for Evidence-Based Policy-Making. Bangkok: ILO Regional Office for Asia and the Pacific.
- ILO (2015b), *Review of the Effectiveness of the MOUs in Managing Labour Migration between Thailand and Neighbouring Countries*. Bangkok: ILO Regional Office for Asia and the Pacific.
- ILO (2016), Building an Inclusive Future with Decent Work: Towards Sustainable Development in Asia and the Pacific. Report of the Director General to the 16th Regional Meeting of Asia and Pacific, held in Indonesia. Geneva: ILO. <u>https://www.ilo.org/wcmsp5/groups/public/---ed_norm/---</u> <u>relconf/documents/meetingdocument/wcms_531409.pdf</u> (accessed 8 June 2020).
- ILO (2018a), Digitalization to Promote Decent Work in ASEAN for Migrant Workers. Thematic background paper for the 11th ASEAN Forum on Migrant Labour held in Singapore in October 2018. Geneva: ILO.
- ILO (2018b), More than 68 Percent of the Employed Population in Asia-Pacific Are in the Informal Economy. 2 May. Geneva: ILO. <u>https://www.ilo.org/asia/media-</u> <u>centre/news/WCMS_627585/lang--en/index.htm</u> (accessed 21 August 2020).
- ILO (2020a), COVID-19 and the World of Work: Impact and Policy Responses. ILO Monitor 1st Edition. 18 March. Geneva: ILO. https://www.ilo.org/wcmsp5/groups/public/---dgreports/---dcomm/documents/briefingnote/wcms_738753.pdf (accessed 25 July 2020).
- ILO (2020b), COVID-19 Causes Devastating Losses in Working Hours and Employment. 7 April. Geneva: ILO. https://www.ilo.org/global/about-theilo/newsroom/news/WCMS_740893/lang--en/index.htm (accessed 21 July 2020).
- ILO (2020c), Sectoral Brief: COVID-19 and the Impact on Agriculture and Food Security. 17 April. Geneva: ILO. <u>https://www.ilo.org/wcmsp5/groups/public/---</u> ed_dialogue/---sector/documents/briefingnote/wcms_742023.pdf</u> (accessed 22 July 2020).

- ILO (2020d), The COVID-19 Crisis: A Wake-up Call to Strengthen Social Protection Systems. 24 April. Geneva: ILO. <u>https://www.ilo.org/global/about-the-</u> <u>ilo/newsroom/news/WCMS_742676/lang--en/index.htm</u> (accessed 21 August 2020).
- ILO (2020e), Thematic Brief: COVID-19 and Employment in the Tourism Sector: Impact and Response in Asia and the Pacific. 24 April. Geneva: ILO. <u>https://www.ilo.org/wcmsp5/groups/public/---asia/---ro-</u> <u>bangkok/documents/briefingnote/wcms_742664.pdf</u> (accessed 20 July 2020).
- ILO (2020f), COVID-19 Crisis and the Informal Economy: Immediate Responses and Policy Challenges. Briefing Note. 5 May. Geneva: ILO. https://www.ilo.org/wcmsp5/groups/public/---ed_protect/---protrav/--travail/documents/briefingnote/wcms_743623.pdf (accessed 11 August 2020).
- ILO (2020g), *COVID-19: Impact on Migrant Workers and Country Response in Malaysia*. 8 May. Bangkok: ILO. https://www.ilo.org/wcmsp5/groups/public/---asia/---robangkok/documents/briefingnote/wcms_741512.pdf (accessed 21 July 2020).
- ILO (2020h), More than One in Six Young People Likely to Be out of Work Due to COVID-19. 27 May. Geneva: ILO. https://www.ilo.org/global/about-theilo/newsroom/news/WCMS_745879/lang--en/index.htm (accessed 23 August 2020).
- ILO (2020i), ILO Brief: Experiences of ASEAN Migrant Workers during COVID-19: Rights at Work, Migration and Quarantine during the Pandemic, and Re-Migration Plans. 3 June. Geneva: ILO.<u>https://www.ilo.org/wcmsp5/groups/public/---asia/---robangkok/documents/briefingnote/wcms_746881.pdf</u> (accessed 21 August 2020).
- ILO (2020j), Briefing Note: Covid-19 and Women Migrant Workers in ASEAN. 4 June. Geneva: ILO. https://www.ilo.org/wcmsp5/groups/public/---asia/---robangkok/documents/briefingnote/wcms_746979.pdf (accessed 6 July 2020).
- ILO (2020k), Impact of the COVID-19 Crisis on Loss of Jobs and Hours amongst Domestic Workers. 15 June. Geneva: ILO. https://www.ilo.org/global/topics/domesticworkers/publications/factsheets/WCMS_747961/lang--en/index.htm (accessed 20 July 2020).
- ILO (2020l), COVID-19: Impact on Migrant Workers and Country Response in Myanmar. 22 June. Geneva: ILO. <u>https://www.ilo.org/yangon/publications/WCMS_749466/lang--en/index.htm</u> (accessed 26 July 2020).
- ILO (2020m), *ILO Warns of COVID-19 Migrant 'Crisis within a Crisis*'. 24 June. Geneva: ILO. <u>https://www.ilo.org/global/about-the-</u> <u>ilo/newsroom/news/WCMS_748992/lang--en/index.htm</u> (accessed 26 July 2020).
- ILO (2020n), A Gender-Responsive Employment Recovery: Building Back Fairer. July. Geneva: ILO. <u>https://www.ilo.org/wcmsp5/groups/public/---</u> ed emp/documents/publication/wcms 751785.pdf (accessed 4 August 2020).
- ILO (2020o), COVID-19: Impact on Migrant Workers and Country Response in Thailand. 3 July. Bangkok: ILO Country Office for Thailand, Cambodia and Lao PDR. https://www.ilo.org/asia/publications/issue-briefs/WCMS_741920/lang-en/index.htm (accessed 4 August 2020).

- ILO (2020p), Regional Events Pave the Way for Global Discussions on COVID-19 and the World of Work. 3 July. Geneva: ILO. <u>https://www.ilo.org/global/about-the-</u> <u>ilo/newsroom/news/WCMS_749885/lang--en/index.htm</u> (accessed 6 August 2020).
- ILO (2020q), 81 Million Jobs Lost as COVID Creates Turmoil in Asia-Pacific Labour Markets. 15 December 2020. Geneva: ILO. <u>https://www.ilo.org/asia/mediacentre/news/WCMS_763819/lang--en/index.htm</u> (accessed 19 January 2021).
- ILO and ADB (2014), ASEAN Community 2015: Managing Integration for Better Jobs and Shared Prosperity, pp.83–100. Bangkok; Manila: ILO and ADB.
- ILO News (2020), *ILO: COVID-19 Causes Devastating Losses in Working Hours and Employment*. April 29. <u>https://www.ilo.org/global/about-the-</u>

ilo/newsroom/news/WCMS_743036/lang--en/index.htm. (accessed 17 May 2020).

- ILOStat (n.d.), *Country Profiles*. [Data Set]. <u>https://ilostat.ilo.org/data/country-profiles/</u> (accessed 31 May 2020).
- International Monetary Fund (IMF) (2020), *World Economic Outlook, April 2020: The Great Lockdown*. Washington, DC: IMF.

www.imf.org/en/Publications/WEO/Issues/2020/04/14/weo-april-2020 (accessed 5 August 2020).

IMF (2020), World Economic Outlook, October 2020: A Long and Difficult Ascent. Washington, DC: IMF.

https://www.imf.org/en/Publications/WEO/Issues/2020/09/30/world-economicoutlook-october-2020 (accessed 19 November 2020).

- Indonesian Migrant Workers Union (n.d.), Timeline [Facebook Page]. https://www.facebook.com/IMWUHK/ (accessed 25 September 2020).
- International Organization for Migration (IOM) (2018), *Migration and the 2030 Agenda: Comprehensive SDG Target and Migration Correlation.* Geneva: IOM. <u>http://www.migration4development.org/sites/default/files/en_booklet_web.pdf</u> (accessed 22 July 2020).
- IOM (2019), Migration Governance Indicators: A Global Perspective. Geneva: IOM. https://publications.iom.int/books/migration-governance-indicators-globalperspective (accessed 5 August 2020).
- IOM (2020a), COVID-19 Analytical Snapshot #16: International Remittances. 17 April. Geneva: IOM. <u>https://www.iom.int/sites/default/files/documents/covid-</u>
 <u>19 analytical snapshot 16 - international remittances.pdf</u> (accessed 15 July 2020).
- IOM (2020b), *Missing Migrants Project*. 11 September. Geneva: IOM. <u>https://missingmigrants.iom.int/region/asia</u> (accessed 2 October 2020).
- Institute of Agricultural Science for Southern Viet Nam (2020), *COVID-19's Long Shadow Darkens the Future of Southeast Asia.* 22 May. Ho Chi Minh City: Institute of Agricultural Science for Southern Viet Nam.
- Inter-agency Coordination Group against Trafficking in Persons (2018), *Issue Brief #5: The Role of the Sustainable Development Goals (SDGs) in Combating Trafficking in persons.* April. Vienna: Inter-agency Coordination Group against Trafficking in Persons.

http://icat.network/sites/default/files/publications/documents/ICAT_Issue_Brief_

SDGs%202018.pdf (accessed 23 July 2020).

- International Centre for Missing and Exploited Children (2020), *Child Protection during the COVID-19 Crisis.* Alexandria, VA: International Centre for Missing and Exploited Children. <u>https://www.icmec.org/child-protection-during-the-covid-19-crisis/</u> (accessed 5 August 2020).
- International Labour Rights Forum (2020), *COVID-19 Impact on Migrant Workers in Thailand.* 27 March. Washington, DC: International Labour Rights Forum. https://laborrights.org/blog/202003/covid-19-impact-migrant-workers-thailand (accessed 25 July 2020).
- International Planned Parenthood Federation (2020), *COVID-19 Response: Situation in East Asia, South East Asia and the Pacific.* April. London: International Planned Parenthood Federation. <u>https://www.ippfeseaor.org/sites/ippfeseaor/files/2020-</u> 04/RDs%20Update%20-%20Final.pdf (accessed 15 August 2020).
- Jalli, N. (2020), 'Lack of Internet Access in Southeast Asia Poses Challenges for Students to Study Online amid COVID-19 Pandemic', *The Conversation*. 17 March. <u>https://theconversation.com/lack-of-internet-access-in-southeast-asia-poses-</u> <u>challenges-for-students-to-study-online-amid-covid-19-pandemic-133787</u> (accessed 21 July 2020).
- Jefriando, M. and G. Suroyo (2020), 'Indonesia Business Chamber Says 6.4 mln Jobs Lost so Far in Pandemic', *Reuters*. 19 June. <u>https://in.reuters.com/article/indonesiaeconomy-unemployment/indonesia-business-chamber-says-6-4-mln-jobs-lost-sofar-in-pandemic-idUSL4N2DW1P1</u> (accessed 5 September 2020).
- Joining Forces and Child Rights Coalition Asia (2020), ASEAN and SAARC Must Put Children's Rights at the Heart of COVID-19 Response. 29 April. https://reliefweb.int/report/world/asean-and-saarc-must-put-children-s-rightsheart-covid-19-response (accessed 21 August 2020).
- Kaicome, J. (2020), 'Out of Work: Options for Myanmar Migrant Workers in Thailand as They Scrape by during COVID-19', *The Irrawaddy*. 22 July (republished in *Global Voices*). <u>https://globalvoices.org/2020/07/22/out-of-work-out-of-options-</u> <u>myanmar-migrant-workers-in-thailand-scrape-by-during-covid-19/</u> (accessed 4 September 2020).
- Karmini, N. (2020), 'After Denial, Indonesia Sees Mounting Coronavirus Deaths', *The Diplomat*. 21 April. <u>https://thediplomat.com/2020/04/after-denial-indonesia-sees-</u> <u>mounting-coronavirus-deaths/</u> (accessed 28 July 2020).
- Kassim, A. and H.M.R. Zin (2011), 'Policy on Irregular Migrants in Malaysia: An Analysis of its Implementation and Effectiveness', *PIDS Discussion Paper Series*, No. 34. Makati: PIDS.
- Kaur-Gill, S. (2020), The COVID-19 Pandemic and Outbreak Inequality: Mainstream Reporting of Singapore's Migrant Workers in the Margins. 30 September. Chua Thian Poh Singapore: National University of Singapore, Community Leadership Center. https://doi.org/10.3389/fcomm.2020.00065
- Kimura F., et al. (2020), 'Pandemic (COVID-19) Policy, Regional Cooperation and the Emerging Global Production Network', Asian Economic Journal. 9 June. <u>https://doi.org/10.1111/asej.12198</u>.

- Kirschner, N. (2020), 'U.S. Private Sector Helps ASEAN Fight COVID-19', Share America. 22 July. <u>https://share.america.gov/u-s-private-sector-helps-asean-fight-covid-19/</u> (accessed 5 September 2020).
- Kluge, H.H., Z. Jakab, J. Bartovic, V. D'Anna, and S. Severoni (2020), 'Refugee and Migrant Health in the COVID-19 Response', *The Lancet*. 31 March. <u>https://doi.org/10.1016/S0140-6736(20)30791-1</u>
- Knight, R.T. (2020), 'Migrant Workers on Virus Front Line as Asia Returns to Work', BloombergQuint. 15 May. <u>https://www.bloombergquint.com/onweb/migrant-workers-on-virus-front-line-as-asia-returns-to-work</u> (accessed 2 August 2020).
- Koh, D. (2020), 'Migrant Workers and COVID-19', *Occupational and Environmental Medicine*, 77(9), pp.634–36. 8 June. DOI: <u>10.1136/oemed-2020-106626</u>
- Kovac, M. (2020), 'COVID-19: Keeping Food Supply Chains Open Will Help Us All Survive This Stress Test', *The Jakarta Post*. 14 April. <u>https://www.thejakartapost.com/academia/2020/04/14/covid-19-keeping-food-supply-chains-open-will-help-us-all-survive-this-stress-test.html</u> (accessed 25 July
- 2020). Kuentak, P. and I.L. Tisnadibrata (2020), 'COVID Report Card', *Bangkok Post.* 4 May. <u>https://www.bangkokpost.com/world/1912228/covid-report-card</u> (accessed 3 August 2020).
- Lai, T. (2020), 'New Rules for Employees' Minimum Housing Standards from September 1, 2020. Employers to Comply or Be Fined RM 50,000', *HRnews.my*. 1 September. <u>http://hrnews.my/2020/09/01/new-rules-for-employees-minimum-housing-</u> <u>standards-from-sept-1-employers-to-comply-or-be-fined-rm50000/</u> (accessed 2 October 2020).
- Lee, E. (2015), 'ASEAN: Growth in the Fast Lane', *Asian Management Insights.* <u>https://ink.library.smu.edu.sg/ami/29</u> (accessed 27 August 2020).
- Letzing, J. (2020), *How COVID-19 Is Throttling Vital Migration Flows*. April 8. Cologny: World Economic Forum. <u>https://www.weforum.org/agenda/2020/04/covid-19-is-throttling-vital-migration-flows/</u> (accessed 28 April 2020).
- Li, Y-T. and K. Whitworth (2016), 'When the State Becomes Part of the Exploitation: Migrants' Agency within the Institutional Constraints in Australia'. *International Migration*, 54(6), pp.138–50. https://doi.org/10.1111/imig.12267
- Lindsey, T. and T. Mann (2020), 'Indonesia Was in Denial over Coronavirus. Now It May Be Facing Looming Disaster', *The Jakarta Post.* 9 April. <u>https://www.thejakartapost.com/academia/2020/04/09/indonesia-was-in-denialover-coronavirus-now-it-may-be-facing-a-looming-disaster.html</u> (accessed 21 July 2020).
- 'Locked Up in Malaysia's Lockdown' (2020), *Al Jazeera*. 3 July. <u>https://www.aljazeera.com/program/episode/2020/7/3/locked-up-in-malaysias-lockdown/</u>
- Long, K., E.M. Vidal, A. Kuch, and J. Hagen-Zanker (2017), *Citizenship, Migration and the 2030 Agenda for Sustainable Development.* Briefing. London: ODI.
- Lopez, M. (2020), 'Remittances Sustain Growth to \$3B in July Despite Pandemic', CNN Philippines. 15 September.

https://cnnphilippines.com/business/2020/9/15/Filipino-remittances-July-

pandemic.html?fbclid=IwAR0ipZKoBiwOz2feGDMubocBJmRAAy3gpn9KOAAilByPIj 74lQihS7JHQzg (accessed 25 September 2020).

- Maizland, L. and Albert, E. (2020), *What Is ASEAN?*. Council on Foreign Relations. 24 November. <u>https://www.cfr.org/backgrounder/what-asean</u>. (accessed)
- Manlangit, H. (2020), 'The Hurting Heroes: The COVID-19 Crisis and Overseas Filipino Workers', *The Diplomat*. 18 May. <u>https://thediplomat.com/2020/05/the-hurting-heroes-the-covid-19-crisis-and-overseas-filipino-workers/</u> (accessed 22 August 2020).
- Medina, A.F. (2020) *Indonesia Employment Issues During COVID-19.* ASEAN Briefing. 9 April. <u>https://www.aseanbriefing.com/news/indonesia-employment-issues-covid-</u> 19/ (accessed 3 July 2020).
- Menon, J. (2020), Assessing the Economic Impacts of COVID-19 on ASEAN Countries. 30 March. Singapore: ISEAS – Yusof Ishak Institute (republished in Thailand Business News). <u>https://www.thailand-business-news.com/asean/78517-assessing-the-</u> economic-impacts-of-covid-19-on-asean-countries.html (accessed 24 July 2020).
- Mendoza, D.R. and G. Sugiyarto (2017), *The Long Road Ahead: Status Report on the Implementation of the ASEAN MRAs on Professional Services*. Mandaluyong City: ADB.
- Migrant Forum in Asia (MFA) (n.d.), *Indonesian Female Domestic Workers Vulnerable to Extreme Exploitation and Abuse.* Quezon City: MFA. <u>https://mfasia.org/female-domestic-workers-vulnerable-to-extreme-exploitation-and-abuse/</u> (accessed 28 August 2020).
- MFA (2020a), An Appeal to Governments on Behalf of Migrant Workers in the midst of the COVID-19 Pandemic. 29 April. Quezon City: MFA. https://mfasia.org/anappeal-to-governments-on-behalf-of-migrant-workers-in-the-midst-of-the-covid-19-pandemic/ (accessed 29 July 2020).
- MFA (2020b), COVID-19: A Time for Businesses to Act Responsibly in Ensuring Justice for Migrant Workers. 1 June. Quezon City: MFA. https://mfasia.org/covid-19-a-timefor-businesses-to-act-responsibly-in-ensuring-justice-for-migrant-workers/ (accessed 2 August 2020).
- MFA (2020c), Appeal No. 2: Urgent Justice Mechanism for Repatriated Migrant Workers Now. 9 July. Quezon City: MFA. <u>https://mfasia.org/appeal-no-2-urgent-justice-</u> <u>mechanism-for-repatriated-migrant-workers-now/</u> (accessed 23 August 2020).
- 'Migrant Workers on Virus Front Line as Southeast Asia Returns to Work' (2020), Japan Times. 15 May. <u>https://www.japantimes.co.jp/news/2020/05/15/asia-</u> pacific/migrant-workers-coronavirus/ (accessed 27 July 2020).
- 'Migrant Workers Poised to Return to Thailand'. (2020), UCA News. 21 July. <u>https://www.ucanews.com/news/migrant-workers-poised-to-return-to-thailand/88847</u> (accessed 2 August 2020).
- Migrant Working Group (2020), A Brief Overview of the Situation, Laws and Policies Concerning Migrant Worker Management during the COVID-19 Outbreak. 22 June. https://mwgthailand.org/en/news/1594880852 (accessed 29 August 2020).
- Migration Data Portal (2020), *Migration Policies and Governance*. 18 March. Geneva: IOM. <u>https://migrationdataportal.org/themes/migration-policies-and-governance#:~:text=These%20six%20policy%20dimensions%20are,and%20safe%2</u>

<u>Oand%20orderly%20migration</u> (accessed 25 July 2020).

- Millard, R. (2020), 'Millions of Migrant Workers Head Home Due to Virus: UN', *The Independent News and Media Singapore Edition*. 25 June. <u>http://theindependent.sg/millions-of-migrant-workers-head-home-due-to-virus</u> (accessed 11 August 2020).
- Ministry of Labour, Employment and Social Security and ILO (2016), *Myanmar Labour Force, Child Labour and School to Work Transition Survey 2015*. Naypyidaw; Geneva: Ministry of Labour, Employment and Social Security and ILO. https://www.ilo.org/wcmsp5/groups/public/---asia/---ro-bangkok/---iloyangon/documents/publication/wcms_516117.pdf (accessed 5 October 2020).

Minter, A. (2020), 'Coronavirus Is Bringing Out the Worst in Malaysians', BloombergQuint. 15 May. <u>https://www.bloombergquint.com/business/coronavirus-is-bringing-out-the-</u>

worst-in-malaysians (accessed 2 July 2020).

- Moore, R.N. (2017), Protected or Put in Harm's Way? Bans, Barriers and Restrictions on Women's Labour Migration in ASEAN. Bangkok: ILO and United Nations Entity for Gender Equality and the Empowerment of Women (UN Women).
- Moritsugu, K. (2020), 'China Considers Bill Limiting Hong Kong Opposition Activity', *The Diplomat*. 22 May. <u>https://thediplomat.com/2020/05/china-considers-bill-limiting-hong-kong-opposition-activity/</u> (accessed 23 August 2020).
- Moroz, H., M. Shrestha, and M. Testaverde (2020), 'Potential Responses to the COVID-19 Outbreak in Support of Migrant Workers', *World Bank Working Paper*. Washington, DC: World Bank.

https://openknowledge.worldbank.org/handle/10986/33625 (accessed 12 August 2020).

- Mortensen, C. et al. (2016), *The Education Situation of Children of Migrant Families in Selected ASEAN Countries.* September. Phnom Penh: Aide et Action International Southeast Asia and China. <u>https://seac.aide-et-action.org/wp-</u> <u>content/uploads/2018/09/AEA-Migration-Desk-Review-Sept-2016-FINAL.pdf</u> (accessed 25 September 2020).
- 'MTUC Calls for Viable and Sustainable Plans to Help Workers' (2020), *Borneo Post Online*. 3 June. https://www.theborneopost.com/2020/06/03/mtuc-calls-forviable-and-sustainable-plans-to-help-workers/ (accessed 11 July 2020).
- Muchhala, B. (2013), 'Amplifying the Private Sector in Development: Concerns, Questions and Risks', *Third World Network Briefing Paper*, No. 69. October. Penang: Third World Network.
- Murakami, E., S. Shimizutani, and E. Yamada (2020), 'Projection of the Effects of the COVID-19 Pandemic on the Welfare of Remittance-Dependent Households in the Philippines'. In *Economics of Disaster and Climate Change*. 25 September.
- Murrugarra, E., J. Larrison, and M. Sasin (2011), *Migration and Poverty: Toward Better Opportunities for the Poor*. Directions in Development; Poverty. Washington, DC: World Bank.

https://openknowledge.worldbank.org/handle/10986/ (accessed 28 September 2020).

- Myat Thura (2020), *Myanmar: Over 60,000 Workers Lost Jobs Due to Factory Shutdowns Caused by COVID-19.* 28 April. London: Business and Human Rights Resource Centre (republished from Myanmar Times). https://www.businesshumanrights.org/en/myanmar-over-60000-workers-lost-jobs-due-to-factoryshutdowns-caused-by-covid-19 (accessed 25 September 2020).
- Myo Pa Pa San (2020), 'Myanmar's Garment Sector Facing Implosion as Orders Slump with COVID-19', *The Irrawaddy*. 15 July. <u>https://www.irrawaddy.com/news/burma/myanmars-garment-sector-facing-</u> <u>implosion-orders-slump-covid-19.html</u> (accessed 21 August 2020).
- Nanthini, S. (2020), 'In Limbo in ASEAN: Pandemic and Irregular Migration', *Relief Web*.
 25 June. <u>https://reliefweb.int/report/world/limbo-asean-pandemic-and-irregular-migration</u> (accessed 20 July 2020).
- National Geographic (2020), 'Look inside Thailand Prevented Coronavirus Gaining Foothold', July. <u>https://www.nationalgeographic.com/history/2020/06/look-</u> inside-thailand-prevented-coronavirus-gaining-foothold/ (accessed 13 August 2020).
- Nezurugo, M.P. and S. Hassan (2020), On World Refugee Day, a Look at How COVID-19 Is Affecting Refugees and Asylum Seekers. 19 June. Cologny: World Economic Forum.
- Ng, C. (2020), 'Singapore's COVID-19 Crisis: Forgetting to Check the Blindspot', *Timmerman Report*. 20 April. <u>www.timmermanreport.com/2020/04/singapores-</u> <u>covid19-crisis-forgetting-to-check-the-blind-spot/</u> (accessed 11 August 2020).
- Noel, N. (2020), Minorities, Migrants, and Social Exclusion During COVID-19. 12 May. New York, NY: Center on International Cooperation. <u>https://cic.nyu.edu/publications/covid-19-minorities-and-social-exclusion</u> (accessed 20 July 2020).
- Nortadujjin, A. (2020a), 'Is Child Labour on the Rise amid the Pandemic?', *The ASEAN Post*. 15 June. <u>https://theaseanpost.com/article/child-labour-rise-amid-pandemic</u> (accessed 15 September 2020).
- Nortajuddin, A. (2020b), 'COVID-19: Pushing Millions into Poverty?', *The ASEAN Post*. 23 June. <u>https://theaseanpost.com/article/covid-19-pushing-millions-poverty</u> (accessed 15 September 2020).
- Nortajuddin, A. (2020c), 'Cambodia's Economic Challenges amid the Pandemic', *The ASEAN Post*. 10 July. <u>https://theaseanpost.com/article/cambodias-economic-</u> <u>challenges-amid-pandemic</u> (accessed 15 September 2020).
- Nyein, N. (2020), 'Thai Border Closures Over COVID-19 Spark Myanmar Migrants Return, Health Concerns', *The Irrawaddy*. 23 March. <u>https://www.irrawaddy.com/news/burma/thai-border-closures-covid-19-spark-</u> myanmar-migrants-return-health-concerns.html (accessed 12 July 2020).
- Organisation for Economic Co-operation and Development (OECD) (2020a), Enterprise Policy Responses to COVID-19 in ASEAN: Measures to Boost MSME Resilience. Paris: OECD. <u>https://asean.org/storage/2020/06/Policy-Insight-Enterprise-Policy-Responses-to-COVID-19-in-ASEAN-June-2020v2.pdf</u>. This asessment was prepared in collaration with ASEAN Secretariat (accessed 8 September 2020).
- OECD (2020b), COVID-19 Crisis Response in ASEAN Member States. 4 May. Paris: OECD. http://www.oecd.org/coronavirus/policy-responses/covid-19-crisis-response-in-

asean-member-states-02f828a2/ (accessed 21 July 2020).

- Office of the United Nations High Commissioner for Human Rights (1979), Convention on the Elimination of All Forms of Discrimination against Women. New York, NY: Office of the United Nations High Commissioner for Human Rights. 18 December. <u>https://www.ohchr.org/EN/ProfessionalInterest/Pages/CEDAW.aspx</u> (accessed 18 August 2020).
- Okamura, K., T.H. Nguyen, and T. Doi (2020), 'Boosting ASEAN+3's Economic Resilience and Recovery during COVID-19'. Nikkei Asia. 13 October. https://asia.nikkei.com/Opinion/Boosting-ASEAN-3-s-economic-resilience-andrecovery-during-COVID-19 (accessed 11 September 2020).
- Olivier, M. (2018), Social Protection for Migrant Workers in ASEAN: Developments, Challenges and Prospects. Bangkok: ILO. https://www.ilo.org/wcmsp5/groups/public/---asia/---robangkok/documents/publication/wcms_655176.pdf (accessed 26 August 2020).
- Ong, C.B. and C. Peyron Bista (2015), *The State of Social Protection in ASEAN at the Dawn of Integration*. Bangkok: ILO Regional Office for Asia and the Pacific. <u>https://www.ilo.org/wcmsp5/groups/public/---asia/---ro-bangkok/---ilo-jakarta/documents/publication/wcms_428982.pdf</u> (accessed 11 August 2020).
- Open Development Mekong (2020), Sustainable Development Goals. 1 May. <u>https://opendevelopmentmekong.net/topics/sustainable-development-goals</u> (accessed 25 July 2020).
- Orbeta Jr, A.J.C. (2013), 'Enhancing Labor Mobility in ASEAN: Focus on Lower-Skilled Workers', *PIDS Discussion Paper Series*, No. 17. Manila: PIDS.
- Orbeta Jr, A.J.C., and K. Gonzalves (2013), 'Managing International Labour Migration in ASEAN: Themes from a Six-Country Study', *PIDS Discussion Paper Series*, No. 26).
 Makati: PIDS. An earlier version of this paper was earlier prepared for the UNSIRD Conference held on 12-13 January 2013 in Geneva.

https://unrisd.org/80256B3C005BCCF9%2F(httpAuxPages)%2F22788DE1584F00B EC1257D42002A5935%2F%24file%2FOrbeta_Managing%20International%20Labo ur%20Migration%20in%20ASEAN.pdf (accessed 12 July 2020).

Overseas Development Institute (ODI) (2020), *Migration and the 2030 Agenda for Sustainable Development*. September. London: ODI. <u>https://www.odi.org/sites/odi.org.uk/files/resource-documents/12422.pdf</u> (accessed 25 August 2020).

- O'Connor, J. (2020), 'Thai Workers Left Facing Mass Unemployment as Chiang Mai Hotel Staff Protest \$10m Wage Arrears', *Thai Examiner*. 15 May. <u>https://www.thaiexaminer.com/thai-news-foreigners/2020/05/15/workers-mass-unemployment-chiang-mai-hotel-dhara-dhevi-staff-unpaid-wages/</u> (accessed 25 July 2020).
- O'Neil, T. and P. Domingo (2016), *Women and Power: Overcoming Barriers to Leadership and Influence.* February. London: ODI.
- O'Neil, T., A. Fleury, and M. Foresti (2016), *Women on the Move: Migration, Gender Equality and the 2030 Agenda for Sustainable Development*. July. London: ODI.
- Paitoonpong, S. and Y. Chalamwong (2012), *Managing International Labour Migration in ASEAN: A Case of Thailand*. February. Bangkok: Thailand Development Research

Institute.

- Paitoonpong, S. (2011), 'Different Stream, Different Needs, and Impact: Managing International Labor Migration in ASEAN: Thailand (Immigration)', *PIDS Discussion Paper Series*, No. 28. Makati: PIDS.
- Palatino, M. (2020), 'What Is Bogging Down the Philippines' COVID-19 Response?', *The Diplomat*. 20 April. <u>https://thediplomat.com/2020/04/what-is-bogging-down-the-</u> <u>philippines-covid-19-response/</u> (accessed 20 August 2020).
- Pasadilla, G. and M. Abella (2012), 'Social Protection for Migrant Workers in ASEAN', *Working Paper*, No. 3914. Munich: CESifo Group.
- Pasion, P. (2017), *Minimum Wage in ASEAN Countries*. 10 November (republished from Rappler.com). <u>https://aseantuc.org/2017/11/minimum-wage-in-asean-countries/</u> (accessed 22 September 2020).
- Peterman, A., A. Potts, M. O'Donnell, K. Thompson, N. Shah, S. Oertelt-Prigione, and N. van Gelder (2020), 'Pandemics and Violence Against Women and Children', *Center for Global Development Working Paper*, No. 528. 1 April (republished from Rappler.com). <u>https://www.cgdev.org/publication/pandemics-and-violence-against-women-and-children</u>. (accessed on 16 July 2020).
- Phutpheng, P. (2020), Surveys Show that COVID-19 Has Gendered Effects in Asia and the Pacific. New York, NY: UN Women. 29 April. <u>https://data.unwomen.org/resources/surveys-show-covid-19-has-gendered-</u> <u>effects-asia-and-pacific</u> (accessed 5 September 2020).
- Piper, N. (2011), Gender, Migration and Development Emerging Trends and Issues in East and South-East Asia. New York, NY: UN Women.
- Pradesha, A., S. Amaliah, A. Noegroho, and J. Thurlow (2020), *The Cost of COVID-19 on the Indonesian Economy: A Social Accounting Matrix (SAM) Multiplier Approach*.
 Washington, DC: International Food Policy Research Institute.
 https://www.ifpri.org/publication/cost-covid-19-indonesian-economy-social-accounting-matrix-sam-multiplier-approach (accessed 15 August 2020).
- Praiwan, Y. (2020), 'Fuel Retailers Abandon Plans as Virus Cripples Demand', *Bangkok Post.* 15 April. <u>https://www.bangkokpost.com/business/1900140/fuel-retailers-abandon-plans-as-virus-cripples-demand</u> (accessed 25 July 2020).
- Prime Minister's Office of Malaysia (2020), *Prihatin Rakyat Economic Stimulus Package Speech Text*. 27 March. Putrajaya: Prime Minister's Office of Malaysia. <u>https://www.pmo.gov.my/2020/03/speech-text-prihatin-esp/</u> (accessed 6 June 2020).
- Rahman, R. (2020), 'COVID-19 Forces Restaurants, Retailers to Cut Wages, Furlough Employees', *The Jakarta Post*. 28 April. <u>https://www.thejakartapost.com/news/2020/04/28/covid-19-forces-restaurants-retailers-to-cut-wages-furlough-employees.html</u> (accessed 22 August 2020).
- Rajah, V.K. and G. Yihan (2020), 'The COVID-19 Pandemic and the Imminent Legal Epidemic', *The Strait Times*. 7 May. <u>https://www.straitstimes.com/opinion/thecovid-19-pandemic-and-the-imminent-legal-epidemic</u> (accessed 25 September 2020).
- Ratha, D.K., S. De, E.J. Kim, S. Plaza, G.K. Seshan, and N.D. Yameogo (2020), *COVID-19 Crisis Through a Migration Lens.* Migration and Development Brief No. 32.

Washington, DC: World Bank Group.

http://documents.worldbank.org/curated/en/989721587512418006/COVID-19-Crisis-Through-a-Migration-Lens (accessed 20 August 2020).

- 'Region's Migrants Struggle with Job Losses, Hunger Caused by COVID-19' (2020), Benar News. 15 April. <u>https://www.benarnews.org/english/news/indonesian/migrantcovid-04152020165752.html</u> (accessed 25 August 2020).
- Reich, R. (2020), 'It's Morally Repulsive How Corporations Are Exploiting This Crisis. Workers Will Suffer', *The Guardian*. 22 March. <u>https://www.theguardian.com/commentisfree/2020/mar/22/large-corporations-</u> exploiting-coronavirus-crisis (accessed 10 July 2020).
- Roach, S.S. (2020), 'Commentary: The Biggest Shock from COVID-19 Will Be Felt on the US Dollar', *Channel News Asia*. 6 July. <u>https://www.channelnewsasia.com/news/commentary/covid-19-impact-us-</u> <u>dollar-negative-savings-rate-current-account-12903334</u> (accessed 18 September 2020).
- Rutkowski, M. (2020), *How Social Protection Can Help Countries Cope with COVID-19.* World Bank Blog. 15 April. <u>https://blogs.worldbank.org/voices/how-social-protection-can-help-countries-cope-covid-19</u> (accessed 20 August 2020).
- 'Sacked and Abandoned, Foreign Workers Now Living in Squalor', Says MTUC' (2020), Free Malaysia Today. 8 April.

https://www.freemalaysiatoday.com/category/nation/2020/04/08/sacked-andabandoned-foreign-workers-now-living-in-squalor-says-mtuc/ (accessed 19 September 2020).

- Samydorai, S. (2009), *Social Security for Migrant Workers in ASEAN*. Hong Kong: Asia Monitor Resource Center. 1 December. <u>https://www.amrc.org.hk/content/social-</u> <u>security-migrant-workers-asean</u> (accessed 10 July 2020).
- Sandar, N. (2020), "'I Guess We're Just Supposed to Starve" Laid-off Myanmar Migrants Denied Wages and Trapped by Thailand Lockdown', *Myanmar Now*. 14 May. <u>https://myanmar-now.org/en/news/i-guess-were-just-supposed-to-starve-laid-off-myanmar-migrants-denied-wages-and-trapped-by</u> (accessed 10 August 2020).
- Sandford, S. (2020), 'Thailand's Migrant Workers Struggle to Qualify for Aid During Pandemic', VOA News. 17 June. <u>https://www.voanews.com/covid-19-</u> <u>pandemic/thailands-migrant-workers-struggle-qualify-aid-during-pandemic-0</u> (accessed 17 September 2020).
- Satrusayang, C. (2020), 'Construction Sites in Bangkok Go Full Steam ahead Despite the Risk of Coronavirus Outbreaks', *Thai Enquirer*. 14 April. <u>https://www.thaienquirer.com/11386/construction-sites-in-bangkok-go-full-</u> <u>steam-ahead-despite-the-risk-of-coronavirus-outbreaks/</u> (accessed 21 September 2020).
- 'Schools May Reopen "If They Can Follow Rules" (2020), *Bangkok Post.* 30 May. <u>https://www.bangkokpost.com/thailand/general/1926504/schools-may-reopen-if-they-can-follow-rules</u> (accessed 14 July 2020).
- SciDev (n.d.), *SDGs* [Search results]. <u>https://www.scidev.net/asia-pacific/governance/sdgs/</u> (accessed 25 September 2020).

- Searight, A. (2020), *The Economic Toll of Covid-19 on Southeast Asia: Recession Looms as Growth Prospects Dim.* 14 April. Washington, DC: CSIS. <u>https://www.csis.org/analysis/economic-toll-covid-19-southeast-asia-recession-looms-growth-prospects-dim</u> (accessed 7 September 2020).
- Sengpaseuth, P. (2020), 'Govt Officials Urged to Expedite Aid to the Unemployed', Vientiane Times. 22 May. http://www.vientianetimes.org.la/freeContent/FreeConten Govt officials 97.php

(accessed 11 August 2020).

- Serrano, M., M. Marasigan, and V. Pupos (2014), Between Flexibility and Security: The Rise of Non-Standard Employment in Selected ASEAN Countries. Bonn; Kuala Lumpur: Friedrich Ebert Stiftung and the ASEAN Services Employees Trade Unions Council. http://library.fes.de/pdf-files/bueros/singapur/10792.pdf (accessed 6 October 2020).
- Sijapati, B. (2015), *Women's Labour Migration from Asia and the Pacific: Opportunities and Challenges* (12). Bangkok; Washington, DC: IOM and Migration Policy Institute.
- Singapore Ministry of Health (2020), 832 More Cases Discharged, 533 New Cases of COVID-19 Infection Confirmed. 27 May. Singapore: Singapore Ministry of Health. https://www.moh.gov.sg/news-highlights/details/832-more-cases-discharged-533-new-cases-of-covid-19-infection-confirmed (accessed 15 September 2020).
- Singapore Ministry of Manpower (2016), *Labour Force in Singapore 2016*. Singapore: Singapore Ministry of Manpower. <u>https://stats.mom.gov.sg/Pages/Home.aspx</u> (accessed 10 August 2020).
- Singapore Ministry of Manpower (2020), Foreign Workforce Numbers. https://www.mom.gov.sg/documents-and-publications/foreign-workforcenumbers (accessed 11 August 2020).
- Soeriaatmadja, W. (2020), 'Half of Indonesia's Workers Stopped Working in April as a Result of COVID-19 Pandemic: Online Survey', *The Straits Times*. 20 April. <u>https://www.straitstimes.com/asia/se-asia/half-of-indonesias-workers-stopped-working-in-april-as-a-result-of-covid-19-pandemic</u> (accessed 9 August 2020).

Somanas, A. (2020), 'Thai Hospitality Staff Locked out of Government Wage Aid by Absence of Hotel Closure Directive', *TTG Asia.* 2 April. <u>https://www.ttgasia.com/2020/04/02/thai-hospitality-staff-locked-out-of-government-wage-aid-by-absence-of-hotel-closure-directive/</u> (accessed 12 September 2020).

- 'Special Issue: ASEAN Integration and its Health Implications' (2015), *Global Health Action*, 8. <u>https://doi.org/10.3402/gha.v8.27368</u>
- Sri Priya, S. (2020), 'Migrant Workers and Refugees Cry for Help', *The Star (Malaysia).* 9 April. www.thestar.com.my/metro/metro-news/migrant-workers-and-refugeescry-for-help (accessed 12 August 2020).
- 'Stranded in Amsterdam: 14 OFWs Denied Entry Back to the Philippines' (2020), *Migrante International*. 16 June.

https://migranteinternational.org/2020/06/16/stranded-in-amsterdam-14-ofwsdenied-entry-back-to-the-philippines-citing-government-imposed-quotamigrante-international-seeks-immediate-assistance-and-repatriation-forstranded-ofws/ (accessed 25 August 2020).

Su, E. and A. Daga (2020), 'Schools Reopen as Singapore Eases Lockdown Restrictions', US News and World Report. 2 June.

https://www.usnews.com/news/world/articles/2020-06-02/schools-reopen-assingapore-eases-lockdown-restrictions (accessed 20 September 2020).

Subbaraman, R. and S. Varma (2020), 'Asia Has an Unemployment Crisis Waiting to Happen', *Nikkei Asia*. 1 July. <u>https://asia.nikkei.com/Opinion/Asia-has-an-unemployment-crisis-waiting-to-happen</u> (accessed 25 August 2020).

Sukma, R. (2015), 'ASEAN beyond 2015: The Imperatives for Further Institutional Changes, Center for Strategic and International Studies', *ERIA Discussion Paper Series*. https://www.eria.org/ERIA-DP-2014-01.pdf (accessed 10 September 2020).

Tantau, M. (2020), 'Is Indonesia Facing a Looming Food Crisis?' *The Diplomat*. 5 May. https://thediplomat.com/2020/05/is-indonesia-facing-a-looming-food-crisis/ (accessed 10 August 2020).

Taghizadeh-Hesary, F. (2020), 'Economic Impact of COVID-19 Pandemic and Oil Price Collapse', Asia Pathways. 18 May. <u>https://www.asiapathways-</u> <u>adbi.org/2020/05/economic-impacts-covid-19-pandemic-and-oil-price-collapse/</u> (accessed 20 August 2020).

Takenaka, A.K., J. Villafuerte, R. Gaspar, and B. Narayanan (2020a), 'COVID-19 Hitting Migration and Remittances Hard in Developing Asia. Asia Pathways. 25 August. <u>https://www.asiapathways-adbi.org/2020/08/covid-19-hitting-migration-remittances-hard-developing-asia/</u> (accessed 21 September 2020).

Takenaka, A.K., J. Villafuerte, R. Gaspar, and B. Narayanan (2020b), 'COVID-19 Impact on International Migration, Remittances, and Recipient Households in Developing Asia', *ADB Briefs*, No. 148. August. http://dx.doi.org/10.22617/BRF200219-2.

Tamagno, E. (2008), *Strengthening Social Protection for ASEAN Migrant Workers through Social Security Agreements*. Bangkok: ILO (also cited in Olivier, 2018).

Taylor, W. (2011), Vietnam's Migrant Workers: Greatest Advantage, Greatest Challenge.
28 September. San Francisco: The Asia Foundation. <u>http://asiafoundation.org</u> (accessed 7 May 2020).

Tembo, D. and R. Adhikari (2020), *How Poor Countries Can Deal with the Economic Shock* of COVID-19. 24 April. Cologny: World Economic Forum. www.weforum.org/agenda/2020/04/5-ways-poor-countries-can-deal-with-theeconomic-shock-of-covid-19/ (accessed 19 May 2020).

Taskforce on ASEAN Migrant Workers (TF-AMW) (n.d.), Timeline (Facebook Page). <u>https://www.facebook.com/TFAMW</u> (accessed 29 April 2020).

 TF-AMW (2018), Regional Civil Society Consultation on the ASEAN Consensus on the Protection and Promotion of the Rights of the Migrant Workers [Outcome document]. 4 May. New York, NY: TF-AMW. <u>https://workersconnection.s3.amazonaws.com/old/Resources 74/6 June 2018</u> <u>TFAMW_CSO_Submission_on_POA_to_implement_ASEAN_Consensus.pdf</u> (accessed 9 April 2020).

TF-AMW (2020), COVID-19: Key Issues During and Beyond Pandemic to Protect the Rights of Migrant Workers. 26 June. New York, NY: TF-AMW. <u>https://info.scoop.co.nz/Task_Force_on_ASEAN_Migrant_Workers</u> (accessed 17 July 2020).

- 'Thailand Logs 38 New Coronavirus Cases, 1 New Death Tuesday' (2020), *Bangkok Post.* 7 April. <u>https://www.bangkokpost.com/thailand/general/1894930/thailand-logs-</u> <u>38-new-coronavirus-cases-1-new-death-tuesday</u> (accessed 25 September 2020).
- 'Thailand Schools Reopen with Strict Hygiene Rules' (2020), *The Jakarta Post*. 1 July. <u>www.thejakartapost.com/seasia/2020/07/01/thailand-schools-reopen-with-strict-hygiene-rules.html</u> (accessed 23 August 2020).
- Thakur J.S., R. Paika, S. Singh, and J.P. Narain (2020), 'COVID-19 and Noncommunicable Diseases: Impact and the Strategic Approaches', *International Journal of Noncommunicable Diseases*, 5(2), pp.29–35.
- Thubchumpon, N. (2020), *Migrant Workers Struggling under Lockdown in Thailand*. 4 May. Bonn: Friedrich Ebert Stiftung. <u>https://www.fes-asia.org/news/migrant-</u> workers-struggling-under-lockdown-in-thailand/ (accessed 22 July 2020).
- 'The Fourth Industrial Revolution, the Pandemics and the Future of Work' (2020), *The* ASEAN Magazine, 1(2). 19 June.
- The Human Capital and Education for Asian Development Foundation (HEAD) Foundation (2020), 'How Is COVID-19 Impacting on Higher Education in ASEAN?', *Higher Education Southeast Asia and Beyond*, Special Issue No. 8. June. https://headfoundation.org/wp-content/uploads/2020/06/HESB-8-COVID19_2020.pdf (accessed 14 August 2020).
- The Interpreter (2020), *Debate Topic Coronavirus Pandemic*. Lowy Institute. <u>https://www.lowyinstitute.org/the-interpreter/debate/coronavirus-pandemic</u> (accessed 3 October 2020).
- Thomas. J. (2020), 'Worker Accuses Top Glove of Lax COVID-19 Measures', Free Malaysia Today. 18 April.

https://www.freemalaysiatoday.com/category/nation/2020/04/18/workeraccuses-top-glove-of-lax-covid-19-measures/ (accessed 20 May 2020).

- 1 September 2020 (2020), *Minimum Wages Asia* [Data Set]. <u>https://tradingeconomics.com/country-list/minimum-wages?continent=asia</u> (accessed 1 September 2020).
- United States (US) Department of Justice (n.d.), *Child Sex Trafficking*. Washington, DC: US State Department. https://www.justice.gov/criminal-ceos/child-sex-trafficking/ (accessed 17 September 2020).
- US State Department (2014), *The 2014 Trafficking in Persons Report*. June. Washington, DC: US State Department. <u>http://www.state.gov/j/tip/rls/tiprpt/2014/</u> (accessed 17 September 2020).
- US State Department (2020), *The 2020 Trafficking in Persons Report*. June. Washington, DC: US State Department. https://www.state.gov/wp-content/uploads/2020/06/2020-TIP-Report-Complete-062420-FINAL.pdf (accessed 17 September 2020).
- US-ASEAN Business Council (2019), *ASEAN's Economy*. 22 July. Washington, DC: US-ASEAN Business Council. https://www.usasean.org/why-asean/asean-economy (accessed 22 July 2019).
- United Nations (UN) (2020a), *The Impact of COVID-19 on Women.* 9 April. New York: United Nations, Department of Economic and Social Affairs, Statistics Division.

https://www.unwomen.org/-

/media/headquarters/attachments/sections/library/publications/2020/policybrief-the-impact-of-covid-19-on-women-en.pdf?la=en&vs=1406 (accessed 29 May 2020).

UN (2020b), Shared Responsibility, Global Solidarity: Responding to the Socio-economic Impacts of COVID-19. March. New York, NY: UN. <u>https://www.un.org/sites/un2.un.org/files/sg_report_socio-</u>

economic_impact_of_covid19.pdf (accessed 8 May 2020).

- UN General Assembly (2015), *Transforming our World: The 2030 Agenda for Sustainable Development*. A/RES/70/1. 21 October. New York, NY: UN General Assembly. https://sustainabledevelopment.un.org/post2015/transformingourworld (accessed 8 May 2020).
- UN Network on Migration (2020), *Standing in Solidarity with Migrants*. 8 July. Vienna: UNODC.

https://www.unodc.org/documents/press/releases/2020/July/NW_Statement_for _CSOs_final_version.pdf (accessed 25 August 2020).

- UN News (2020), 'COVID-19: Unparalleled Economic Shock Threatens Development Hopes and Gains'. 10 June. <u>https://news.un.org/en/story/2020/06/1066032</u> (accessed 21 August 2020).
- UN Thailand Office (2020), *Technical Note on 'Protecting the Most Vulnerable from the Impact of COVID-19'* (as of 28 March 2020). Bangkok: UN Thailand Office.
- UN Women (2017), Women Migrant Workers in the ASEAN Economic Community. New York, NY: UN Women. <u>https://asiapacific.unwomen.org/-</u> /media/field%20office%20eseasia/docs/publications/2017/06/aec-womenmigration-study.pdf?la=en&vs=4122 (accessed 11 May 2020).
- UN Women (2019), Gaining Protection for Indonesia's Migrant Workers and Their Families. 12 July. https://www.unwomen.org/en/news/stories/2019/6/featurestory-of-change-protection-for-indonesias-migrant-workers (accessed 11 May 2020).
- UN Women (2020a), Action Brief: A Guide for Action to Stem Increasing Violence against Women amid the COVID-19 Pandemic in Asia and the Pacific. New York, NY: UN Women. https://asiapacific.unwomen.org/-

/media/field%20office%20eseasia/docs/publications/2020/05/final%20evaw%20r oap%20action%20brief%20covid-19_4%20may%202020_1.pdf?la=en&vs=744 (accessed 16 August 2020).

UN Women (2020b), *Policy Brief: The Impact of COVID-19 on Women.* 9 April. New York, NY: UN Women. https://www.unwomen.org/-

/media/headquarters/attachments/sections/library/publications/2020/policybrief-the-impact-of-covid-19-on-women-en.pdf?la=en&vs=1406 (accessed 28 July 2020)

UN Women (2020c), Surveys Show that COVID-19 Has Gendered Effects in Asia and the *Pacific.* 29 April. New York, NY: UN Women.

https://data.unwomen.org/resources/surveys-show-covid-19-has-gendered-effects-asia-and-pacific (accessed 16 May 2020).

UN DESA (2016a), International Migrant Stock 2015: By Age and Sex [Data Set]. New York, NY: UN DESA.

www.un.org/en/development/desa/population/migration/data/estimates2/estim ates15.

- UN DESA (2016b), Sustainable Development Goals. New York, NY: UN DESA. https://sustainabledevelopment.un.org).
- UN DESA (2017). Trends in International Migrant Stock: The 2017 Revision [Data Set]. New York, NY: UN DESA. <u>https://www.un.org/en/development/desa/population/migration/data/estimates</u>

2/estimates17.asp (accessed 5 February 2021).

- UN DESA (2018), Promoting Inclusion through Social Protection: Report on the World Social Situation. New York, NY: UN DESA. https://www.un.org/development/desa/dspd/wpcontent/uploads/sites/22/2018/07/1-1.pdf
- UN Economic and Social Commission for Asia and the Pacific (UNESCAP) (2017), ASEAN SDG Baseline. Bangkok: UNESCAP.

https://www.unescap.org/sites/default/files/publications/ASEAN_SDG_Baseline_ 0.pdf (accessed 23 September 2020).

UNESCAP (2017b), Southeast Asia Subregion Challenges and Priorities for SDG Implementation. 17 May. Bangkok: UNESCAP.

https://www.unescap.org/resources/southeast-asia-subregion-challenges-andpriorities-sdg-implementation (accessed 12 February 2021).

UNESCAP (2020a), Asia and the Pacific SDG Progress Report 2020. 25 March. Bangkok: UNESCAP.

https://www.unescap.org/sites/default/files/publications/ESCAP_Asia_and_the_P acific_SDG_Progress_Report_2020.pdf (accessed 18 August 2020).

- UNESCAP (2020b), Impact of COVID-19 on Agriculture in the Asia-Pacific Region and Role of Mechanization. 18 June. Bangkok: UNESCAP. <u>https://www.unescap.org/news/impact-covid-19-agriculture-asia-pacific-regionand-role-mechanization</u> (accessed 24 July 2020).
- UNESCAP (2020c), *Policy Brief: Impact of COVID-19 on Southeast Asia.* 30 July. Bangkok: UNESCAP. <u>https://www.unescap.org/sites/default/d8files/2020-07/SG-Policy-brief-COVID-19-and-South-East-Asia-30-July-2020.pdf</u> (accessed 30 August 2020).
- United Nations Capital Development Fund (2020), *Impact of COVID-19 on Myanmar's Migrants and Remittances.* 15 June. New York, NY: United Nations Capital Development Fund. https://www.uncdf.org/article/5725/impact-of-covid-19-onmyanmars-migrants-and-remittances (accessed 28 July 2020).
- United Nations Children's Fund (2020), *COVID-19 May Push Millions More Children into Child Labour – ILO and UNICEF.* 12 June. New York, NY: United Nations Children's Fund. <u>https://www.unicef.org/rosa/press-releases/covid-19-may-push-millions-</u> <u>more-children-child-labour-ilo-and-unicef</u> (accessed 25 September 2020).
- United Nations Department of Economic and Social Affairs and IOM (2019), *Policy Brief No. 1: SDG Indicator 10.7.2.* October. New York, NY; Geneva: United Nations Department of Economic and Social Affairs and IOM.

https://www.un.org/en/development/desa/population/publications/pdf/technica

I/Policy%20Brief 10.7.2.pdf (accessed 21 August 2020).

United Nations Development Programme (2020), COVID-19 Sparks Urgency around Justice for Women, New Report Calls for Action. 21 May. New York, NY: United Nations Development Programme.

https://www.undp.org/content/undp/en/home/newscentre/news/2020/COVID19_sparks_justice_for_women_new_report_calls_actio n.html (accessed 11 August 2020).

- United Nations Office on Drugs and Crime (UNODC) (2015), *The Role of Recruitment Fees and Abusive and Fraudulent Practices of Recruitment Agencies in Trafficking in Persons*. Vienna: UNODC. https://www.unodc.org/documents/humantrafficking/2015/Recruitment_Fees_Report-Final-22_June_2015_AG_Final.pdf (accessed 25 August 2020).
- UNODC (2016), *The Global Report on Trafficking in Persons 2016*. New York: United Nations. Sales No. E.16.IV.6. https://www.unodc.org/documents/data-and-analysis/glotip/2016_Global_Report_on_Trafficking_in_Persons.pdf (accessed 22 August 2020).
- Umanath, M. (2020), 'COVID-19: Link between Migrants, Development Goals Becomes More Clear', *Down to Earth*. 23 June.

https://www.downtoearth.org.in/blog/governance/covid-19-link-betweenmigrants-development-goals-becomes-more-clear-71918 (accessed 4 September 2020).

- Uy, J.D. (2020), 'Singapore Morning News for May 7, 2020', AEC News Today. https://aecnewstoday.com/2020/singapore-morning-news-for-may-7-3/
- Uznanska, P. and M. Fila (2020), 'The New US-China Battlefield over the COVID-19 Narrative: Poland', *The Diplomat*. 22 May. <u>https://thediplomat.com/2020/05/the-new-us-china-battlefield-over-the-covid-19-narrative-poland/</u> (accessed 7 July 2020).
- Villafuerte, J. and A.K. Takenaka (2020), 'Tracking COVID-19's Devastating Toll on Asia's Remittances', Asian Development Blog. 3 August. <u>https://blogs.adb.org/blog/tracking-covid-19-s-devastating-toll-asia-s-remittances</u> (accessed 4 October 2020).
- Virgil, D. and R. Lie (2020), 'ASEAN's COVID-19 Response: Why Minorities and the Most Vulnerable Matter', Global Asia, 15(2). June. <u>https://www.globalasia.org/v15no2/focus/aseans-covid-19-response-why-</u> <u>minorities-and-the-most-vulnerable-matter_dominique-virgilroberto-lie</u> (accessed 7 July 2020).
- Võ Hải Minh (2012), 'Challenge of the Imbalance of Economic-social Development in ASEAN', in Y. Zhang, F. Kimura, and S. Oum (eds.), *Moving Toward a New* Development Model for East Asia – The Role of Domestic Policy and Regional Cooperation. ERIA Research Project Report 2011-10. Jakarta: ERIA. pp.121–57.
- Vu, K. (2020), ASEAN Economic Prospects amid Emerging Turbulence: Development Challenges and Implications for Reform. Washington, DC: Brookings Institution, Foreign Policy. July. <u>https://www.brookings.edu/wp-</u> <u>content/uploads/2020/07/FP_20200710_asean_economic_prospects_vu.pdf</u> (accessed 19 September 2020).

- 'WHO (World Health Organization) Appeals for Aggressive Measures as European Nations See Silver Linings in Combating COVID-19' (2020), China Daily. 29 April. <u>http://www.vientianetimes.org.la/Region_world/Reworld_WHO.php</u> (accessed 10 July 2020).
- World Health Organizations (WHO) (n.d.a), Southeast Asian Region: Noncommunicable Diseases. Noncommunicable Diseases and Mental Health. Geneva: WHO.
 https://www.who.int/nmh/ncd-tools/who-regions-south-east-asia/en/ (accessed 27 October 2020).
- WHO (n.d.b), Tuberculosis and COVID-19. Geneva: WHO. https://www.who.int/teams/global-tuberculosis-programme/covid-19 (accessed 27 October 2020).
- WHO (2018), Health of Refugees and Migrants: Regional Situation Analysis, Practices, Experiences, Lessons Learned and Ways Forward. New Delhi: WHO South-East Asia Region. January. <u>https://www.who.int/migrants/publications/SEARO-</u> report.pdf?ua=1 (accessed 27 October 2020).
- WHO (2020a), Rolling Updates on Coronavirus Disease (COVID-19). Geneva: WHO. 31 July. <u>https://www.who.int/emergencies/diseases/novel-coronavirus-2019/eventsas-they-happen</u> (accessed 27 October 2020).
- WHO (2020b), The Impact of the COVID-19 Pandemic on Noncommunicable Disease Resources and Services: Results of a Rapid Assessment. Geneva: WHO. September. <u>https://www.who.int/teams/noncommunicable-diseases/covid-19</u> (accessed 27 October 2020).
- Walden, M. and S. Wijaya (2020), 'Millions of Indonesian Migrant Workers Face Hardship in Malaysia amid Coronavirus Fallout', *ABC News*. 26 July. https://www.abc.net.au/news/2020-07-26/indonesians-face-hardship-inmalaysia-amid-coronavirus-fallout/12486548 (accessed 10 September 2020).
- Walk Free Foundation (2016), 'Global Slavery Index' [Data Set]. <u>https://www.globalslaveryindex.org/</u> (accessed 25 February 2021)
- Weng, L. (2020), 'Myanmar's Mon State Consolidates Quarantine Centers for Returning Migrants', *The Irrawaddy*. 6 April. <u>https://www.irrawaddy.com/specials/myanmarcovid-19/myanmars-mon-state-consolidates-quarantine-centers-returningmigrants.html</u> (accessed 25 June 2020).
- Wherli, Z. (2020), COVID-19 Brief: Impact on Refugees. U.S. Global Leadership Coalition.
 18 July. Washington, DC: US Global Leadership Coalition.
 https://www.usglc.org/coronavirus/refugees/ (accessed 10 August 2020).
- Wickramarasekara, P. (2002), 'Asian Labour Migration: Issues and Challenges in an Era of Globalization', *International Migration Papers' Series*, No. 57. Geneva: ILO.
- Women's Legal and Human Rights Bureau (2017), *The Missing Women: The Implications of ASEAN Economic Integration on Women Migrant Workers*. Kuala Lumpur: International Women Rights Action Watch Asia-Pacific.
 https://l6dayscampaign.org/wp-content/uploads/2019/01/missing-women.pdf (accessed 18 August 2020).
- World Bank (2020a), East Asia and the Pacific in the Time of COVID-19 Regional Economic Update. April. Washington, DC: World Bank.

economic-update (accessed 19 June 2020).

World Bank (2020b), World Bank Predicts Sharpest Decline of Remittances in Recent History. 22 April. Washington, DC: World Bank.

https://www.worldbank.org/en/news/press-release/2020/04/22/world-bank-predicts-sharpest-decline-of-remittances-in-recent-history (accessed 19 June 2020).

World Bank (2020c), COVID-19 to Impact Lao PDR Growth, Debt in 2020: New World Bank Report. 20 May. Washington, DC: World Bank. <u>https://www.worldbank.org/en/country/lao/publication/covid-19-to-impact-lao-</u>

pdr-growth-debt-in-2020-new-world-bank-report (accessed 19 June 2020).

World Bank (2020d), Remittance Prices Worldwide, Issue 35. September. Washington, DC: World Bank.

https://remittanceprices.worldbank.org/sites/default/files/rpw_report_septembe r_2020.pdf (accessed 10 October 2020).

- World Council of Churches (2020), *Migrant Workers Bear Brunt of COVID-19, Observes Webinar from Christian Conference of Asia.* 4 May. Amsterdam: World Council of Churches. <u>https://www.oikoumene.org/en/press-centre/news/migrant-workers-</u> <u>bear-brunt-of-covid-19-observes-webinar-from-christian-conference-of-asia</u> (accessed 10 May 2020).
- World Economic Forum (2020), *The Future of Jobs Report*. October. Washington, DC: World Economic Forum. <u>https://www.weforum.org/reports/the-future-of-jobs-report-2020</u>.
- World Law Group (2020), Legal Briefing on the Philippines and COVID-19 Restrictions that Affect Privacy Laws. 5 May.

<u>https://www.theworldlawgroup.com/news/covid-19-legal-information-briefing-focus-on-tmt</u> (accessed 8 July 2020).

- World Travel and Tourism Council (2020), *Travel and Tourism Will Be Critical to Asia Pacific's Economic Recovery, Says WTTC*. 9 April. https://wttc.org/News-Article/Travel-and-Tourism-will-be-critical-to-Asia-Pacifics-economic-recoverysays-WTTC (accessed 19 June 2020).
- Worldometer (2020), COVID19 Coronavirus Pandemic. https://www.worldometers.info/coronavirus/ (accessed 9 May 2020).

 Yasmi, Y., D. Dawe, J. Zhang, J. Balie, and G. Dixie (2020), *Policy Brief: Safeguarding Food Systems in Southeast Asia Amid COVID-19.* Policy brief dated June 2020. Los Baños, Philippines: International Rice Research Institute. <u>https://www.irri.org/news-and-events/news/policy-brief-safeguarding-food-</u> <u>systems-southeast-asia-amid-covid-19</u> (accessed 29 October 2020).

- Yayboke, E. (2020), Can I Stay or Can I Go Now? Longer-Term Impacts of Covid-19 on Global Migration. Washington, DC: CSIS. 20 April. https://www.csis.org/analysis/can-i-stay-or-can-i-go-now-longer-term-impactscovid-19-global-migration (accessed 12 May 2020).
- Yee, T.H. (2020), 'Forced Out of Jobs, Migrants Face Suspicion on Return Home', *The Straits Times.* 10 April. https://www.straitstimes.com/asia/se-asia/forced-out-ofjobs-migrants-face-suspicion-on-return-home (accessed 12 May 2020).

- Yong, C. (2020), 'Coronavirus: Maids Should Still Stay Home on Rest Days, Not Meet up with Friends from June 2, Says MOM', *The Strait Times*. 21 May. <u>https://www.straitstimes.com/singapore/coronavirus-maids-should-still-stayhome-on-rest-days-not-meet-up-with-friends-from-june-2</u> (accessed 14 August 2020).
- Yusof, A., M. Muuti, L. Ariffin, and M.K.N. Tan (2020), 'Sharing Information on COVID-19: The Ethical Challenges in the Malaysian Setting', *Asian Bioethics Review*, 12(25 June), pp.349–61. https://doi.org/10.1007/s41649-020-00132-4
- Zaw Zaw Htwe (2020), 'Without Work and Fearing Arrest, Undocumented Myanmar Migrants in Malaysia Take Their Own Lives', *The Irrawaddy*. 17 June. <u>www.irrawaddy.com/news/burma/without-work-fearing-arrest-undocumented-</u> <u>myanmar-migrants-malaysia-take-lives.html</u> (accessed on 4 September 2020).
- Zaw Zaw Htwe (2020), 'Quarter of a Million Myanmar Workers Left Jobless Due to COVID-19', *The Irrawaddy*. 26 June. <u>https://www.irrawaddy.com/news/burma/quarter-million-myanmar-workers-left-jobless-due-covid-19.html</u> (accessed on 4 Septemebr 2020).
- Zulkhibri, M. and J.B. Sinay (2020), ASEAN Policy Brief No. 2: Assessing ASEAN Economy Policy Responses in a Pandemic. 28 May. Jakarta: ASEAN. <u>https://asean.org/storage/2020/06/ASEAN-Policy-Brief-02_31-May-2020.pdf</u> (accessed 28 October 2020).